REIMAGINING POLITICAL ADVOCACY IN THE AGE OF SOCIAL MEDIA: INSIGHTS FROM THE GRASSY MOUNTAIN MINE PROJECT

By © Katarina Nedeljakova

A Research Thesis Submitted to the School of Graduate Studies in partial fulfillment of the requirements for the degree of

Master of Arts in Environmental Policy

Environmental Policy Institute Grenfell Campus- Memorial University of Newfoundland Corner Brook, Newfoundland and Labrador

September 2023

Abstract

The emergence of social media platforms as a major communication technology has resulted in the reimagining of political advocacy and mobilization, as well as our position in relation to these tools. Through a case study of the recently terminated Grassy Mountain Mine project in Alberta, this work explores the tensions and complexities of environmental political participation in online and physical space, focusing on the use of Facebook as a tool for expressing political interests and enabling collective action. A mixed methods approach was taken to investigate this objective, including a scoping social media analysis of 484 posts made on public Facebook groups related to coal mining and in-depth participant interviews. Within this case study, motivations for political engagement via social media often stem from deep emotional ties to the physical space at the center of the movement. Additionally, this study provides evidence that the digital infrastructure of Facebook plays a role in how political participation is enacted online. This thesis contributes to existing knowledge on how humans navigate political participation and conceptualize space. It can also serve as a useful document for governing bodies seeking to effectively utilize online platforms for public engagement, such as consultations.

Keywords: collective action, social movements, social media, online platforms, physical space, coal mining

General Summary

Social media platforms have changed how we voice our political interests and come together with the people around us to support these efforts. This research focuses on how Facebook was used as a tool for people to advocate for their stance on the (now terminated) Grassy Mountain Mine project in Alberta. Doing this will help us understand the complexities of environmental political participation in both online and physical spaces. Both Facebook posts and interviews with the people who posted are used to explore these ideas. This research finds that people are often motivated to engage politically on social media because of strong emotional connections to the physical space and also shows that the design of Facebook influences how people come together to talk about their political beliefs. This thesis adds to our understanding of how individuals navigate political engagement and perceive spaces and may be helpful for government bodies using online platforms to reach citizens.

Acknowledgements

I extend my deepest gratitude to the amazing people across Canada (and beyond) that made this thesis work possible:

First of all, everyone who dedicated their time to participate in this research. From those who created the online content at the heart of this work to the amazing folks I had the pleasure of interviewing; this thesis would not exist without you.

My incredible supervisor, Roza Tchoukaleyska, who easily deserves at least five more pages of acknowledgement. I am so lucky to have you as my supervisor- your expertise, level-headedness, and kindness made even the toughest parts of this process feel doable. I consistently leave our meetings feeling inspired and with renewed appreciation for the value of my work. Working with you these past three years has been an absolute privilege and makes me excited at the prospect of pursuing research beyond this thesis.

My thesis committee, Stephen Decker and Garrett Richards, for your invaluable guidance and expertise throughout all steps of writing this thesis. I wholeheartedly appreciate all the time and energy you invested into helping make this work what it is today.

The many individuals at Memorial University of Newfoundland, the School of Science and the Environment, and Grenfell Campus for your help along the way. From working through the logistics of doing school remotely to conversations that offered new perspectives, your contributions to my positive experience in this program are invaluable.

All my support systems beyond the academic setting. I am endlessly grateful for the unconditional support of my amazing friends, family, and pets throughout this process- the last three years of my life would have looked a lot different without all of you. Thank you for always feeding me when I was stressed, letting me bounce ideas off you, and convincing me that it is okay to take breaks.

Finally, a big shout-out to Pippin the cat. The many cryptic messages you left when I stepped away from my laptop inspired me to think outside of the box and trained me to *always* check that AutoSave is on :).

Table of Contents

Figures	
Tables	ix
Acronyms	X
Chapter 1: Introduction	1
1.1 Theoretical Background	2
1.2 Case Description and Context	3
1.3 Research Approach	5
1.4 Justification, Methodology, and Methods	7
1.5 Thesis Organization	
Chapter 2: Literature Review	
2.1 Collective Action	11
2.1.1 Definitions	
2.1.2 Theoretical Roots of Social Movement Theory	
2.1.3 Social Movements in a Technological Era	
2.2 The Online World	
2.2.1 Where is "Online"?	
2.2.2 Techno-social Dimensions of Social Media	25
2.2.3 Political Participation Online	
2.3 Covid-19	
2.3.1 Interactions with Space During Covid-19	
2.3.2 Social Movements During Covid-19	
2.4 Next Steps	
Chapter 3: Methodology and Methods	
3.1 Guiding Frameworks	
3.1.1 Defining Political Participation	
3.1.2 What is a Social Media Analysis?	
3.2 Methodology	
3.2.1 Positionality	
3.2.2 Justification	
3.2.3 Methodology	
3.3 Methods	45

3.3.1 Social Media Analysis	45
3.3.2 Supplementary Interviews	
3.4 Case Description	53
3.4.1 The Politics of Alberta's Energy Sector	
3.4.2 Environmental Regulation of the Energy Sector	
3.4.3 Crowsnest Pass and the Grassy Mountain Coal Project	60
3.5 Conclusion	
Chapter 4: Social Media Analysis	67
4.1 Basic Group Information	67
4.2 Structures and Processes Within a Social Movement	77
4.3 Conclusion	
Chapter 5: Interview Findings	
5.1 Individuals as the Unit of a Social Movement	
5.2 Social Media Processes: Building a Network	
5.3 Politics of the Platform	
5.4 Covid-19, Social Media, and Political Advocacy	
5.5 Conclusion	
Chapter 6: Discussion and Conclusions	
6.1 Thesis Overview	
6.1.1 Background and Theoretical Basis	
6.1.2 Case Study	
6.1.2 Research Design	
6.1.3 Research Findings and Analysis	
6.2 Discussion and Future Directions	
6.2.1 Emotion, Identity, and Space	
6.2.2. Impacts of Platform Structure on Collective Action	
6.2.3 Concluding Remarks	
Appendices	
Appendix A: Supplementary Social Media Data	
Weekly summaries, by group	
Embedded content structures	
Appendix B: Application for Ethics Review Form (GC-REB)	

Appendix C: 1976 Coal Development Policy Map	140
Appendix D: Social Media Terminology	142
Works Cited	144

Figures

Figure 1a. Map: Grassy Mountain Project Area Overview (pg. 6)

- Figure 1b. Map: Alberta Map with Crowsnest Pass (pg. 6)
- Figure 2.1. Connective action model summary (pg. 21)
- Figure 3.1. Political participation flowchart (pg. 36)
- Figure 3.2. Social media analysis data collection framework (pg. 48)
- Figure 3.3. Map: Crowsnest Pass situated within Alberta (pg. 59)
- Figure 3.4. Map: Grassy Mountain Project Area Overview (pg. 60)
- Figure 3.5. Timeline of Grassy Mountain Project key events (pg. 63)
- Figure 4.1. A timeline of posting activity across all groups (pg. 68)
- Figure 4.2. A count of post types across all groups (pg. 70)
- Figure 4.3. Average number of interactions for each post type across all groups (pg. 71)
- Figure 4.4. Engagement breakdown for posts analyzed across all groups. (pg. 74)
- Figure 4.5a. Post purpose in the PARH group (pg. 76)
- Figure 4.5b. Post purpose in the CPGT group (pg. 76)
- Figure 4.6. A word cloud of hashtags in the PARH group (pg.79)
- Figure 4.7. Embedded content in the PARH and CPGT groups (pg. 81)
- Figure 4.8. References to external organizations (pg. 82)
- Figure 5.1. A word cloud created from interview data (pg. 89)

Tables

Table 3.1. Summary table: total number of posts collected (pg. 48)

 Table 4.1. Groups examined in social media analysis (pg. 65)

Table 4.2a. CPGT text analysis, codes, and examples (pg.80)

 Table 4.2b. PARH text analysis, codes, and examples (pg.81)

Acronyms

AACP- Around and About Crowsnest Pass
AB - Alberta
AER- Alberta Energy Regulator
API - application programming interface
BLM- Black Lives Matter
BC - British Columbia
CAPP- Canadian Association of Petroleum Producers
CBC- Canadian Broadcasting Corporation
CEAA - Canadian Environmental Assessment Act
CPAWS- Canadian Parks and Wilderness Society
CPGT- Crowsnest Pass. Ghost town
CPR- Canadian Pacific Railway
CPU- Crowsnest Pass- Uncensored
EIA- Environmental Impact Assessment
EPEA- Environmental Protection and Enhancement Act
GDP- gross domestic product
GHG- Greenhouse Gas
GIF- Graphics Interchange Format
IAA- Impact Assessment Act
LLG- Livingstone Landowners Group
MLA- Member of the Legislative Assembly
NDP- Alberta New Democratic Party
PARH- Protect Alberta's Rockies and Headwaters
PC- Progressive Conservative Party

REDA- Responsible Energy Development Act

SMO- social movement organization

SARA- Species at Risk Act

UCP- United Conservative Party

Chapter 1: Introduction

Over the course of the past two decades, we have been living through a sociotechnical transformation of the space around us (Dijck & Poell, 2015). From talking to friends to working a job to participating in politics, the ubiquity of human interactions with the world through technology can make it easy to forget the extent to which our lifestyles are not only intertwined with but experienced through the online world. Many of these cultural shifts can be attributed to increased connectivity due to widespread internet access, the convenience of personal electronic devices, the availability of digital communication tools, and recently, a push further into the online world in response to the Covid-19 pandemic (Milan & Velden, 2016; Tufekci, 2018; Rader et al., 2020; Ritchie, 2022). As part of a transformation that blurs the lines between the physical and the virtual, advocating for political interests and mobilizing networks looks different than ever before. In parallel, there has also been increasing attention to the environmental and social impacts of human development, with global issues such as climate change at the forefront of conversation within communities, governments, academia, and industry (Arya & Henn, 2021; Oláh et al., 2020; Kuai et al., 2023).

At the center of this research are tensions around how environmental political participation within physical space is conceptualized and enacted through the online sphere. Through a case study of the terminated Grassy Mountain Mine in southwest Alberta and the voices that shaped this decision, this thesis explores political advocacy and the complexities of building meaning between tangible and intangible realities in a period when our relationships with the spaces we occupy was different than ever before. With social media platforms acting as an organizational agent to most of today's social movements in some capacity, gaining a full picture of these tensions also requires attention to how digital communication platforms themselves are capable of influencing activist action and power dynamics of this relationship between human-technology interactions (van Dijck & Poell, 2015). The use of Facebook to organize and express political interests in the case of the Grassy Mountain mine allows us to take a closer look at the role of social media platforms as tools for political participation and how digital infrastructure shapes the ways individuals navigate and collect within the greater activist landscape. As such, this project offers valuable insight on the intersection of the environment as a political space and the online sphere to advocate for political interests.

1.1 Theoretical Background

The study of social movements has a long history, but the widespread access to the internet in the mid-2000s marked a new era of connectivity and social movement theory. The Arab Spring in 2011 paved the way for a shift towards social movements utilizing large-scale digital media and online platforms, giving rise to the concept of "connective action" (Bennett & Segerberg, 2012; Tufekci, 2018). Bennett and Segerberg's (2012) framework of connective action emphasizes personalized communication, individualized action frames, and the role of digital networks in mobilizing collective action. It highlights the decentralization of organizing agents and the formation of weak tie networks. However, questions have been raised about the political nature of online spaces and the impact of social media as a commercial entity on collective action (Milan, 2015; Milan & Velden, 2016).

Understanding the influence of social media on activist action is valuable to this conversation, as these platforms often serve as organizational agents for social movements. While they offer benefits such as rapid content turnover, personalized narratives, and weak tie networks that facilitate online collective action, they also contribute to the short-lived nature of online movements, echo chambers, and decreased political deliberation (Vitak et al., 2010; Greijdanus et al., 2020). Social media entities have biased interests that shape user experience, but online social movements are not

solely a product of their setting (Milan, 2015). Research suggests generally positive relationships between online and offline activism at the individual level, with the context of participation playing a significant role (Halpern et al, 2017). Facebook, in particular, has been associated with higher collective efficacy and a greater impact on offline political participation compared to other platforms like Twitter (Halpern et al., 2017).

One concept that can guide the interpretation of interactions across online and physical space is the notion of thirdspace, as proposed by Soja (1998). Thirdspace challenges existing binaries and recognizes that the value assigned to a space, whether online or physical, is dynamic and continuously enacted through individuals' experiences¹. Applied to the online sphere, thirdspace finds parallels in the dynamics of social media, enabling connectivity and meaning making while transcending traditional boundaries (Milan, 2015). However, it is important to acknowledge that the online space, despite its transformation, is owned, facilitated, and constructed by profit-driven corporations (Heyman & Pierson, 2015). Milan's (2015) concept of "cloud networking" further emphasizes the sociotechnical nature of digital collective action, allowing users to renegotiate boundaries, form personalized narratives, identities, and networks.

1.2 Case Description and Context

The Grassy Mountain Mine project was a proposed metallurgical coal mining project that, after six months of scrutiny by both the federal and provincial governments, was terminated in August of 2021. Grassy Mountain itself is situated within the Rocky Mountains of south-west

¹ The hybridity of the online and physical world through thirdspace is one way of understanding the complex scaffolding of human interaction with and within space. Other frameworks, such as place attachment theory applied to landscape, offer valuable insight into this topic through the lens of human psychology (Proshansky, 1983; Rajala et al., 2020). Though acknowledged within this thesis, a full exploration of landscape attachment and identity across physical and digital space is left for future work.



Grassy Mountain mining project near Crowsnest Pass, Alberta (Government of Canada, 2020b)

Alberta and is adjacent to the specialized municipality of Crowsnest Pass, AB, whose 5,642 residents are largely employed in mining or related sector (Municipality of Crowsnest Pass, 2019; Government of Alberta, 2020a). The broader geographical region in which Grassy Mountain is referred to as the Eastern Slopes and is home to the headwaters of the South Saskatchewan River, which serves as a primary water source for over 3 million people ad supporting variety of wildlife (Figure 1).

Though the Grassy Mountain Mine approval process began in December 2015, the project did not gain significant media attention until June 2020 (during major Covid-19 restrictions), when the 1976 Coal Policy was repealed without any public consultation (Government of Alberta, 2022a). This resulted in decreased environmental protection for the Eastern Slopes and began the public outcry against the Grassy Mountain Coal Mine. By the end of 2020 the Government of Canada had received pushback against the approval of the mine during online consultations and established a review panel to further examine the impacts of the mine. On the side of the public, a grassroots movement called "Protect Alberta's Rockies and Headwaters" emerged as a Facebook group, advocating against the Grassy Mountain Mine due to concerns of environmental damage. The movement gained rapid support, amassing over 37,000 followers within a few months. However, there was also an opposing, pro-mine group called "Crowsnest pass. Ghost town." established shortly after, as well as various conversations in other groups not specific to the issue of coal mining. Ultimately, the project was denied by the Alberta Energy Regulator in June 2021 and officially shut down by the Government of Canada in August 2021 (2021c).

As background to this case study, it is important to note that these debates and policy decisions took place between December 2020 and August 2021, meaning that this research is set within a greater context of Covid-19. The pandemic has caused a significant shift in how we navigate the world, with people increasingly relying on digital spaces for various aspects of their daily lives, including social interaction and information consumption (Brückmann, 2022). Social media platforms have played a dual role during the Covid-19 pandemic, acting as both a safe alternative to pre-pandemic activities and a space for protest and contention (Arya & Henn, 2021). Simultaneously, individuals' interactions with physical spaces changed due to safety concerns and public health restrictions (Xu et al., 2021). This has led to restrictions on the use of public and common spaces and heightened surveillance, creating a novel experience for many individuals while exacerbating existing inequalities in with marginalized communities facing greater challenges in protecting their well-being and safety (Bhui, 2021; Stahl, 2021).

1.3 Research Approach

Through an analysis of online content regarding the Grassy Mountain Mine project and participant interviews, this thesis examines how social media as an advocacy tool has shaped the ways in which individuals advocate for their environmental and political interests in both the online sphere and in physical space. I explore this topic through the following questions:

- How is political activism conceptualized in the online-to-real world continuum in the Grassy Mountains Mine debate?
- 2. In what ways do individual actors within the Grassy Mountains Mine debate use social media tools to communicate their interests?
- 3. How does digital infrastructure transform capacity building and communication within social movements?

To address the research questions, a thematic analysis of social media content (both posts and comments) regarding the proposed Grassy Mountain Mines resource extraction project in Crowsnest Pass was conducted. To supplement the social media analysis findings, select Facebook group participants were interviewed to gain a deeper understanding of the political climate and provide further context to themes which emerged from the social media analysis. Data collection and analysis focused primarily on the content and discourse within public Facebook groups regarding proposed resource extraction projects in the Crowsnest Pass region from March 17, 2020 (provincial state of emergency first declared) to August 6, 2021 (Grassy Mountain Mines project suspended). Specifically, the groups "Protect Alberta's Rockies and Headwaters", "Crownsest pass. Ghost town", "Around and About Crowsnest Pass", and "Crowsnest Pass- Uncensored" were examined, as they were either dedicated primarily towards expressing interests regarding resource extraction projects specific to the Crowsnest Pass region or are community groups where local issues are discussed.

1.4 Justification, Methodology, and Methods

Sitting at the nexus of digitally-mediated political participation in both physical and online settings, the rich political discourse that occurred within the online sphere regarding the Grassy Mountain Mine, the diversity of actors taking a political stance on the project, and the final policy decision of behalf of the Government of Canada makes this a valuable case study for exploring how individuals mobilize to advocate for their environmental interests in the online sphere. In addition, the characteristics of the social media platforms that enable these communications are rooted in proprietary algorithms and generally lack both transparency and predictability, which adds further complexity to the topic. This research draws connections between a variety of existing theory through an analysis of online discourse regarding the Grassy Mountain Mine project and aims to fill existing gaps in research on environmental social media activism in an increasingly technological age. Beyond this case study, the insights gained through this work also hold relevance when examining the ongoing integration of emerging technologies with our lifestyles and resulting shift in societal values. With a current spotlight on the impacts and relevance of general-purpose technology, such as machine-learning systems, questions around the intersections between human and machine systems are growing increasingly relevant from both a policy and academic standpoint (Brescia, 2020).

Though the main component of this project is a social media analysis, supplementary semistructured interviews are also conducted to support findings from the social media analysis. Conversations regarding the mining industry are a controversial topic and more specific prompts during a semi-structured interview are likely to give greater depth to the findings than just a standalone analysis of social media. Additionally, research by Elder (2020) shows that online comments and responses, especially on public groups, are likely to focus on attacks of character when viewpoints are questioned, which often shuts down conversations on online platforms. As such, the

aim of interviews with individuals active in the groups is to have a more nuanced conversation and clarify any points that may be unclear in the social media analysis.

1.5 Thesis Organization

This thesis is organized into the following six chapters to examine how environmental advocacy is enacted through the use of social media as a communication tool.

Chapter 1 provides a brief introduction of the theoretical foundation of digital media activism and the purpose of this thesis. It consists of a background section, a case description of the Grassy Mountain mining project, the purpose statement, research questions, project justification and methodology, a brief methods overview, and an outline of this thesis work.

Chapter 2 provides a review of current literature relating to digital social movements and associated frameworks. It examines existing work regarding political participation online, the use of social media during, and intersection of these two topics.

Chapter 3 describes the methodology and methods used to meet the research objectives. This includes a justification of any frameworks used, a detailed description of data collection methods used for the social media analysis, the approaches taken to the interviews, and an outline of the data analysis process.

Chapter 4 is the first analytic chapters discussing the results of this research. This Chapter focuses on presenting and discussing findings from the social media analysis, applying a more quantitative approach to the research objectives.

Chapter 5, the second analytic Chapter of this thesis, takes a deep dive into the personal experiences of four individuals advocating their political interests regarding the Grassy Mountain

mine, using qualitative analysis. Interview findings are presented and provide a deeper understanding of individual experiences, as well as the nuances behind key themes that from the social media analysis.

Chapter 6 provides a summary of the work completed and discusses the overall findings in relation to existing literature. As the conclusion of this work, Chapter 6 presents an exploration of the value of this research, key take-away messages, the limitations of this research approach, and opportunities for future work.

Chapter 2: Literature Review

How we think about, describe, and execute social movements as a collective has changed immensely since it was first formally theorized, largely due to the wide spread of the internet and increased use of digital media (Tufekci, 2018). Though the current frameworks being used to study the nature of social movements and mass online communication are build upon existing schools of thought within political sociology and communication studies², these ideas are rooted in concepts spanning numerous other disciplines including philosophy, industrial relations, psychology, and gender studies.

This chapter discusses literature containing existing frameworks, approaches, theories, and examples used to better understand the forces that shape how both individuals and networks come together to achieve a shared goal, especially in during times of uncertainty. The chapter is divided across four distinct sections. Section 2.1.0 takes a deep dive into exactly what collective action entails it and how this has evolved throughout the past century. Section 2.2.0 discusses the construction of the online realm and how technology shapes both collective and individual behaviour on social media. Section 2.3.0 gathers concepts from the previous two sections and expands on them under the context of the Covid-19 pandemic, specifically how the use of space and social movements have changed since the onset of restrictions in March 2020. Finally, Section 2.4.0 presents gaps within the literature examined and situates my research within current perspectives and existing theory on social movements in a digital age.

² Such as Bennett and Segerberg's connective action (2012) and Milan's cloud networking (2016).

2.1 Collective Action

2.1.1 Definitions

To frame the discussion, it is necessary to clarify some of the language being used in current literature. With terms such as collective behaviour, collective action, protests, mass movements, social networks, and social movements in use, it can be difficult to understand the differences and nuances around each definition. Clarifying the meaning of these terms allows us to understand, describe, and apply concepts around online political action with greater precision, especially since much of this language is used interchangeably in literature. This section opens with an overview of the terminology and tools commonly used to understand collective action. The construction and evolution of these frameworks is then explored through two sections- collective action prior to the use of the internet and collective action after the internet became widely available.

Collective behaviour (sometimes used interchangeably with collective action) is the umbrella term used to refer to any group behaviour which is done voluntarily and is not regulated through institutional means. Little and McGivern (2016) break this down into four subcategories including (1) crowds, (2) the public, (3) masses, and (4) social movements. Crowds are defined simply as a large number of people in close proximity to each other- this can include anything from a busy bus to a wedding to a protest. Protests or riots fall under this category and are described as acting crowds, where individuals gather with shared intentions or goals. Contrary to crowds, both the public and masses refer to a dispersed group of people who remain anonymous to each other. However, members of the public are united through a shared idea on a specific issue, whereas members of masses are united in a common interest. For example, libertarians would constitute a public and fans of the same musician would constitute a mass. Little and McGivern (2016) note that

a defining characteristic of masses is the inability of individuals within the mass to act in an organized way to achieve a shared goal.

Social movements (also referred to as mass movements) also fall under the category of collective behaviour and can broadly be defined as "purposeful, organized groups striving to work towards a common social goal" (Little & McGivern, 2016). This definition of a social movement is deeply contested, with debates on how to best define the phenomenon. McAdam (1999) defines social movements as "rational attempts by excluded groups to mobilize sufficient political leverage to advance collective interests through noninstitutionalized means" (pg. 36), while Tilly and Tarrow (2015) describe them as a "sustained campaign of claim making, using repeated performances that advertise the claim, based on organizations, networks, traditions, and solidarities that sustain these activities" (pg. 11) with the government involved as a claim-maker or a target of claims. It is worth noting that in reality, none of these terms are perfect, as social movements are a complex concept that can hardly be defined in one sentence (Kolers, 2016). For example, the shortcomings of these terms are evident when attempting to describe collective action within the context of the internet. The difference between a protest and a social movement becomes increasingly ambiguous when describing individuals coming together in virtual spaces, such as the #BlackoutTuesday movement spanning social media networks on June 2, 2020 (Griffin, 2020). From here on, social movements are used as an overarching term for any collective efforts towards a shared goal whereas protests are used to describe specific and intentional action within a social movement.

Frame theory (also known as framing or collective action frames) is a helpful tool for understanding organizational motives behind collective action. First applied to mass movements by Snow and Benford in 1992, a frame is an "an interpretative schemata that signifies and condenses the "world out there" by selectively punctuating and encoding objects, situations, events, experiences, and sequences of action in one's present or past environment' (Snow & Benford, 2000, pg. 137). Collective action frames offer a structure to make sense of various individual, group, and organizational interests, as well as the relationships between them. On an individual level, this structure plays a key part in changing the perception of an issue as one that is beyond control of the individual to a grievance which can be changed (McAdam, 1999). On an organizational or collective level, framing is vital to the direction a social movement takes. Organizing bodies frame issues in a way which is most beneficial to their overarching goals and adopt strategies for action a based on these guiding frameworks.

2.1.2 Theoretical Roots of Social Movement Theory

The action of individuals coming together to advocate for a shared interest was initially described in the late 19th and early 20th century in the context of riots. Written within the backdrop of western philosophy's postulations on rational beings and free will, early collective behaviour scholars observing rioting individuals concluded that even though each individual is a rational being on their own, when these individuals form a collective, they get swept up in the whims of the crowd and become irrational actors (Blumer 1969; Le Bon 1895). These early theories of collective behaviour were heavily critiqued on the assumption of individuals as irrational actors and have been replaced by the more contemporary understanding that individuals remain rational actors, whether or not they are part of a collective movement (Olson, 1965; Gamson, 1989). This body of literature has since evolved to better account for the multitude of factors that influence and shape what collective action looks like. Theoretical contributions to this topic can be understood through four key frameworks which shaped (and continue to shape) perspectives on social movements including rational action theory, resource mobilization, political process theory (also referred to as political opportunity theory), and cultural cognitive theory (Millward & Takhar, 2019), each of which is examined below.

Some of the first bodies of works that challenged earlier schools of thought are based in rational action theory, including *The Logic of Collective Action* by Mancur Olson (Olson, 1965; Coleman, 1973). *The Logic of Collective Action* differed from previous theory as it emphasized that the primary role of collective action is to further the common interests of groups of individuals, meaning that individuals become part of the collective due to their own rational decision making. Olsen (1965) explores the difficulty in creating effective collective movements due to the incentive for individuals to free ride (the collective action dilemma), where they can reap the rewards of the action without necessarily having to contribute to the cause. Early rational action works were written primarily through observations of economic-labour movements, especially in post-industrialization contexts in which organized labour organizations were just emerging.

Resource mobilization theory as we know it works upon the idea of rational actors but describes the success of mobilization efforts on the resources (money and labour) a movement possesses and asserts that collective movements cannot occur unless sufficient resources are available (McCarthy & Zald, 1977). This emphasizes the importance of social movement organizations (SMOs) as a basis for mobilization by providing resources and organizational structure to movements, which dominated the social movement landscape well into the early 2000's. This observation shifted views on social movements from informal, spontaneous gatherings of selforganizing individuals to a more formal, organized entity mediated by an organization, often with paying members (Caniglia & Carmin, 2005). In this theory, the SMOs themselves are viewed as rational actors on their own accord. An example of these types of organizational structures is the Sierra Club protesting dam development in the Grand Canyon in the 1960's or EarthFirst! organizing to protest deforestation in 1980 (Dreiling & Wolf, 2001). Due to the central role of SMOs in resource mobilization theory, the vast majority of this scholarship focuses on organizational aspects, how resources are obtained and used, and the general SMO structure

(Schaefer Caniglia & Carmin, 2005). Since the success of a movement is dependent on the resources an SMO is able to obtain, how the organization frames the issue it seeks to address must resonate with not only its existing members but also prospective recruits which can provide additional resources (Snow & Benford, 2000). Framing the contested issue is a dynamic process that requires the SMO to find the fine balance between member interests and external resource holders, such as donors (Zald & Ash, 1966).

An alternate approach to examining collective movements without foundations in resource mobilization theory is the political process or political opportunity theory, which focuses on external political factors that shape collective movement (Tilly, 1978; McAdam et al., 2001). Political process frameworks claim that large-scale social movements can be explained by external political opportunities. Individual, group, and community actions are shaped as a response to the external political climate and the political leverage that this setting provides for their cause through noninstitutionalized means. As an extension of this framework, Tilly and Tarrow (2001) describe the contentious politics model as the "intersection of the familiar social processes of contention, collective action and politics" (pg. 7). Moving beyond political opportunity theory, the contentious politics model specifically describes of the role of governments in the processes of collective actioncontention becomes political when it affects government interests or agents, due to the government's responsibility to regulate protests and power to either repress or facilitate collective claim-making (Tarrow, 1998). Some critiques of this approach question whether the government needs to be involved for collective action to be considered a social movement under the contentious politics definition and the observation that not all power lies solely within the government but is often dispersed among various centers (Armstrong & Berstein, 2008; Kolers, 2016).

The final major development in collective action theory is the cultural cognitive model, which broadly refers to theories which argue that collective movements cannot be understood without their social and cultural contexts, referred to as 'identities' (Gahan & Pekarek, 2013). This theory rejects approaches based in structuralism and rational choice as a comprehensive model to describe social movements, including political process theory and resource mobilization theory (Melucci, 1996). Cultural cognitive models focus on how framing can shape the ways in which meaning is constructed within collective movements and the influences of cognitive attributions (culture, values, beliefs, ideas) (Klandermans & Goslinga, 1996; Caniglia & Carmin, 2005). A subset of this lens is Habermas' new social movement theory (1981) which is rooted under the premise that it is not internal resources or political climate that distinguish a movement, but the conflict which it is built upon. Working largely through observations of shifts away from 'old' labour movements, Habermas suggests that 'new' movements organize around lifestyle, accountability, and identity as opposed to material demands (Habermas, 1981; Edwards, 2004). Habermas describes 'new' social movements as a "defense of existing lifestyles" (pg. 116)- where the principles of rationality used to examine economic-labour movements are extended into new areas of life (Edwards, 2004). This theory also has implications for school of thought on surrounding public and private space scholarship, extending to modern movements such as environmental or feminist movements, among others.

2.1.3 Social Movements in an Era of Digital Communication

Though the classic theories discussed above provide an important foundation for thinking about collective action, they can seem outdated when applied to the social movements we observe today. A large component of this shift is due to the internet- in the last few decades, digital media has become more accessible than ever before with profound impacts on how we learn, communicate, and interact with the space we exist in (whether this be physical or online). Protests such as the Arab Spring, los Indignados, and the Occupy movement embody these changes- the late 2000's to early 2010's mark the integration of digital media technology with social movements (Tufekci, 2018). The central point of this transition is the role of communication tools and digitally mediated networks in collective action (Bennett & Segerberg, 2012; Kavada, 2015). With social media platforms, websites, and other digital media making connecting with others accessible at a minimal cost, previous explanations of social mass movement behaviours and trends fail to account for these rapidly changing phenomena. For example, social movements or protests often occur entirely online without an official organization at the forefront, such as the #MeToo movement (Suk et al., 2021). This means that describing contemporary social movements that utilize the online sphere require thinking beyond just the role that formal resources and political opportunity play.

The idea of connective action, coined by Bennett and Segerberg in *The Lagic of Connective Action* (2012), integrates various aspects of communications theory with previous knowledge of collective action to expand social movement understanding within the context of social media. The cornerstone of this logic is the shift away from the centralized, organizationally mediated action frames of the late 1900's and the emergence of more personalized communications and as a result, more personalized action frames. Collective action that occurs through digital space (such as social media platforms) results in highly individualized action frames reflecting individual wants, needs, grievances, and hopes (Bennett & Segerberg, 2012). These messages are easy to customize and encourage users to contribute to the collective by sharing their own personal experiences with the greater collective. Personalized action frames are more flexible than traditional action frames revolving around a formal, central organization with a higher requirement for participation, giving more space for self-expression and making movement membership more inclusive of personal reasons behind political involvement (Bennett & Segerberg, 2012). The "we are the 99%" slogan of the Occupy movement is an example an easy to personalize action frame- a relatable, catchy slogan that gave individuals space to offer their personal experiences.

Other differences exist between collective and connective action on both an individual and organizational level. For individuals, the logic of connective action is the process of situating personal ideas, beliefs, and actions within an overarching political purpose as a contribution to the common good, whereas collective action models generally view social networks and relationship building as one of many informal conditions required for centralized mobilization, but not the essence of a social movement (Bennett & Segerberg, 2012). On an organizational level, the difference between the two logics stems from the replacement of formal organizations as the organizing agent with technological processes, leading to the formation of a decentralized network with various content creators as opposed to one or two central organizations, meaning that there is (usually) no clear movement leader (Bennett & Segerberg, 2012). Instead, emphasis is placed on the co-production/distribution of content by numerous individuals who are driven to contribute through personal and self-motivated reasons (Milan, 2015). The high requirement for frequent, selfmotivated participation within online networks forges connections between individuals or groups who previously did not know of each other or existed on the periphery of each other's networks, also known as the formation of 'weak tie networks' (Kim et al. 2020). Weak-tie networks refer to the observation that within social networks, the connections that tend to be most influential in spreading information is acquaintances, as opposed to close ties like friends and family (Granovetter, 1973). The use of digital media as the organizing agent in combination with self-motivated and loosely coordinated membership removes many traditional resource demands of social movements such as network building between organizations, member recruitment, and issue framing (Bennett & Segerberg, 2012).

Bennett and Segerberg (2012) classify three distinct logics of action and place them on a spectrum which also acknowledges a hybrid model (Figure 2.1). Though it can be tempting to view this spectrum as the degree of digital media use in each collective action framework, the true



Figure 2.1. An overview of Bennett & Segerberg's (2012) model of connective action, adapted with examples from recent social movements.

determining factor of where a movement falls on the spectrum in Figure 2.1 is the central organizing agent. The Extinction Rebellion is a good example of this distinction- though the movement itself has a variety of online components such as a website with educational content, a strong social media presence aimed at humanizing the issue and assembles by creating events on platforms such as Facebook or Instagram, the central organizing agent of this movement is the Extinction Rebellion organization itself. True to the collective action model, the emphasis of this organization is on membership recruitment, member engagement and education, and meeting up as a collective both online and physically (Corry & Reiner, 2021). As such, though at first glance it may seem to fit the hybrid typology, when looking solely at the role of social media as an organizing agent it is clear that this movement leans heavily towards the collective action framework, as opposed to other movements such as #MeToo or #BlackOutTuesday (Kountouris & Williams, 2023).

On paper, the ample differences that exist between connective and collective action make it seemingly simple to classify movements. In reality, social movements are not linear processes, and these categories are by no means mutually exclusive or static, with layers of meaning that is constructed and deconstructed simultaneously in multiple ways (Ginneken, 2003; Shafi & Ran, 2021). Bennett and Segerberg's typology and coining of connective action is effective in highlighting how the rise of digital media has impacted mass movements, but various criticisms and modifications of the logic of connective action have risen as the field of technological communications expands. Criticisms of Bennett and Segerberg's work include an over-emphasis on self-organizing of people as opposed to the role of social media as an organizing agent and the dismissal of the "politics of the platform" (Milan, 2015, pg.2). To fully explore the dynamic and complex nature of today's social movements, these critiques beg for further examination- how do the properties of this novel organizing agent impact collective action? Is this online space an impartial and apolitical one? Or does it come with its own biases, boundaries, and rules? If so, how do these rules shape how we interact with and navigate these spaces as both collectives and individuals?

2.2 The Online World

If you are an internet user in some capacity, the answers to the above questions regarding the logic of connective action raise may seem fairly intuitive. The most popular networking sites, notably Twitter and Meta Platforms (parent company of Instagram and Facebook), are marketed as an open, accessible space to connect with others at your convenience (Milan, 2015). While this may be true to a certain extent, the reality that earlier social media scholars such as Bennett and Segerberg (2012) fail to account for is that these spaces also are proprietary, centralized, and have internal motives beyond just the facilitation of online socializing, putting social media platforms in a unique position

in the chasm between the public and the private (Burkell et al. 2013). To add to its complexity, the online sphere is undergoing a constant and dynamic process of meaning-building which brings content from an imagined space into a real one (Milan, 2015). This section begins with an examination of how this techno-social realm can best be understood using existing work on the concept of space, then continues on to explore existing perspectives on how these technological structures intersect with the very human process of building collective identities. Finally, trends in individual interactions with political content in the online sphere is examined, with a specific focus on how the issues discussed are navigated on a micro level.

2.2.1 Where is "Online"?

To understand how individuals come together for shared political interests via social media, it is important to first examine some core concepts around sense of place. The thirdspace (or third space) is a term coined by Bhabha (1994) and brought to life in the works of Edward Soja. Soja's work was inspired by Lefebvre's (1974) interdisciplinary approaches to the discourse of spatiality, namely the "triple dialectic" (p.7) of spatiality, historicity, and sociality (Soja, 1998). The thirdspace represents a simultaneously real and imagined, "constantly shifting and changing milieu of ideas, events, appearances, and meanings" (pg.2) that space holds (Soja, 1998). It can be understood as an assemblage of physical space, the meaning we give it, and the meaning which emerges from humans interacting with and within the space throughout time (Gamson, 1990; Soja, 1998). To anchor the abstract nature of this concept in the 'real' world, let's take (an extremely simplified) look at the various layers of meaning that a national park would hold. The park itself is the physical land which it is on- the one that can be highlighted on a map. But this neatly packaged bundle of land is also much more than that- it is the meaning we assign to it as individuals, communities, and collectives. On the most superficial level, society generally understands it is an area which serves the purpose of conservation and recreation. Beyond this, it is a space inseparable with the vast cultural meaning it holds to the Indigenous communities- long before it was colonized and legally defined by a rigid settler system, simultaneously making it a politicized space (Carroll, 2014; Kelly, 2011). On an individual scale, it is a wedding, a business retreat, a first date- a place of connection that everyone leaves with a slightly different experience shaped by their past and present. This is a simplified example of what Soja describe in his writing- a series of tangible and intangible processes of meaning building that are both dynamic and abstract, but at the same time also very real.

The non-linear nature of thirdspace is one that challenges existing binaries³ and opens up the "Other" - a space beyond an inflexible dualism which allows room for concepts defying the rigid boundaries. In relation to thirdspace, Soja (1998) describes the Other as:

"a meta-space of radical openness where everything can be found, where the possibilities for new discoveries and political strategies are endless, but where one must always be restlessly and self-critically moving on to new sites and insights, never confined by past journeys and accomplishments, always searching for differences, an Otherness, a strategic and heretical space "beyond" what is presently known and taken for granted." (Soja, 1998, pg. 35)

This poetic account of Otherness captures the marginality and radical nature of thirdspace as an Other while remaining deeply applicable as we examine the relevance Soja's works on space and place hold today. Reading the above quote today without previous knowledge of Soja's work, one might easily mistake it for a description of existing in online sphere. Much like thirdspace, the ambiguity and flexibility of the online sphere opens space for the Other.

³ Some examples of concepts commonly presented as binary include subject/object, past/present, nature/culture.

The observation that many attributes of Soja's thirdspace parallel current conversations around the online sphere is no coincidence, given that the ability to connect through digitally networked media is the very act of collectives creating meaning through performance and interaction (Gamson, 1990). Though the internet contains various elements and types of platforms, my focus is specifically on social media. As described by Edirisinghe (2011), social media sites are designed to be "one platform which can be customized yet universally defining similar functions, ambiguous but with one clear intention- connectivity" (pg. 401). The dynamics and conceptualization of the online world on paper makes much more sense when situated in broader knowledge of spatiality and its meanings. Tangible elements of social media are the visual elements of the interface (and the code itself) and intangible ones are the connection and access that it provides to others (Latour, 2005; Edirisinghe, 2011).

Where a binary understanding of space struggles to adequately describe social platforms, the conceptualization of social media through Soja's thirdspace excels. Milan (2015) applies these concepts to mobilizing via social media platforms through the term 'cloud networking' and 'cloud protesting'. Building off the critiques of Bennett and Segerberg's connective action, 'cloud protesting' acknowledges that digital collective action is a sociotechnical process as opposed to just a communicative or sociological one- meaning social media shapes interactions instead of just facilitating them (Milan, 2015). The cloud represents an imaginary space in the sense that much like a phone conversation, the discourse between individuals does not exist solely on one device or the other, nor does it exist entirely in the mechanics responsible for the technical aspects of communication (Milan, 2015). It is intended to be a framework which simultaneously emphasizes the organizing abilities of social media, the sense-making efforts of individual activists, and the technological processes which shape how online interaction occurs. The cloud hosts a set of symbolic 'ingredients' which enable users to form personalized narratives, identities, and networks.

Looking back to original collective action theory, these 'ingredients' or 'soft' resources described by Milan contrast the hard resources described by early social media scholars, such as money and labour which are no longer significant barriers for social movements organizing through social media (McCarthy & Zald, 1977; Milan, 2015). Though the 'ingredients' are immaterial, they are enacted and conceived through a symbolic negotiation between users in both online and physical space and brough to life through the material infrastructure of social media platforms (Milan, 2015). The concept of 'cloud networking' establishes that social media is a political space, enacted through an identity-building process entangled within the symbolic and the material.

Though literature describing space and its theoretical parallels to the conceptualization social media provides insights into the relationships between physical and digital space, this is a complex concept shaped by a multitude of factors. Place attachment theory explores how people experience attachment and build identity upon a given space (Proshansky, 1983; Rajala et al., 2020). This is composed of two elements- the attachment to a place and the identity associated with a place (Proshansky, 1983). A large segment of this research focuses on landscapes and outdoor spaces, with factors such as length of exposure to the space, knowledge, social connection, and exposure to natural landscape all shaping the relationships formed between people and spaces (Brown & Raymond, 2007; Matarrita-Cascante, 2010). This is a robust set of concepts that though applicable in many ways, require analysis and attention beyond the scope of this thesis, especially when examining attachment and identity that transcends both physical and digital space. As such, an in-depth exploitation of the Grassy Mountain Mining Project and place attachment within the context of online social media activism is best left for future research.

2.2.2 Techno-social Dimensions of Social Media

Social media is marketed and positioned as an equal and accessible space that provides opportunities, but the structures shaping how content is presented on social media often tell another story (Heyman & Pierson, 2015). Described as a 'platform society', today's online networking climate is a global conglomerate of various platforms that though independent in structure, exist under the common mechanisms of the web (Gillespie, 2010). A handful of global mega-corporations such as Google or Meta Platforms retain access to huge amounts of data on user demographics and behaviour, meaning the structure of the online sphere has implications for 'real life' (Helmond, 2015). With nearly all societal actors present online and intertwined in the ongoing online negotiation of meaning-building, the far-reaching nature of these platforms has an impact on politics, economies, and culture which forces all social actors to, either consciously or unconsciously, recalibrate their position in public space (Dijck & Poell, 2015). It is known that the design of physical spaces affects how people use it and how deliberation occurs in that space, and this extends to the material structure and design of digital networking platforms (Drucker and Gumpert, 1996; Milan, 2015). However, rapid technological advancements in combination with a general lack of transparency on behalf of corporations makes it difficult to make conclusive statements on the techno-commercial impacts of social media platforms on online behaviours.

Existing concerns include the centralized and profit driven nature of platforms, opaque and proprietary nature of algorithms, and the resulting privacy concerns (Heyman & Pierson, 2015; Milan, 2015). The algorithms utilized by networking platforms are the programmed, invisible forces which form the rules of interaction within a platform, with the intention of keeping users engaged though rapid turnover and the sorting of content based on past-use data (Milan & Velden, 2016). Algorithms create and reinforce 'weak tie networks', as users are more likely to see types of content
shared by their indirect network that aligns with their interests and motivates the user to continue interacting with the content and hence, the platform (Zhong, 2022). Not only are social media platforms brokers between individual interaction, but they also gain profit by mediating interactions between the consumer and the seller, generally through paid advertising (Heyman & Pierson, 2015). Ultimately, algorithms are commercial mechanisms whose proprietary nature offers little information about the logic or mechanisms behind its structure.

Specific to collective identity building, the impacts of this techno-commercial structure are evident in the algorithmic encouragement of recurring interaction and engagement. Properties of social media are designed for rapid content turnover, constantly introducing novel elements into individual meaning-building efforts (Milan, 2015). The positive effect of this infrastructure is the reduced cost of mobilization through increased access to information and decreased barriers to participation in political discourse (Bennett & Segerberg, 2012). It also means that individuals can easily customize and personalize their narrative, that the collective then 'votes' on by selecting which content to emphasize and share (Milan, 2015). This collection of personalized content and interactions with accounts of other individual's experiences is the basis of digitally mediated collective action. However, it also results in a need for the re-enactment of social action through constant negotiation and reinvigoration of identity (Edirisinghe, 2011). The rapid cycle of content production and interaction favors engagement over sustainability, resulting in the short life cycle of online movements which makes performance and participation a key aspect in the visibility of online political action (Edirisinghe, 2011; Milan, 2015). Another major concern with the structure of social media platforms as facilitators of social movements is the formation of echo chambers, as individuals generally seek out content which they are already interested in which the algorithm further reinforces by showing content related to those specific preferences, resulting in views being reinforced rather than challenged (Heyman & Pierson, 2015). The fear is that this leads to the

polarization of political conversation online, leading to decreased deliberation. It is also important to note that the majority of research on the topic of online echo chambers has been conducted on a macro scale which may not be representative of the realities of individual conversations online (Vitak et al., 2011; Greijdanus et al., 2020).

These points raise an interesting question about the nature of collective action online- how closely does individual behaviour mirror that of a collective when it comes to online action? Early collective action scholars, Bennett and Segerberg (2012), Milan (2015), and others all agree that regardless of the mediating technology or organizations, individuals are the core of the formation and meaning-building efforts of collective actions. However, we have yet to discuss the individual perspectives and behaviours around political deliberation online that contribute to the collective and how those contributions are defined.

2.2.3 Political Participation Online

Collective action and the organization of movements is a robust topic spanning numerous disciplines. As we've seen so far, social movements as a whole are adaptable to different environments, social settings, and forms of communication. In the field of collective action theory, the discussion of individual behaviors, attitudes, and motivations that contribute to the organization of a collective has been largely neglected by scholars. Specific frameworks, such as Milan's "cloud protesting" (2015), have intentionally left space for understanding the individual but still keep to a fairly high-level analysis. Close up, the understanding of online political participation is much less straightforward. We begin with a discussion surrounding the meaning of variables most commonly looked at when describing online behavior, including political participation more broadly and concepts such as 'slacktivism'. Once the semantics have been established, factors that make a specific individual more or less likely to participate in online political discourse is examined, as well

as trends in individual online political behavior specific to Facebook. Finally, connections are made to previous theory to examine how individual behavior oscillates between the physical and online spheres.

What does it mean to participate in online politics? Some individuals may consider political participation to be sending a link to a Facebook friend without having any other interaction with the content, while others may classify this as far from genuine engagement with the subject. At its very broadest, political participation is defined simply as "citizens' activities affecting politics" (pg.2) (van Deth, 2021). This definition, however, does not offer much support in today's reality where increased access to information, decreased borders between the public and private sphere, and general higher education levels result in the expansion of political participation (van Deth, 2021). Various scholars have attempted to redefine political participation in ways which are more suited to today's society, but the broadness or specificity of such definitions can sway data analysis and findings (van Deth, 2021). Specific to online political participation, a 2021 systematic literature review finds that only 32.4% of publications on the topic of online political participation have definitions of the term that refer exclusively to the online sphere, with 30.5% of publications referring to online and offline participation jointly (Ruess et al., 2021). Additionally, less than onefifth of definitions in literature account for passive activity, though evidence suggests that passive attention to political content, such as reading news articles, is closely linked to information dissemination action and it the more passive levels of involvement that are commonly the beginnings of individual involvement (Cantijoch et al., 2016; Boulianne, 2019). Van Deth (2021) suggests giving political participation an "operational definition... specifying the exact properties that are required to determine its existence" (pg.6) and puts forth a decision tree to help determine the type of political participation being conducted. This model prioritizes the acting aspect of political action, but also leaves space for intentions as a determining factor. One critique to note is

that this model does not account for non-action, such as deliberately not purchasing unethically sourced products online. However, this is the beginning of a greater debate beyond the scope of this thesis.

When discussing political participation as a collective, concerns of superficial or ineffective contributions to a movement or cause are often voiced. The fear of freeloading and other "feel-good forms of political participation" (Vitak et al., 2011, pg. 6) is especially prevalent in the world of online activism and has come to be known by the term 'slacktivism'. In literature, dismissal of online mobilization as 'slacktivism' or of lesser value than physical presence at a protest is primary voiced by pre-internet scholars but still occasionally echoed in more recent works (Vitak et al., 2011). The majority of evidence suggests that overall, there are primarily positive relationships between online and offline activism as well as increased visibility for the cause through social media use (Tufekci & Wilson, 2012). It is important to note that this evidence is specific to activism non-repressive contexts and under different regimes, increased visibility online may lead to further repression and thus, a negative relationship between online and offline activism (Greijdanus et al., 2020). A subsection of research finds neither a positive nor negative relationship between online and offline activism due to individuals restricting their actions to one domain or acting differently depending on the context (Greijdanus et al., 2020). However, little research supports a direct trade-off or between online and offline activism (Greijdanus et al., 2020). Recent evidence does not support digital dualism and instead demonstrates that individual relationships between the online and offline sphere are complex and multi-faceted (Milan, 2015; van Dijck & Poell, 2015; Greijdanus et al., 2020). This means that the research surrounding the relationships between online and offline participation is highly contextual, which partly accounts for the high degree of variance between conclusions regarding the effectiveness of online political participation.

With individuals being the unit of a social movement, the emotional state and intensity of the person generating online content is a major driving force in online discussion and participation, for both themselves and those interacting with their content (Chmiel et al., 2011; Garcia, 2016). When encountering online content which evokes emotional arousal, the user is more likely to participate in online discussion, regardless of whether this emotion is positive or negative (Garcia, 2016). However, in a thread-based social media environment⁴ both the type of content the user contributes and the length of the interaction between users engaging in discussion, is primarily influenced by the valence (perceived intensity) of their emotions (Chmiel et al., 2011; Garcia, 2016). Chmiel et al., (2011) found that though the length of a thread is largely dependent on its mean valence, all threads tend to conclude at a similar emotional level regardless of their length.

Individuals being more likely to engage in online spaces when they feel strong emotion has strong ties to collective action, especially when emotion is evoked through shared injustice. A metaanalysis of 403 papers, conducted by Agostini and van Zomeren (2021), found that politicized identities, efficacy, the emotional experience of injustice, and moral conviction had the strongest positive relationship with collective action on an individual level. Politicized identity refers to identification with a certain movement or the perception of oneself as an activist. The development of a politicized identity is strongly linked to moral conviction, especially when it is tied to tangible outcomes (Agostini & van Zomeren, 2021; Judge, 2022). Often the emotional response to injustice is based on a trigger event, with "felt injustice" (p. 686) being a stronger motivator than perceived injustice (Agostini & van Zomeren, 2021). This is also consistent with the observation that higher emotional valence and arousal leads to increased online participation. Though the factors discussed play a significant role in understanding the emotional and identity-based motivators of collective

⁴ For example, Twitter discussions, Facebook or Instagram comments with replies, or discussionboard style platforms such as Reddit.

action, it is worth noting that many other variables shape collective action and are highly context dependent.

2.3 Covid-19

This section begins with a brief overview on how individual online behavior and use of physical space has shifted since the onset of the pandemic, then continues with a discussion of the impact Covid-19 had on social movements.

2.3.1 Interactions with Space During Covid-19

On both an individual and a community scale the pandemic resulted in an inevitable shift in how we use our space, interact with others, and make our voices heard, as well as the medium through which these actions occur. Nearly overnight digital devices became central to most aspects of social connection with dates, parties, family get-togethers, and other activities being hosted in virtual spaces (Beech, 2020; Lovari & Bowen, 2020). In addition to socializing, much of the time spent online was dedicated to keeping up with current events and pandemic updates, with 'quarantine' ranking amongst the most common globally searched terms during times of severe outbreaks (Brückmann & Martinez, 2022). As a result, the average internet user experienced a 50-70% increase in online activity, amounting to a screentime of five hours per day, with at least half of this time spent on social media (Beech, 2020; Balhara et al., 2020; Dienlin and Johannes, 2020; Vanderloo et al., 2020). Though this led to a wave of concern regarding the impact of screentime on well-being, it is difficult to draw conclusions on well-being at such a complex time. Little research has been done specifically during Covid-19 on screen use and health, but pre-pandemic work generally finds no impact to a moderate impact on personal well-being, which is generally dependent on the perception of individual users on their engagement and types of online activity, as opposed to time spent online (Hancock et al., 2019; Liu et al., 2019). A 2020 study done by van Bavel and their colleagues examined the impacts of social media use during Covid-19 and found a positive relationship between social media use and social connectedness, capacity-building, and acquiring knowledge. This is supported by qualitative findings, with ~40% of individuals reporting a stronger sense of community and stronger ties to friends and family during lockdown periods (Macmillan, 2020). Similar to individual trends in online behavior during the pandemic, governments also pivoted to greater use of the online sphere for public opinions on decision-making through the use of online discussions, forums, polls, and other tools (Chwalisz, 2021).

With such a dramatic transition into the online sphere, what was going on in physical spaces throughout the pandemic? One major shift that occurred is how people view and use natural spaces. Data emerging from the first six months of the pandemic shows that the usage of natural spaces increased in proportion to Covid-19 cases (Lu et al., 2020). This shift also impacted perceptions of natural spaces. When asked about the state of the environment during lockdown, people reported observations that the environment around them is cleaner or that natural spaces and their inhabitant are recovering with decreased human use. For instance, 51% of UK residents said they noticed cleaner air and 27% noticed more wildlife during lockdowns (Macmillan, 2020). Though this may be true in the short term or under observable means (people see more wildlife due to factors such as decreased traffic) there is little data to support the long-term environmental benefits of Covid-19 mandates or significant contributions to climate change, resource extraction, or conservation methods (Rume & Islam, 2020).

2.3.2 Social Movements During Covid-19

Another major outcome of the pandemic was the magnification of existing injustice surrounding the use space and who is free to move between different spheres during a major health crisis. Soon after restrictions on mobility were put into place, it became evident that the ability to transition important aspects of life online for personal safety is a privilege. More severe health outcomes are tied to economic status, race, sex, gender, and other social determinants of health (Government of Canada, 2021b). Various studies show that lower income individuals continued to move around during the pandemic to access work, while higher income individuals were able to limit their movement as they were more likely to transition to remote work (Valentino-DeVries et al., 2020). As a result, it was more difficult for lower income workers to protect their personal safety and if they became sick, lack of job security or health benefits made the financial repercussions of contracting Covid-19 much more severe (Valentino-DeVries et al., 2020). Racial minorities faced disproportionate challenges- including racism due to misconceptions about the pandemic, fear of abuse by authority figures with increased surveillance methods, exacerbated systemic healthcare inequities, and beyond (Addo, 2020, Bhui, 2021; Stahl, 2021). Use of domestic abuse helplines and reports went up around the world within the first few weeks of Covid-19 (Townsend, 2020; Usher et al., 2020). Clearly, how space is used and who has access to it has a major impact on well-being.

Dozens of social movements of varying size and impact have occurred within the backdrop of the Covid-19 pandemic. In Canada and the US, the Black Lives Matter protests, #EveryChildMatters, the Freedom Convoy, and various other smaller political movements dominated the online sphere and directly impacted policy surrounding the issues in question (Carothers & O'Donohue, 2020). Every one of these movements was composed of a novel blend of online and offline action that enabled participants to feel connected to a political cause without compromising their safety (van der Zwet et al., 2022). This also means that each movement took up space in unique ways- from the blaring physical occupation of space on behalf of the Freedom Convoy to the flood of black squares on social media during #BlackOutTuesday to nearly everything in-between. Though none of these protest tactics are necessarily new, navigating

restrictions on movement during a global crisis through the use of the (seemingly boundless) online sphere places online collective movements in a very new context, especially given the risks associated with large gatherings, subsequent restrictions, and increased policing (Lee, 2021).

Social movements advocating for environmental causes also faced significant shifts in lieu of the pandemic, resulting in changes to uses of online and offline space. The years prior to the pandemic saw a rise in the visibility environment related movements such as the Youth Climate Strikes, Canada-wide mobilization in support of the Wet'suwet'en nation's opposition to the Coastal GasLink project in British Columbia, and the Extinction Rebellion protests. After the outbreak of Covid-19, updates on the virus dominated media. Ongoing environmental movements got pushed out of the spotlight and were driven to change how they performed activist actions, especially activities such as protests and strikes which traditionally require a collection of individuals in a physical space (Rauchfleish, 2020). In line with other social movement trends in 2020 and 2021, Canadian environmental movements relied heavily on online tools to mobilize (Forrest, 2020). To demonstrate the relevance of environmental action and increase the visibility of their efforts, many activists intentionally framed environmental issues within the context of Covid-19 (Rauchfleish, 2020). Though little research exists on these shifts and their effectiveness, a study by Arya (2021) examined the challenges and opportunities Covid-19 has brought to youth climate protests. Interviewees found that that shift online allowed more network building opportunities, participants in rural communities felt more involved, and lifestyle changes due to the pandemic allowed more time to spend on online actions (Arya & Henn, 2021). The challenges voiced in this study capture many of the concerns around online political movements and increased use of technology during the pandemic, including difficulties taking direct action, content getting lost in the quick turnover of social media, online surveillance, and fear of online action being less impactful than physical mobilization (Arya & Henn, 2021). Other, smaller environmental movements that arose during the

pandemic also faced unpredictability from their inception, however, few articles exist on the adaptations of grassroot movements during the pandemic.

2.4 Next Steps

Environmental movements are a unique subset of collective action to examine when studying how political interests in both physical and online spaces are advocated for during the pandemic. Since the central element in most environmental protests is physical space, the observation that most of the action surrounding these movements was done primarily online has the potential to deepen our understanding of the complexity of political spaces in a digital age. From a policy perspective, this knowledge can offer a deeper understanding of the impact online space has on setting the political agenda, whose interests are being voiced through these channels, and the factors that shape how influential their voices are. Insights can also be gained on interactions with physical environments and individual motivations for voicing political interests within that space. Only a handful of literature exists on environmental political activism during Covid-19, with all peerreviewed works focusing on environmental social movements which were well established prior to the beginning of the pandemic.

In pursuit of filling these gaps, the research undertaken in this thesis examines the evolution of an environmental social movement set entirely within the peak of the Covid-19 pandemic and ensuing restrictions. This research offers valuable insight on the intersection of the environment as a political space and the online sphere to advocate for political interests.

Chapter 3: Methodology and Methods

3.1 Guiding Frameworks

3.1.1 Defining Political Participation

As was discussed in Chapter 2, there is no universal definition for political participation, especially when it comes to the online sphere. However, conducting research online often requires a deliberate choice of posts and clear exclusion criteria. Since this project examines online political participation, relatively clear guidelines for identifying political participation are required. An approach developed by van Deth (2021), which is reproduced as Figure 3.1 in this thesis, is used as the guiding work for determining what constitutes political participation. The decision to use this specific approach was made primarily because it encompasses various levels of political participation which systematically account for both action and intention. This allows room to classify different forms of participation which allows for a more in-depth understanding of exactly what political participation online looks like. Within the context of this research, one weakness of van Deth's (2021) operational definition is that it is not exclusive to online political participation. Though valuable to note, this did not present issues in data collection or analysis due to the research designsince the social media analysis was restricted to only posts on Facebook groups, the 'online' aspect of online political participation is difficult to dispute.

The complexity in individual political interactions on social media is difficult to capture, as various factors can aid in predicting both participation trends and the value which individuals extract from online activism. A useful tool for predicting engagement in politics via social media is perceived political efficacy. This can be broken down into three separate sub-types: external efficacy, internal efficacy, and collective efficacy (Halpern et al., 2017). External efficacy is an individual's



Political Participation - non-professional, voluntary activities which are not political but occur in a political context (p.10); Political Participation V - all voluntary, nonpolitical activities by citizens used to express their political aims and intentions that do not fit into one of the previous four types of participation (p.11).

sphere

of

perception of a politician's or government's willingness to respond to their opinions (Esaiasson, Kölln, & Turper, 2015). Internal efficacy is an "individual's perceived capacity to influence the political system" (Halpern et al. 2017, p 322). Collective efficacy is an individual's belief in their capacity to reach a collective goal with others (Bandura, 1997). Haplern et al's (2017) examination of political sharing on social media finds a positive link between high collective efficacy and offline political participation, as well as collective efficacy and internal efficacy. This means that social media users who feel like they are able to successfully advocate for their political interests both individually and as a collective are significantly more likely to engage in political participation offline. This also supports findings that internal and collective political efficacy is strengthened through the discussion of issues with others (Finkel, 1985).

When discussing specific social platforms, Twitter users show higher internal efficacy whereas Facebook users show higher collective efficacy (Halpern et al., 2017). Facebook was also found to have a higher overall impact on offline political participation, which ties into previous observations that ceasing Facebook use reduces acute understanding and knowledge of political news (Mosquera et al., 2020, Allcott et al. 2020). These observations can be attributed largely to the differing structures of each platform, as Facebook networks are generally built through personal connections with real-life 'friends', whereas the social networks formed on Twitter are usually mostly separate from user's offline networks (Halpern et al., 2017). These networks have been linked to increased political participation, likely through increased access to political content (Gil de Zúñiga et al., 2012; Loader, Vromen, & Xenos, 2014). Individuals with stronger 'weak tie networks' on social media are also observed to have higher self-efficacy and political knowledge- factors which correlate with higher levels of offline political activism (Kim et al. 2020). The structural ability to organize 'events' on Facebook also contributes to increased connection to offline political participation, with the specifics of many on the ground protests such as the Climate Strikes or the Arab Spring being organized using this feature (Arya & Henn, 2021; Tufekci, 2018)

The above observations raise a variety of interesting questions, especially within the context of online political participation during Covid-19 restrictions. Evidence suggesting that social media platforms and their specific structure has an impact on various aspects of political participation forms the basis of my research approach. This thesis examines how shifts in navigating space due to the pandemic and the subsequent increase in the use of social media as a communication tool has shaped how individuals advocate for their political interests in the online sphere and how this translates into tangible change within physical space. Through a case study of the proposed Grassy Mountain coal project, I examine how various actors conceptualize and communicate political interests within social movements occurring in both online and offline settings. Within this process,

the structure of social media and its contributions to shaping online social movements is also examined. To best capture conversation within the online sphere, I use a two-pronged research approach that aims to gain an overview of online political participation as well as understand the nuances and impacts of these conversations. This includes a social media analysis of four public Facebook pages and supplementary interviews with individuals that contributed social media content to these pages.

3.1.2 What is a Social Media Analysis?

The existence and extensive use of social media in the public domain has great potential for both commercial and academic purposes. Since 2014, social media platforms have become increasingly popular for obtaining large data sets as they allow for access to both qualitative and quantitative data, including levels of engagement with each piece of content posted, who it reached, positive and negative reactions to the content, and conversations surrounding the topic (Batrinca & Treleaven, 2016). When combined with the rapid content turnover on popular social media platforms such as Twitter and Facebook, it is possible to access comprehensive data on a topic in a fragment of the time it would take using traditional methods such as a document analysis. Beyond this, social media also offers access to dynamic and real-time data, meaning that it is not only the content itself that can be examined but the ways in which this content changes over time or in response to current events. Social media can be open source (such as Wikipedia), entirely private (an organization's Microsoft Teams), or a mix of both (Batrinca & Treleaven, 2016). Popular platforms such as Facebook and Twitter are mixed as they contain both elements accessible to the public and elements that are private (at least to the average user) (Batrinca & Treleaven, 2016). Social media analysis refers generally to the use of public content, making it a data collection method at its core (Franz et al., 2019). This data can be understood through various qualitative and quantitative analytic approaches. Though it offers great potential, social media analysis as a research method also comes with some setbacks- especially when used in an academic (as opposed to commercial) context, as much of the data collected through mixed social media platforms is proprietary and not open (Andreotta et al., 2019). Often, even when social media data is perceived as public, the platform is not designed in a way that the average user can access a full dataset.

Social media data can be obtained in two ways- manual extraction from the platform and the use of intermediary software to access the database where information is stored, which is called application programming interfaces (APIs) (Andreotta et al., 2019). The latter generally requires strong knowledge of programming languages such as Java and is intentionally made difficult to access by the platforms, creating a barrier for many researchers. Third party software exists and can aid researchers in navigating API's by automating the process, called 'data scraping' or 'data mining' (Andreotta et al., 2019; Batrinca & Treleaven, 2016). Even though they are marketed as digital public spaces, most social media platforms are commercial entities and are not entirely open source. They are also quick to flag and ban IP addresses that continuously access content, meaning that extra precautionary steps are required to use this approach (when web pages ask you to confirm that you are not a robot, they are screening for automated processes such as these) (Andreotta et al., 2019). Manual data extraction is just that- manually scrolling through the page and entering the data into another program, such as excel, in the desired format for analysis. A variety of examples of each approach exist within literature, with the appropriate choice of method depending on research goals, access to resources, and the researcher's skillset (Franz et al., 2019). Nearly all quantitative social media analyses benefit from the use of API's to extract data as it allows for quick access to thousands of data points which exist as discrete values (for example, number of likes or number of people the post reached) (Andreotta et al., 2019).

For qualitative approaches, such as a thematic analysis of post content or interactions, it is often more advantageous to manually extract data from the original source within the platform (Batrinca & Treleaven, 2016). Qualitative methods are primarily used to gain an in-depth understanding of the research topic and its narratives, especially in instances where the data is not easily quantified (such as the text of a post, image, or specific language choices in a comment) (Batrinca & Treleaven, 2016). Manually extracting data offers a more intimate relationship with the content and allows the researcher to become familiar with general sentiments before a formal analysis. Additionally, qualitative analysis often requires less data as one data 'point' often contains a wealth of information- one post on Facebook could have a text block, an embedded link, and an entire conversation about the topic in the comments. Though examples of researchers using API's to collect data for qualitative analysis exist (Kent et al., 2016; Xu et al., 2021) and in some situations, such as examining Twitter hashtags, the use of an API may be appropriate, automating the extraction of qualitative content can often be more time consuming in terms of formatting raw data and has significant gaps in the type of content it can recognize (Batrinca & Treleaven, 2016). Since most data used for qualitative research approaches is not discrete, categorizing and formatting data is often a judgement call- automating this by training a computer to make these judgement calls is a technical and time-consuming endeavor.

Though social media data provides a multitude of research opportunities, a variety of downfalls to social media analysis exist. Online social interactions relating to a specific topic are rarely exclusive to one platform and as discussed, different platforms are used for different purposes and by different demographics, specifically in terms of age (Franz et al., 2019). This means that even if all the posts on a social media platform regarding a topic are examined, the reality is that there is much more content spanning other platforms. Generally, a network is created- social media posts link to other forms of media such as videos or websites that users interact with, and often websites

link back to various different social platforms (Bennett & Segeberg, 2012; Franz et al., 2019). This presents limitations in the amount of data a researcher can realistically access- mapping out networks takes years of work by a dedicated team and even then, these networks are always changing as content is created (Franz et al., 2019). As such, most literature tends to focus on one or two specific platforms as opposed to mapping the entire networks, which requires methods that require a deep dive into computer science. When conducting research using qualitative methods within a social media analysis, it can also be difficult to accurately account for various forms of information (Franz et al., 2019). For example, it is common for a single post to contain text, emojis, photos, and links, along with the responses of other users to this content (which also often contain various media forms). Interaction between users is a key aspect of social media, which means that discussion within the comments of a post can provide valuable insight (Franz et al., 2019). However, most research studies to date fail to capture this conversation by examining either only user-generated content or only user-directed content⁵ (Franz et al., 2019). Finally, social media analysis methods are relatively easy to scale, especially when automating the process, but it is difficult to replicate exact methods. This is due to a variety of factors including the dynamic nature of social media, the proprietary nature of data on commercial platforms, and a lack of transparency in current research employing these methods (Franz et al., 2019; Kent, 2016). In a 2019 meta-analysis of social media analysis done on Facebook, the majority of research articles did not describe their methods thoroughly enough to replicate (Franz et al., 2019). This makes it difficult to establish best practices when conducting social media analyses in a research context.

⁵ "User-directed content" is not a widely used term in literature but is used in this thesis to refer to individuals engaging with someone else's content, most commonly by commenting on it.

3.2 Methodology

3.2.1 Positionality

The non-renewable resource extraction industry- specifically oil and gas and mining, is a politicized topic in Alberta. Growing up in the province and supported largely by this industry, I am familiar with this political climate and the nuanced conversations surrounding it. My background in the sciences, current pursuit of an environmental policy degree, and personal interests resulted in much of my post-secondary work being deeply intertwined with these conversations from a range of technical, environmental, and social perspectives. Quickly, I came to realize that finding an individual with no strong opinions on the industry was an exception- everyone (including myself) seemed to have a relatively strong stance on the value of non-renewable resource extraction and its future. My long-time exposure to the topic gives me a strong understanding of the evolution and complex nature of the issue, as well as effective strategies for navigating these conversations.

My relationships and experiences with the space that I exist in are intertwined with the benefits that come alongside my place within it. My views and approaches to research are inherently shaped by the socio-economic position I hold as a white, established colonizer living in an urban region. I also benefit from Alberta's fossil fuel economy, while experiencing minimal environmental risk from the extraction and consumption process. I also have access to a reliable, affordable internet connection, which is not the reality for many individuals around the world.

3.2.2 Justification

The rich political discourse that occurred within the online sphere, the timing of this project, and final policy decision of behalf of the Government of Alberta makes the proposed Grassy Mountain mining project an ideal case study for exploring how individuals advocate for their

political interests in the online sphere and its impact on policy decisions. Given the complexity of how online resources are mobilized in acts of advocacy for the physical space we occupy, this research draws connections between a variety of existing theory through an analysis of online discourse regarding the Grassy Mountain Mine project. Examining these connections and their relationships through a case study of the Grassy Mine project sheds further light on how various aspects of policymaking can be shaped by public participation in online space and the processes through which this occurs. This research also provides insights that can be applied beyond this case study, as questions regarding the ongoing integration, impacts, and governance of emerging technology are becoming increasingly pressing policymaking, academic research, and industry.

3.2.3 Methodology

At its core, the nature of this research is exploratory, as provides a novel examination of the intersection of digitally mediated political participation in both physical and online settings. This analysis of online discourse regarding the Grassy Mountain Mine project draws valuable connections between various realms including the public sphere, political participation via social media, the meaning-space holds across various contexts, and the environment. Due to the context within which the case study is situated, this research also increases the understanding of how online grassroots movements get translated into on the ground activism and policy change in times of crisis. The decision to take an explanatory, single-case approach was made since no prior academic research around the Grassy Mountain Mine activism exist and a wealth of social media data is available to explore this project. In addition, the volume of both textual and quantitative data collected in the social media analysis provides a solid foundation for pursuing a wide breadth of future research related to the topic.

Though a significant component of this project is a social media analysis, supplementary semi-structured interviews were conducted to support findings from the social media analysis. Conversations regarding the energy sector are a controversial topic and more specific prompts during a semi-structured interview were thought likely to give greater depth to the findings than just a stand-alone analysis of social media. Additionally, research by Elder (2020) shows that online comments and responses, especially on public groups, are likely to focus on attacks of character when viewpoints are questioned, which often shut down conversations on online platforms. As such, the aim of the interviews with individuals active in the groups is to have a more nuanced conversation and clarify any points that may be unclear in the social media analysis.

3.3 Methods

3.3.1 Social Media Analysis

The primary data for this project was obtained through a social media analysis, which includes the manual collection and thematic analysis of data from Facebook (this is be guided by the primer written by Franz et al., 2019). Since social media analysis data was collected through a public Facebook group, where all posts are visible to viewers of the page (regardless of whether they have a Facebook account), GC-REB approval was only sought for the qualitative interview aspect of this research. To ensure privacy for the individuals posting on the pages examined, any information that could reveal the poster's personal identity was removed. This includes names, gender, locationspecific information, business names, or any other identifying characteristics.

Data collection and analysis focus primarily on the content posted within public Facebook groups and various engagement metrics of individuals' posts that were made from December 21, 2020 (when the first Facebook page dedicated specifically to the Grassy Mountain Mining project was established) to August 6, 2021 (Grassy Mountain Mines project suspended). Specifically, the groups "Protect Alberta's Rockies and Headwaters" and "Crowsnest pass. Ghost town" are examined, as they were created within the last three years and are dedicated primarily towards expressing interests regarding resource extraction projects specific to the Crowsnest Pass region. Additionally, relevant content from the "Around and About the Crowsnest Pass" and "Crowsnest Pass- Uncensored" public groups is also included in the analysis. Only posts which are directly or indirectly linked to the politics of Grassy Mountain are collected- for example, content describing the general impacts of open pit mining is included in data collection, whereas content regarding mask mandates is not. In instances where uncertainty regarding the relevance of the content exists, the posts were flagged and re-examined in the context of key themes upon the completion of the analysis. For example, multiple posts in the two community pages that are not explicitly about the issue of coal mining ("Around and About the Crowsnest Pass" and "Crowsnest Pass- Uncensored") contained artwork of Grassy Mountain and the Eastern Slopes, without any other political content or any personalized comments. These were flagged during the initial data collection and once a preliminary analysis of all four groups was conducted, it became clear that there was a distinction between posts containing artwork for political purposes (such as the painting of a landscape accompanied with a message describing the artist's intention of commemorating the region before it is, in the perspective of the viewer, destroyed by a mining project) and artwork posted with no explicit political purpose (such as post containing artwork of a landscape to advertise the artist's presence at an upcoming community sale). Though there will always be some degree of judgement to be made in order to enforce the distinction of political versus apolitical posts, applying van Deth's (2021) model of political participation, and re-examining these posts within the context of the whole dataset gives a systemic approach and justification for making these distinctions.

Given the relative novelty of social media analysis as a research method, a variety of additional barriers were encountered in the initial stages of data collection. These were primarily due to the structure of Facebook and somewhat unexpectedly, resulted in valuable insights into the research objectives (see Chapter 6 discussion). A consistent approach to collecting posts across the four groups was required. Scoping research and a test case conducted upon the proposal of this research showed a third-party data scraping software which could automate the collection process to be the most thorough and time efficient method of collecting posts from public Facebook pages. However, upon the time of data collection six months later, Facebook had filed a lawsuit and blocked the collection of their data through the software⁶ (Octoparse). As such, an alternative method of data collection was deployed and posts were collected manually. Due to the high number of posts in the "Protect Alberta's Rockies and Headwaters" public page during the time period examined (an estimated 1600 posts and 33,000+ comments) and an inaccessibly high level of computing power required to load the posts via scrolling, a systematic method for manual collection was created using keyword searches using the Facebook search function. Though time consuming, this ultimately proved to be beneficial, as it allowed the researcher to become familiar with the content prior to analysis and gave a general overview of the themes being discussed. This approach also maximizes data security and avoids breaching any policies Facebook may have put in place against the use of automated programs to obtain and store data in light of the recent lawsuit.

Five keywords related to the topic were chosen based upon an exploratory review of the recurring words that were present in political posts regarding the issue, where searches were

⁶ More information on the legal dispute can be found here:

https://techcrunch.com/2022/07/06/meta-sues-chinese-companys-us-subsidiary-for-scraping-facebook-and-instagram-data/

conducted with variations of keywords related to coal mining and their synonyms, with the conclusion that the five words listed above yielded content matching inclusion criteria most consistently. Word modifiers (such as "-ing", "-ed", "-s") and capitalization did not seem to make a difference in search results. The words "Alberta", "coal", "Grassy Mountain", "mining", and "environment" were individually searched in each group. Within a public Facebook group, various filters can be applied when a keyword is searched, including the year it was posted. As such, a total of 10 keyword searches per group were made (keyword+ year filter). To ensure that posts were being captured proportionately over the timeframe in question; posts from each group were collected for every week between December 21, 2020 and August 6, 2021. Within this process, some weeks had many more posts than others and as such, a cap of five posts per week was established, collected in order of appearance within the search. This decision was made to capture a representative number of posts from each week, while ensuring data collection and analysis remained within the scope of a master's thesis. Data was gathered into numbered "bundles" which consisted of the posting date, type of post (text, link, video, etc.), textual post content, any embedded content with a brief description⁷, post positive reaction count⁸, post negative reaction count⁹, surprise reaction count¹⁰, total reaction count, share count, number of comments (direct comments + comment responses), total engagement (total reaction count + shares + direct comments + comment responses), and the text of the first five direct comments (comments containing only a tagged user were disregarded to ensure adequate data for analysis) (Figure 3.2). The decision of taking the first five comments (as opposed to the top most "liked") was made to get a

⁷ Brief descriptions consisted of 2-3 words describing visual content or the title displayed for links. For example- "photo; landscape" or "link; Alberta awards 11 coal leases in SW Alberta for \$36 a hectare; [hyperlink]"

⁸Positive reactions included "like", "love", "care", and "haha".

⁹Negative reactions included "sad" and "angry".

¹⁰ Surprise reaction includes "wow".

Facebook group		Total (# of posts collected per				
	Alberta	Coal	Grassy Mountain	Mining	Environment	group)
Protect Alberta's Rockies and Headwaters (PARH)	121	123	104	103	98	549
Crowsnest pass. Ghost town (CPGT)	10	13	9	12	0	44
Around and About the Crowsnest Pass (AACP)	1	0	1	1	1	4
Crowsnest Pass- Uncensored (CPU)	2	4	4	4	1	15
Total (# of posts collected per keyword)	134	140	118	120	100	612
ALBERTA "Bundle" Date Type of # posted post	content c (text only) li	imbedded ontent title & nk (if pplicable)	Post Post positive negative reaction reaction count count		Share Number of count comments (direct comments + comment	Direct comments (first 5)

Table 3.1. Summary table: total number of posts collected.

											+ comment responses)					
Apr	96.00	2022-04-26	text	Hi,	NA	68	0	0	68	7	45	They	You	l see a	1	1
	97.00	2022-04-28	text;	Acknowled	photo; Threats	57	0	0	57	5	12	Thos	Last I	Its all	#Ca	Tha
	98.00	2022-04-26	text	I think	NA	135	1	0	136	11	11	Well	Very	Well	Wel	You
	99.00	2022-04-30	text;	I am so	photos;	1347	2	0	1349	106	197	My	preci	The	Tha	PIL
	100.00	2022-04-29	text;	Last night, I	artwork	280	6	2	288	45	52	#Can	good	What	1	L

Figure 3.2. A snapshot of the data collection framework and raw data management. The columns correspond with the type of information being gathered and the rows correspond with the date and each unique post "bundle".

representative snapshot of comments, as opposed to comments that were most popular with the users (though the latter would be a fascinating endeavor for future research and expansions of this dataset). As well, Facebook's structure for filtering comments under each post changed multiple times over the time period in which data was collected, including a series of months when it was not possible to filter through comments based on the most "liked". Given the high volume of comments, manually finding the most liked comments would have significantly extended the data collection timeframe for this thesis. All data was anonymous, with no personal information collected on posters and any potential identifiers, such as names and select locations, redacted from the

dataset. To clean the data in preparation for analysis, all bundles matching more than one keyword within a group were removed to ensure there were no duplicate posts within the dataset. Table 3.1 shows an overview of the number of unique posts gathered per keyword and group. See Appendix A for supplementary social media analysis information, including a weekly breakdown of post counts.

To analyze the data, a social media analysis was conducted employing methods similar to ones used by Xu et al. (2021) to analyze Facebook comments. Qualitative elements of the dataset (post content, type of post, type of embedded content, and comment text) were analyzed using NVivo and visualized using Excel. The remainder of the dataset was analyzed quantitatively via Excel for each group (both by keyword and by date) through the use of summary tables, including both sums and averages of each metric. This approach allows for the coding of key themes qualitatively while also making it simple to cross reference and support findings with visualizations of quantitative elements.

3.3.2 Supplementary Interviews

The latter component of this project utilized supplementary interviews with individuals who have contributed social media content on Grassy Mountain to the Facebook groups. As such, the aim of the interviews is to have a more nuanced conversation and clarify any points that may be unclear in the social media analysis. Ethics approval (GC-REB; Reference no. 20230638) was obtained prior to conducting interviews- the letter of approval can be found in Appendix B.

The interviews were conducted online (via a Memorial University WebEx account or phone call) and followed a semi-structured style, to ensure main points were covered but interviewees still had the space to elaborate on the points important to them and the nuances of a politicized topic. An interview guide based on recommendations put forth by DiCicco-Bloom and Crabtree (2006) was followed. The decision to conduct interviews online was made for multiple reasons. First, since the nature of this project examines behavior in virtual space, the online sphere is a fitting place to have a conversation and being in a similar environment may make it easier for interviewees to think back on specific instances of their online participation. Second, resource extraction is a highly politicized topic, especially in areas such as Crowsnest Pass where the local economy is fueled by the industry. The ideal interview space is a private one where the interviewee feels comfortable expressing genuine feelings towards the topic. Online interviews minimized both health risks and the personal risk associated with an interviewer discussing a highly politicized topic in a public space. Finally, online interviews are generally easier to plan for both partners and are easier to record without background noise. Interviews were voice recorded with the permission of participants and handwritten notes were taken throughout the interviews by the researcher. All interviews were conducted remotely- three using WebEx and one via phone call.

Participants were recruited through a public post made on each of the public Facebook groups of interest, using a Facebook profile created by the researcher for the purposes of this thesis. Though all the groups are public, the admin(s) of each group were privately messaged via Facebook prior to publicly posting to show that post is legitimate, nothing posted goes against the group guidelines set by the admin(s), and as an invitation to share the opportunity with their network. The public recruitment post was not made unless approved by an admin. To qualify as an interview candidate, the participants must have:

(1) Contributed content (in either the form of a post or a comment) regarding either the Grassy Mountain Mining project or the energy sector to the group(s) in question; and

(2) Made these contributions between December 2020 and August 2021.

These criteria were outlined in the post and group members interested in participating were asked to private message the researcher over Facebook if they were interested in being interviewed (see Outreach Materials document). Individuals that expressed their interest via private message were contacted within 24 hours on a first-come-first-serve basis. To ensure potential participants were serious about being interviewed, only text-based responses that address the purpose of recruitment (i.e., being interviewed) in some manner were considered. Messages including attachments such as links or images were excluded. When confirmation that the criteria have been met was received and the individual was still interested in participating, the researcher did all further planning via email, including sending the consent form.

Though the researcher aimed for six interviews distributed in proportion to each group's following (three participants from "Protect Alberta's Rockies and Headwaters" and one participant from each of the remaining groups), only four interviews total were obtained due to lack of admin response to recruitment posts. During the interviews, the consent form was reviewed with the participants and verbal consent to both the interview and to recording the interview was obtained. Verbal consent was recorded and stored separate from the interview recording to ensure anonymity, with all interview data stored under password protected files. The Facebook group that each participant is affiliated with was verbally confirmed in the first interview question to guide the interview, but due to anonymity concerns within a small community, group affiliation was not used within analysis. The interviews took between 45-90 minutes with each participant. Once interviews were completed, recordings were transcribed using Descript, and any identifying information (including group affiliation) was redacted from the document. Interview content was analyzed through qualitative methods using NVivo, with data coded and sorted following the process described by Hsieh and Shannon (2005). The themes emerging from each interview were examined individually, then by the themes and processes which emerged within the social media analysis.

3.4 Case Description

3.4.1 The Politics of Alberta's Energy Sector

Alberta holds a rich political history around hydrocarbon extraction¹¹ (referred to as the 'energy sector'). Being a foundational economic venture for the province since the first major oil boom in 1947, the industry is inseparable from both the politics and the culture of the region (Takach, 2017; Adkin, 2016). As a result, this also makes the energy sector a highly politicized topic-critiques of the industry are often interpreted as critiques of identity, livelihoods, and lifestyles. The full story of non-renewable resource extraction and the energy sector in Alberta is a complex one featuring intergovernmental relationships, political conflict, collective identity, and highly contested vision of the future (Adkin, 2016; Adkin et al., 2017; Blue et al., 2018). In this section I offer a snapshot of the province as a global player in the industry, including a general overview of the energy sector, the politics of the industry, and the associated culture. I then briefly discuss the energy sector from an environmental perspective, including existing environmental policy and the murky future of the sector.

Though it is the subject of debate, Alberta is often understood through the label of a "petrostate", which refers to "a nation whose economy is highly dependent on the extraction and export of oil or natural gas" (Adkin, 2016). Prior to this discussion, it is important to note that this in an incredibly complex topic and petrostate theory is not the sole predictor of provincial trends and outcomes, nor does it perfectly describe the situation- there are a variety of other economic, political, and cultural factors at play. Regardless, the concept of a petrostate offers an effective framework for an examination of Alberta's political economy. Canada itself is not generally

¹¹ I refer to this as the 'energy sector' or 'resource extraction industry'. However, coal mining is often categorized as the "oil & gas industry" in less technical discussions.

considered a petrostate- though it is a major oil producer, the economy is sufficiently diverse for the state to not be solely dependent on oil and gas sector (Adkin, 2016). However, both social and economic indicators demonstrate that the province of Alberta as an oil and natural gas exporter is comparable to other so-called first world petrostates, such as Russia or Venezuela (Adkin, 2016). A large portion of the provincial GDP comes from the oil and gas industry, meaning that the Government of Alberta's revenue is deeply intertwined with the energy sector. The "mining, quarrying, and oil and gas extraction" industry accounted for 26.6% and 25.3% of the provincial GDP in 2019 and 2020, respectively, making it the highest earning industry (followed by real estate, retail, and leasing at ~10%) (Government of Alberta, 2021). The economic effects of the industry include a significant provincial profit from crown land leases for the extraction of hydrocarbon resources, provincial tax policy aimed at keeping oil affordable, and much of the province's other industries and employment reliant both directly and indirectly on investment into the sector (Government of Alberta, 2020b, 2022). When profits from the energy sector were at their highest in the mid-to-late-1900's, Alberta's population increased by two thirds as individuals pursuing work (primarily young men) moved to the province (Takach, 2017).

With the large impact of the oil and gas industry on Alberta's political economy in the last half-century, it is not surprising that the energy sector is deeply political and intertwined with governance structures. For four decades prior to 2015, the Progressive Conservative (PC) party remained in power and relied heavily on the boom-and-bust nature of the energy sector, with mass oil and gas infrastructure expansion during times of prosperity (Adkin, 2016). An interesting change occurred in May 2015 in the aftermath of the 2008 recession, when the New Democratic Party (NDP) gained majority government and implemented Alberta's Climate Leadership plan, which included a carbon tax. In the background of the NDP's term, the two primary conservative opposition parties merged to form the United Conservative Party (UCP). Running with the slogan

"Alberta Strong and Free: Getting Alberta Back to Work", the UCP's platform revolved largely around a repeal of the NDP's "job-killing carbon tax" (pp. 17) and proceeded to secure the majority vote in the 2019 provincial elections (United Conservative Party, 2020). Shortly after, the Climate Leadership Plan was abolished under Bill 1 (2019) and various other changes were made with regards to environmental policy. Faced with the unique governance challenges of the pandemic, the UCP's platform remains focused on various facets of freedom, including- freedom of speech, economic freedom, and free individuals (United Conservative Party, 2020).

Interwoven with the province's political economy, a strong collective identity around the oil and gas industry exists in both rural and urban Alberta. On the surface, this is readily observed in abundance of oil and gas themed truck decals and lawn signs. However, the significant political and economic role of the energy sector also shapes workforce composition, race and gender dynamics, and overarching ideologies. Specific to workforce composition, employment in the non-renewable resource extraction industry and related labor work is largely targeted at young men and has historically reinforced gender roles within the workforce. This imbalance goes hand-in-hand with the social dynamics of the industry that present a specific picture of masculinity, coined as 'frontier masculinity', which is:

"[P]remised on a vision of the masculine as strong, rugged, self-sufficient – conquering the dangerous wilderness in the hope of striking it rich....In the context of Canadian settler society, frontier masculinity is a working-class notion of masculinity that defines itself against all that is considered feminine, nonwhite, and urban." (O'Shaughnessy & Doğu, 2016, p.268)

Though opportunities for working traditionally male jobs have become much more accessible to non-male individuals in the past decade, reports of harassment, a fear of personal safety, and lesser wages are far from uncommon (Adkin, 2016; O'Shaughnessy & Doğu, 2016). The idea of a 'frontier masculinity' also ties into the racial dynamics within Alberta, presenting significant barriers to nonwhite individuals within the area including Temporary Foreign Workers (disproportionately women employed in childcare or housekeeping), recent immigrants, and Indigenous people¹² (O'Shaughnessy & Doğu, 2016; Parlee, 2016).

In addition to collective identities shaped by the dynamics of Alberta's political economy, framing on behalf of well-funded organizations representing industry interests harnesses the power of an already existing collective experience to further strengthen ties to the oil and gas sector and cultivate a very specific culture (Davidsen, 2016). A variety of both corporate and non-governmental organizations exist specifically to cultivate strategic messaging around the oil and gas sector- the Canadian Association of Petroleum Producers (CAPP), the Canadian Energy Center, and Canada Action have been at the forefront of rebranding the resource extraction sector and mobilizing citizens as industry advocates (Davidsen, 2016; Wood, 2019). This is exemplified in a variety of efforts over the last decade. Canada's Energy Citizen's campaign in 2014, a subset of CAPP that harnessed the use of social media for news coverage in support of the industry (without explicitly stating its affiliation with CAPP), created shareable content around the energy sector and directed followers to CAPP's website which provided an advocacy toolkit with pre-made media content (Wood, 2019; Canadian Association of Petroleum Producers, 2022). This specific campaign strategy was intended to amplify the voices of industry supporters to "make previously invisible citizen support more legible in public debates" (Wood, 2019, p.83). The Canadian Energy Center is funded directly by the Government of Alberta and works to "defend natural resources owned by Albertans, and of benefit to the entire country" (Canadian Energy Center, 2022, pg.3), which includes the highlighting the positives of the oil and gas sector for the economy, environment, and community,

¹² For further reading on the subject of Indigenous communities' experiences and challenges within the Alberta context, Parlee's 2016 work offers a thoughtful and in-depth account of Indigenous rights and governance with regard to various facets of the energy sector.

as well as conducting research¹³ on the sector in the form of articles distributed through websites and social media. Canada Action, the distributor of I ♥ Canada Oil & Gas content, is a non-profit organization marketed as a grassroots movement of Canadian citizens in support of the oil and gas industry, often working in partnership with other organizations advocating for the sector through shared messaging on social media (Davidsen, 2016). Often collaborating on campaigns, these organizations form a backbone for strategically framing the resource extraction industry in both Alberta and Canada.

3.4.2 Environmental Regulation of the Energy Sector

Specific to environmental policy and regulations surrounding the energy sector, there have been various responses to concerns of the environmental impacts of the industry and approaches to minimize the harm caused by hydrocarbon extraction and consumption. Hydrocarbons refer to highly combustible organic compounds made of hydrogen and carbon which are found in petroleum (oil), natural gas, and coal. Coal can be classified into two categories- thermal coal, which is burned to produce heat; and metallurgical coal, which is used in industry. This case study pertains specifically to a proposed metallurgical coal mining project. Though it did not impact the project directly, an important aspect of understanding the case of Crowsnest Pass is the turbulent journey of Alberta's coal policy. Currently, both the Government of Canada and the Government of Alberta are in the process of phasing out thermal coal, which is burned to produce electricity. Metallurgical coal, such as that proposed for mining at Grassy Mountain, is not in the process of being phased out, but policies which further regulate its environmental impacts (such as emissions regulations and

¹³ This research is not peer-reviewed and generally does not contain references.

EIA's) still apply to these endeavors. This section outlines broad environmental regulations regarding the energy industry on both a federal and provincial level, then specifically discusses Alberta's coal policy.

In accordance with international commitments, the Government of Canada aims to reduce GHG emissions by 2030, with net-zero emissions by 2050 (Government of Canada, 2022a). Multiple emission reduction projects are currently underway to meet these goals and are enforced by various policies and legislation including the Pan-Canadian Framework on Clean Growth and Climate Change (2016). The resource extraction industry, which is the largest emitter and accounts for roughly 27% of Canada's overall GHG emissions, has been specifically targeted through initiatives such as new cap-and-trade systems, modified carbon pollution pricing, cut subsidies for both domestic and international projects, a coal fired electricity phase-out by 2030, and a coal transition fund (Government of Canada, 2020a, 2021). Additionally, the mining of all coal will be further regulated by the proposed Coal Mining Effluent Regulations which limits the levels of various toxic substances in coal mining effluent and will apply to all existing and future coal mines in Canada (Government of Canada, 2022b). Beyond policies targeted specifically at GHG emissions, other pieces of federal legislation help shape the energy sector and its environmental impacts, notably the Canada Water Act (1985), the Species at Risk Act (2002), and the Impact Assessment Act (2019).

Within the federal context, the province of Alberta also has a variety of laws and policies that govern the specifics of the energy industry within the province. Due to the high provincial revenues brought in by the oil and gas sector, industry players also held a high degree of influence over policy decisions regarding the management of energy resources (Goodday, 2021). Provincially, a variety of regulation exist under the Emissions Management and Climate Resilience Act including an outputbased emission benchmark for large emitters (Technology Innovation and Emissions Reduction Regulation), a cap on oil sands emissions (Oil Sands Emissions Limit Act), the 2030 coal phase out, a 45% methane emission reduction (Methane Emission Reduction Regulation), renewables in fuels (Renewable Fuels Standard Regulation), and reporting regulations (Specified Gas Reporting Regulation) (Government of Alberta, 2022b). Indirectly, the Environmental Protection and Enhancement Act (EPEA) (2000), the Mines and Minerals Act (2000), the Public Lands Act (2000), and the Water Act (2000) govern oil and gas related activities.

In 1976, A Coal Development Policy for Alberta was introduced to as the primary policy regulation coal development within the province, with the purpose of encouraging coal development in Alberta to maximize benefits from surface-level mining and to formally designate areas where coal exploration is viable and areas where development should be limited (Government of Alberta, 2022a). There was a spectrum of four categories indicating various levels of restriction on mining, with Category 1 allowing no coal development (in areas such as Parks) and Category 4 allowing any coal development (Government of Alberta, 1976). Controversially, the 1976 Coal Development Policy was rescinded on June 1, 2020 without any public consultation and amidst a major Covid-19 lockdown (Northern Saskatchewan Watershed Alliance, 2019). In the press release announcing this change on May 15, 2020, the rescindment was justified as "[t]he coal categories are no longer required for Alberta to effectively manage Crown coal leases, or the location of exploration and development activities, because of decades of improved policy, planning, and regulatory processes" (Chamberlain, 2020, pp.1). Even with other existing regulatory frameworks, this change meant that all but Category 1 restrictions that were previously designated in the Coal Development Policy would be removed, leaving various aspects of mining unregulated. This included the parts of the Eastern Slopes (designated a mix of Category 2 and 4 under the original Coal Development Policy), the mountain region around which the activism examined in this case study revolves (Northern

Saskatchewan Watershed Alliance, 2019). See Appendix C for a full map of areas impacted by the 1976 Coal Development Policy.

This change resulted in strong opposition from the public, due to both the environmental impacts of these changes and the lack of public consultation. In response, the Government of Alberta reinstated the *1976 Coal Development Policy* on February 8, 2021, and announced they would reconsider the coal categories, specifically through public engagement around coal exploration and approvals on Category 2 land, including the pause of all new exploration approvals (Government of Alberta, 2020b). This resulted in the continuation of the 1976 Coal Development Policy as well as a Ministerial Order restricting all coal exploration and development on the Eastern Slopes, regardless of their category (Lammie, 2022). It was in the greater context of these fluctuating policy decisions that the proposed development and public outcry on the future of Grassy Mountain and the surrounding area, referred to as the Eastern Slopes, took place.

3.4.3 Crowsnest Pass and the Grassy Mountain Coal Project

Crowsnest Pass is a specialized municipality located in southwest Alberta, sitting on the traditional territory of the Ktunaxa, Tsuut'ina, and Niitsítapi nations. The municipality is 80km north of the US border and the west side of the region sits adjacent to the British Columbia border (Municipality of Crowsnest Pass, 2019). Altogether, it incorporates the area of Improvement District no. 5 and five distinct communities: Bellevue, Blairmore, Coleman, Frank, and Hillcrest (Municipality of Crowsnest Pass, 2019). As of 2021, 5,663 individuals resided in the municipality (Statistics Canada, 2021). Both historically and today, the residents of the municipality are largely employed in the resource extraction sector, including both minerals, coal, and oil and gas (Statistics Canada, 2016). This section familiarizes the reader with the Crowsnest Pass region and provides background on the proposed Grassy Mountain Coal project.



Crowsnest Pass was shaped into the area it is today largely due to its advantageous geography in an industrialized world- a corridor of gentle slopes at a low elevation surrounded by the high, mineral rich Rocky Mountains. These physical attributes made it both the ideal location for a railway and for mining endeavors which, in combination with federal economic development initiatives in the mid-late 1800's, resulted in the Canadian Pacific Railway's (CPR) expansion into the region in 1898 and the first coal mine of Crowsnest Pass opened in the town of Frank in 1901 (Alberta Culture & Tourism, 2022). The coal industry thrived under these conditions, but after Alberta's first major oil boom in 1947 and the subsequent transition of the CPR locomotives from coal to dieselelectric, much of the demand that fueled the coal industry of Crowsnest pass tapered (Alberta Culture & Tourism, 2022; Wetherell, 2005). More recently, much of the Crowsnest Pass population still works in the resource extraction industry. As of 2016, 19.2% of residents worked within mining, quarrying, or oil and gas (in comparison to 6.3% of Albertans and 1.5% of Canadians) and 24.3% of residents worked in the trades, transport, or equipment operation sector (Statistics Canada, 2016).
Unofficial reports place around 400 individuals making the daily commute over the border to work in B.C. coal mines. Apart from the resource extraction industry, a 2020 Crowsnest Pass community survey reported that 39% of residents choose to live in the area because of the recreation opportunities, with an overwhelming majority of residents stating that the elements of the region which they value most are access to nature, outdoor recreation opportunities, and the small-town atmosphere (Municipality of Crowsnest Pass, 2020).

The geographical region described above is referred to broadly as the Eastern Slopes and in addition to the local economic benefits it has provided throughout history, this area is also home to the South Saskatchewan Basin headwaters (Marshall et al., 2011). This basin is associated with the South Saskatchewan River, which begins flowing at the headwaters in the Eastern Slopes, continues across Alberta, Saskatchewan, and Manitoba, and ultimately drains into Hudson's Bay. For over 10,000 years, the basin has been a primary water source for the communities surrounding it, with an estimated 3 million people using this water system today for both irrigation, consumption, and recreation (Eamen et al., 2020). One key segment of Alberta's economy that places high stakes on the well-being of the basin is the farming and ranching community, who require healthy water and pastures to sustain their industry and livelihoods. In addition, numerous animal species are dependent on the river, including a variety of fish, birds, amphibians, and mammals protected under the Species at Risk Act (SARA) (Government of Alberta, 2007). There are also four internationally recognized Important Bird Areas established along the South Saskatchewan River (Marshall et al., 2011). Both the quality and quantity of this river are negatively impacted by rural and urban water use, pollutants from commercial activity, damming, and climate change (Eamen et al., 2020). This raises concerns regarding the future of the river and living beings that depend on it, especially when damage occurring at the most upstream point of the water system has the potential to impact millions of individuals. Grassy Mountain falls into Category 4 of Alberta's 1976 Coal Development

Policy, meaning that though the area was monitored, unrestricted metallurgical coal mining could still occur until February 8, 2021 (Government of Alberta, 2022a) (See Appendix C for a full map of areas).

Benga Ltd, the company leading the Grassy Mountain Mine project, submit their initial application to the Government of Canada for approval in 2015. In 2018, a Joint Review Panel consisting of the Alberta Energy Regulator and the Minister of Environment was established to further investigate the impacts of this project as per the Responsible Energy Development Act (REDA) and the Canadian Environmental Assessment Act (CEAA) (Government of Canada, 2020b). At this point, the project was actively opposed by CPAWS and had some coverage by media, but opposition had little mainstream traction until an online public hearing was launched by the Joint Review Panel in 2020 (Government of Canada, 2020c). On October 27, 2020, a public hearing for the proposed open-pit, metallurgical coal mining project on Grassy Mountain began, using primarily electronic means due to the pandemic. Throughout the hearing, the project gained increased media attention and opposition. In November 2020, the Niitsitapi Water Protectors were formed to protest the lack of consultation with Indigenous communities regarding the project and partnered with CPAWS in hosting information sessions regarding the project. By the time the hearing was concluded on December 2, 2020, the proposed project was met with great pushback from the public. At this time, the Joint Review Panel asked for an extension on a final decision regarding the project, due to both Covid-19 and the need to further investigate the newfound public concern (Government of Canada, 2021c).

An online grassroots movement of citizens advocating for the denial of the project began on December 21, 2020 through a public Facebook page called "Protect Alberta's Rockies and Headwaters". This group was established as the primary channel voice to concerns and rally support



against the mining project, primarily due to concerns of widespread environmental degradation due to the coal mine contaminating the headwaters and consequently, the basin and river. Over the course of a few months, the group gained a 37,000-person following (for context, this is over six times the population of the Crowsnest Pass region). However, pushback to this movement also occurred, with a Facebook group titled "Crowsnest pass. Ghost town" created specifically as a promine response to the large anti-mine movement, and several other public pages ("Around and About the Crowsnest Pass" and "Crowsnest pass. Ghost town") expressing views on the issue. As of June 17th, the Joint Review Panel released their findings, and the Alberta Energy Regulator denied the project. By August 6th, 2021, the proposed development on Grassy Mountain was officially shut down by the Government of Canada, under the Canadian Impact Assessment Act (IAA) (2012) (Government of Canada, 2021c).

3.5 Conclusion

Once a fuller story of Grassy Mountain and the surrounding regions is understood, it is not surprising that tensions surrounding the Eastern Slopes and Grassy Mountain exist. Given the economic, environmental, and cultural value the region holds, this case study is situated at the collision of two opposing interests regarding physical space, which seem to be mutually exclusive to some degree. The peak of this conflict occurred within the backdrop of fluctuating restrictions on movement due to Covid-19 and both formal and informal communications regarding this issue were conducted primarily online (Assaly, 2021). These qualities, along with the researcher's lived experience and understanding of the political nuances of this area, make the proposed Grassy Mountain Mine an excellent case study for examining political advocacy within the online and offline worlds.

The impacts of the many individual voices that participated in these discussions and the collective they created can be observed through the various Facebook pages advocating different political outcomes for the Grassy Mountain coal mine. Using public Facebook posts and conversations surrounding the proposed coal mine as a case study, a qualitative social media analysis and interviews with online participants are used to better understand the various of aspects of political participation in both physical and online spaces during Covid-19.

Chapter 4: Social Media Analysis

As was outlined in preceding chapters, a social media analysis was conducted to further explore the use of Facebook as a communication tool used by both individuals and collectives to advocate for their political interests. Chapter 4 begins by setting the scene with a primarily quantitative analysis of basic group make-up and purpose, then continues on to present findings on social movement processes, networks, and approaches. The chapter concludes with a brief discussion of the social media analysis findings in their entirety, drawing connections to literature and their relevance in today's world. Diverse perspectives on the issue of coal mining between the selected groups and varying levels of participation, among other factors, tell an elaborate story of the nature of environmental movements in the 21st century.

4.1 Basic Group Information

Working off Milan's (2015) concept of cloud networking and other literature on both individual and collective political action, a mixed methods approach was taken to examine the discourse between individuals over Facebook and the complex processes of digitally mediated social movements. Activity surrounding the proposed Grassy Coal mine was examined over the span of 35 weeks on four public Facebook pages - "Crowsnest Pass- Uncensored" (CPU), "Around and About Crowsnest Pass" (AACP), "Crowsnest Pass. Ghost town" (CPGT), and "Protect Alberta's Rockies and Headwaters" (PARH). This section provides further information about these groups and trends in both levels of activity and types of activity between December 2020 and August 2021.

Across all four groups examined in this social media analysis, 484 posts with 2,420 corresponding comments were analyzed qualitatively, amounting to a total of 147,525 words of text-based data. In addition, the engagement metrics for the comments and posts were examined through

Table 4.1. A summary of the groups examined, their purpose, and engagement key metrics (number of members, number of posts analyzed, number of comments analyzed). Only original, non-duplicate posts and comments were counted.

Group name	Date created	Purpose Statement	Number of members	Number of posts analyzed	Number of comments analyzed
Around and About the Crowsnest Pass (AACP)	May 22, 2013	Welcome! This group is about stories and the happenings around the Crowsnest Pass. Please feel free to post missing people, pets, weather road conditions, whatever can help someone out, employment opportunities, yard sales, music and any other events that people maybe interested in.	5,779	1	5
Protect Alberta's Rockies and Headwaters (PARH)	December 18, 2020	We are a "loose coalition of concerned citizens." (The Tyee) This group is dedicated to raising awareness and information related to Alberta's proposal to restart open pit coal mining in environmentally sensitive areas like Crowsnest Pass and Kananaskis and Clearwater County (It's called that for a reason.) While we understand Albertans are hurting economically, starting open pit coal mining in sensitive areas is NOT the solution.	36,236	455	2275
Crowsnest Pass. Ghost Town (CPGT)	January 19, 2021	"I started a public group after being censored and banned from every anti coal group that are lobbying to shut it all down. They cost me and 140 other people to now be out of a job so feel free to join and I will try expose some of the people behind this lobbying like the md of ranchlands and Corb lund." -Page Admin	141	24	120
Crowsnest Pass – Uncensored (CPU)	June 1, 2021	Anything Crowsnest Pass related without the censoring of some of the other groups. Be kind and polite and anything goes.	1,421	4	20

a quantitative analysis of 2,904 unique data points (see Appendix D for term definitions). Table 4.1 provides a summary of the four groups including their purpose, the number of followers the group had at the time of data collection (follower counts are not available retroactively), and a count of the data analyzed for each group. Two of the groups were created specifically for the discussion of coal mining in Alberta, with PARH taking a strong anti-mining stance and CPGT taking a strong pro-



Figure 4.1. A timeline of posting activity from December 16, 2020 to August 14th, 2021. Colored lines show the number of posts collected each week for Protect Alberta's Rockies & Headwaters (PARH), Crowsnest Pass Uncensored (CPU), Crowsnest Pass Ghost Town. (CPGT), and Around and About Crowsnest Pass (AACP). Three major regulatory events are shown (red, dashed lines).

mine stance. The remaining two groups are specific to the Crowsnest Pass region but were not created as a space for the discussion of coal mining, which is partially reflected in the low number of posts meeting the researcher's search criteria (Table 4.1). This section covers general information on activity within these groups including posting activity time over the timeframe studied, the types of posts within the group, and post engagement.

An overview of activity levels in each group over time and in relation to key regulatory events that marked milestones in the project's downfall is shown in Figure 4.1 (the links between these governance actions and posting activity are further explored below). The AACP group only had one post, during the week of December 20th. Overall trends in the PARH group show fairly consistent posting patterns throughout the timeframe examined, with an average of 16 posts per week and a slight decrease in posting frequency over time (overall, ~5 posts per week). The posting activity of the CPGT and CPU groups begins on the week of June 6th and remains active until the end of the time frame in question, with an average of 5 and 2.5 posts per week, respectively. When CPU, CPGT, and PARH are all active, the same trends within posting patterns can be observed across the three groups (Figure 4.1).

Generally, a spike in posting activity can be observed during three major policy decisions- the Government of Alberta reinstating the 1976 Coal Policy after strong pushback against its rescindment, the Alberta Energy Regulator (AER) recommending against the Grassy Mountain Project, and the Impact Assessment Agency officially denying the project on the federal level (Figure 4.1). During and around these events, the number of post spikes increased only slightly in comparison to overall posting activity in the PARH group. However, posting activity increased significantly in relation to overall activity in the CPU and CPGT groups from June 13th to June 26th. An increase can also be seen in the CPGT group the week of August 6th. Interestingly, a major spike in posting activity can also be seen on the week of July 18-24 in CPGT, CPU, and PARH but possible reasons behind this trend are not yet known. This increase is also slight in relation to other weeks in the PARH group, but constitutes one of the two major activity spikes in the CPGT and CPU groups. Finally, a drop in posting activity in the first two weeks of July is observed- possibly due to a combination of statutory holidays, the Calgary Stampede, and a major easing of Alberta's Covid restrictions on July 1st (Assaly, 2021).

Figure 4.2 shows a breakdown of the types of posts that were made across all four groups (see Appendix A for post types in specific groups). Text was the most popular choice for communicating, with 94% of the posts examined containing some textual content added by the user



Figure 4.2. A count of post types across all groups (total 483). "Post type" refers to the general structure of a post and are the combination of six possible formats- text, link, image, attachment, video, and reposts. See Appendix C for term details.

and 50% of the posts containing only text. Text-based posts ranged from 3 to 1294 words and primarily used the English alphabet with occasional emojis. Most of the text seemed to be written directly by the user, with the remainder either pasted in from another source or presented as a quote from an individual who is not in the Facebook group. Without including the users that posted personalized versions of text-only content that was directly copy and pasted from other sources, 44% of content included some combination of text and another type of media (such as a photo+text or link+text combination). This is an excellent indicator of individuals customizing the content they post with messages that resonate with them, without fully creating their own content for posts. The ease of personalizing already existing content on social media is one of the characteristics Bennett and Segerberg highlighted in their theory of connective action (2012), as it allows individuals to share their personal connection or views on a topic without needing to invest significant time or efforts into creating the primary content.

Images and links were the second most popular post types, making up 25% and 20% of all posts, respectively. Apart from text, links were the most frequently used standalone post type by a significant margin. This observation is most likely due to the structure of Facebook, as hyperlinks pasted into a post automatically get expanded into a cover image, the link title in large letters, and a preview of the content. This results in a complex presentation equivalent to posting an image and text, making it a lower-effort way for individuals to communicate a variety of information through one click.

Participation and interaction with posts (referred to as "post engagement") throughout the groups was defined as the total number of times the post was interacted with, i.e. the sum of likes, shares, and comments (Appendix A). As was discussed in Chapters 2 and 3, current definitions of online political participation have their limitations- the framework for identifying political participation used in this thesis is not perfect, but it allows for a relatively clear quantification of group member activity and a comparison across groups (van Deth, 2021, Ruess et al., 2021). Average engagement with various post types can be found in Figure 4.3 and a breakdown of average engagement across all four groups can be found in Figure 4.3. To ensure accuracy when comparing engagement across groups of different sizes, an average of the likes/comments/shares per post in each group is displayed.

Building upon the data presented in Figure 4.2, Figure 4.3 shows the relationship between post type and engagement across all groups. Page followers seem to interact the most with reposts (either via Facebook or other social media platforms) and posts that contain a combination of text and either video or images. However, text-based posts with a third element (such as text; link; image)



Figure 4.3. Average number of interactions for each post type throughout all groups. Lighter bars signify a sample size of <10. These values account for variation in the number of post types analyzed (shown in Figure 4.3).

or posts containing only text were not interacted with as intensely as those containing only two elements (Figure 4.3). Reposts were the exception to this observation. Since the 'repost' category refers solely to another social media user's content without specifying the type of post the original content is, it is difficult to determine the type of media a 'repost' constitutes and how this relates to interaction levels. Engagement with reposts is interesting to note but it is also difficult to draw conclusions based solely off of this dataset, as it only looks at engagement metrics from 8 reposts and further research would be needed to clarify trends in repost engagement. This also applies to a variety of other post categories which had a small sample size (denoted in Figure 4.3 as light blue bars). As well, it is important to note that the structure of a post is not the only determinant of its popularity. Though structural components make a post more noticeable or convenient to engage with, the content of the post cannot be dismissed. Exciting, catchy, interesting, or otherwise engaging post content has been shown to play a role in user's decisions to engage with a post or user (Ackland & O'Neil, 2020). Section 4.4 takes a more qualitative approach to analysis through an exploration post content, including keywords and the themes individuals throughout groups posted about.

Another observation from Figure 4.3 is that any combination of post types containing links saw a low level of engagement relative to their non-link counterparts. For example, 'text; link; image' had an average of 120 interactions, whereas 'text; image' had an average of 289 interactions. Further, the 'text; link' post type had 107 interactions, while the 'text' post type had 147 interactions and the 'link' post type had only 72 interactions. This observation may be due to the additional commitment that interacting with a link entails. While posts with text or images allow for the content to be consumed in that moment, links require the user to leave the social media platform (whose structure is often designed to prevent this (Tufekci, 2018; Milan, 2015), navigate a new site, consume the content on that site (generally longer or more robust than a social media post), and then navigate back to the social media platform to engage by liking, sharing, or commenting. This process is both time and labour consuming when compared to simply liking a photo on Facebook while scrolling. Barriers to connectivity, such as slow or unreliable internet access, may also be an influence on how likely individuals are to interact external content, such as links.

When comparing user engagement to the post types shown in Figure 4.2, it is interesting to observe the discrepancy between what users post versus. what users interact with. As we saw in Figure 4.2, individuals were most likely to post text-based content with some combination of text, link, and image posts accounting for the majority of content posted on the pages. However, an examination of engagement shows that users are most likely to interact with other posts if they are in the form of a repost or contain text and images/videos, with links showing the biggest discrepancies in popularity between posting and engagement. There are a variety of possibilities that



Figure 4.4. Engagement breakdown for posts analyzed within four Facebook groups- CPGT, CPU, AACP, and PARH. Engagement is calculated as a sum of the number of shares, comments, and reactions on the posts examined. For optimal comparison across groups of different sizes, the values shown are the average engagement for a post in each group.

may begin to explain these findings, such as the ease of posting a link (easy) versus the ease of accessing a link (hard, especially with poor connectivity). Reposts, videos, and images accompanied by text may also be more engaging to users they evoke an emotional response within the first few seconds of encountering them, as opposed to a block of text with requires more time and attention (Garcia et al., 2016). It is unsurprising that visual content may evoke more emotion and consequently increases users' desire to interact with the content, given that the combination of resource extraction in Alberta is highly linked to identity and that the subject of protest being a physical space that is often valued as beauty and status as 'pristine nature' (Castells, 2012; Adkin et al., 2017, Davidsen, 2016). This observation is discussed in greater depth in the chapters that follow.

A variety of post engagement patterns are observed across the groups examined (Figure 4.4). PARH, the group created specifically to advocate against the Grassy Mountain mining project due to environmental concerns, has the highest number of average interactions per post, a total of 260. About 70% of this total consists of reactions to posts, as opposed to comments or shares (16% and 14%, respectively). AACP has the second highest average interactions per post, a total of 177, with the majority of interactions coming from comments on the post (91%). A similar pattern can be observed in the CPU group which has a much lower number of average interactions per post, but is also dominated by comments as the primary method of engagement with posted content (62%). Finally, the CPGT group has the lowest post engagement of the four groups examined, with 21 average interactions per post of which 62% are reactions (Figure 4.4).

When post engagement is examined as an average per post as opposed to an absolute value, it is evident that the posting activity with regard to a specific topic is not proportional to the number of posts within a groups or posting frequency. The AACP group, for which only one post related to the Grassy Mountain Project was analyzed, had a much higher engagement rate than CPU or CPGT and was compared to the PARH group, for which 455 posts were examined. However, engagement with individual posts seems to, unsurprisingly, increase as the group following increases. It is interesting to note that post engagement in groups that are specifically dedicated to the topic of coal mining (CPGT and PARH) is primarily in the form of reactions to the post, as opposed to comments or shares (Figure 4.4). The groups that were not specifically established to discuss coal mining, with less reactions and shares on average (Figure 4.5). Under the assumption that comments signify discussion within the group, this observation ties into the idea that online spaces dedicated to a specific issue can often act as echo chambers where members will show high support for posts, but there is relatively low discussion on the matter. Though this is a trend observed within the data

collected and aligns with literature on the topic (c.f. Vitak et al., 2011; Greijdanus et al., 2020), it is worth noting that the sample size for both CPU and AACP were small in comparison to the other two groups and may not be the best representation of political discussion in non-political groups¹⁴. There is also a slight gap in the PARH comment data as there are no comments available from May 13-June 2, due to admin regulating comments perceived as going against group guidelines, meaning that in reality the ratio of comments to other engagement metrics may be slightly higher than the data shows. However, this accounts for less than 7% of all comment data from PARH and given that the average post engagement is presented, should not drastically affect the trends observed.

4.2 Structures and Processes Within a Social Movement

To learn about social movement processes in a digital space and what shapes them, Section 4.2.0 presents findings on structures, networks, and communication strategies. Across the four groups examined, PARH and CPGT were established with the intention of achieving political interests with regard to coal mining on the Eastern Slopes. PARH was established first to protest coal mining and less than a month later, CPGT was established to express support for the coal mines and voice discontent with anti-mining activism in the Crowsnest Pass. The type of content posted within the two groups are examined, including the type and structure of posts, as well as any external links. Then, the broader activism landscape regarding the Grassy Mountain Project is briefly explored.

A thematic analysis of the structure and intention of the posts on both PARH and CPGT were examined. The audience and intention(s) of posts in both groups were coded into seven main themes¹⁵- calls to action; informational; addressing the group; explicitly political content; letters and

¹⁴ Groups not dedicated specifically to political topics or conversations.

¹⁵ Note that these themes are *not* mutually exclusive. One post may fall into more than one category.



messages; personal experiences; and other. All posts within the two groups fit within these seven themes, with seven subthemes emerging in the PARH group and four subthemes emerging in the CPGT group (Figure 4.5a and 4.5b). Both groups share the subtheme of posts with questions and messages for the group. In the PARH group, the 'explicitly political content' theme is further broken down into posts about the United Conservative Party of Alberta (UCP), government in general, and the Alberta New Democratic Party (NDP). The "other" theme includes hashtags, 'NA' (where posts did not include any text), and any other posts not fitting any other themes (Figure 4.5a and 4.5b). The 'letters and messages' theme in the PARH group defaulted to 'letters and messages for officials', as there were no other types of letters sent to external bodies. In the CPGT group, the only unique subtheme emerged from 'letters and messages for...', which included letters and messages to both government officials and Corb Lund, a local country musician who was publicly outspoken against the Grassy Mountain Project. Though no formal analysis was done on the frequency that individuals were active in the groups, it is worthwhile to note that the CPGT group had more reoccurring users (multiple posts from the same individual or individuals) than the PARH group. Based on observations made during data collection, at least half of the posts made on CPGT were made from one Facebook user.

In the PARH group, 40% of mentions fall into either the 'informational' or 'calls to action' themes. 'Calls to action' refer to any posts that are intended to prompt page followers to act on the issue with either specific action items, such as requests to write letters to MLAs, or less tangible calls to action, for example "Stop all exploration activity immediately. Period." (Protect Alberta's Rockies and Headwaters, May 28, 2021). 21.4% of coded PARH posts fell into this category (Figure 4.5a). The 'informational content' theme included any posts that are mean to educate or inform page followers, such as symptoms of selenium toxicity or the overall environmental impacts of mining. Posts reflecting these themes were coded solely because of their intention to communicate

information to inform their audiences, not based on accuracy or neutrality of the information

(Figure 4.5a).

	Thomas	Number of	Description	Energy (and the	
	Theme	mentions	Description	Example (excerpt) Livingstone land owners did a study but as	
CPGT	Informationa I content	9	Any content designed to communicate informational material realted to the political aspects of activism (ex) coal mine), without addressing	soon as their "supposed" scientists called Selenium a toxic pollutant it lost any credibility as a professional study. Selenium	
	Personal experience	5	Any posts telling a first-hand story of an experience that occurred to the user	I was one of the 140 people that lost our jobs suddenly march 1st when I found out via a youtube video that they stopped all exploration and I had to call the 2 guys i had coming to work and say sorry. No more job.	
	Addressing the group: messages	roup: 7 "calls to action" or "informational content" category. This code		Thanks for stopping by 🙂	
	Addressing the group: questions	1	IANY dijestions posted to the group that reduire a response (not	Which businesses in town support the coal mines[?]	
	Calls to action	1 4	Messages explicitly asking for political action or show of support for the cause from group members.	We need more people to voice their personal impacts, you do not need to go in depth as I did but any words expressing the shock of rejection of a mine in a mining community needs to be heard!	
	Explicitly political content	1	Any content that is specifically making a political statement or expressing views related to specific figures or political entities. Letters & messages addressed to political figures do not fall into this category.	The current Liberal Government intends to "end all thermal coal mines" by 2035.	
	Other	1	All other posts not fitting a category	Corb Lunds cows.	
	Letters and messages for:	5	Any formal letters or social media messages mailed to or posted on the pages of various individuals. Sub-categories contain information on Corb Lund or general government	Dear Honourable Sonya Savage, My name is [name], I had moved to Alberta recently from Saskatchewan to pursue a career as a Geologist to work in the Metallurgical Coal Industry of Southern Alberta	

Table 4.2a. Codes, their counts, and an example for CPGT text analysis.

'Addressing the group', 'political content, and 'personal experiences' were the next most common themes, cumulatively accounting for 38% of posts coded. The 'addressing the group' code was split fairly evenly between questions for the group, such as "Are there any other hash tags you want?" (Protect Alberta's Rockies and Headwaters, January 13, 2021) and comments specifically addressing the group, for example "Thanks for listening. Thanks for standing up and using your voices!" (Protect Alberta's Rockies and Headwaters, July 16, 2021). The 'political content' theme

	Theme	Number of mentions	Description	Example (excerpt)	
PARH	Calls to action		Messages explicitly asking for political action or show of support for the	You are in the position to make a stand, make a difference, don't make the mistake of ignorance and allow these mines to happen.	
	Informational content	134	Any content designed to communicate informational material realted to the political aspects of activism (ex) coal mine), without addressing particular members or requiring an action. This information does not have to be correct to be assigned to this code, only intended to spread information	With the development of roadways, clear cutting trees, increased industrial traffic, loss of native grasses and then the air and water pollution that will follow with open pit coal mining we add the wild horses to the list of species that will be negatively effected.	
	Addressing the group: messages	52		I want to say I admire everyone in this group who is taking time out of their lives to oppose this. I wish we didn't have to!	
	Addressing the group: questions	40	Any questions posted to the group that require a response (not rhetorical)	Are there any avenues worth pursuing to inform Australian citizens about what is going on here in Southern Alberta?	
	Explicitly political content	89	addressed to political figures do not fall into this category. Sub-categories	There is a long fight ahead against a government that is unwilling or unable to see past their own ideologies.	
	Other- hashtags	49	All hashtags	#MountainsNotMines	
	Other- NA	24	Posts that had no text.	NA	
	Letters and messages for:	69	Any formal letters or social media messages mailed to or posted on the pages of various individuals. This was primarily letter writing templates or campaigns.	Below is an open letter I've drafted, but unfortunately I have no idea as to the best recipient, if you feel this is useful, and applies any level of credence to the situation, please forward to the relevant authorities.	
	Personal experience	84	Any posts telling a first-hand story of an experience that occurred to the	I am in my 60's now and can truly say I am not an "activist" but this is a topic that will make me take up my markers or paint, make a sign and join protesters that are trying to defend and protect our Eastern Slopes, our watershed, our environment and our health.	

Table 4.2b. Codes, their counts, and an example for PARH text analysis.

included posts regarding the government as a whole, the United Conservative Party (UCP) specifically (this includes any mentions of political figures within the party, such as Jason Kenney or Sonia Savage), and one mention of the opposing New Democratic Party (NDP). Over half of the posts coded to this category fell into the 'UCP' subcategory and had a strong negative sentiment. The 'personal experience' theme contained first-hand accounts or personal stories, such as a post on growing up in a coal-based economy (Figure 4.5a).

Finally, around 10% of posts contained letters or messages to officials (Figure 4.5a). This theme consisted of either letters mailed or emailed to government officials and their responses, or

posts that specifically addressed officials, often by using the tag function of the platform that gives the user or page tagged a notification and access to the post they were tagged in. The latter approach is especially interesting, as it relies solely on a social media platform to convey a very public message. This seems comparable to challenging the individual or actively starting a conversation in a public space, as opposed to the more passive approach of users posting letters and their responses (Edirisinghe, 2011; Elder, 2020). These are useful in giving PARH followers a recap of the conversation the user had, but do not allow for real-time deliberation with the person being questioned. To note, none of the political figures tagged or mentioned in the posts analyzed provided a response or otherwise engaged via Facebook.

In the CPGT group, 52% of mentions fall into either the 'informational content' or 'addressing the group' categories (Figure 4.5b). With 27% mentions, 'informational content' was the most prevalent theme in the CPGT group and focused primarily on providing information on the economical impacts of mining or providing information to counter claims made by a variety of promining groups. 'Addressing the group' was broken down into two sub-themes which were consistent with sub-themes in PARH, 'questions' and 'messages''. Seven of the nine mentions coded to this theme addressed the group, as opposed to asking the group questions, with admin often providing updates on activity regarding both pro- and anti- coal mining activism.

The next most frequent themes accounted for 42% of all mentions and included 'personal experiences' (5 mentions), 'letters and messages for...' (5 mentions), and 'calls to action' (4 mentions). 'Personal experiences' accounted for 15% of mentions and generally focused on stories of the importance of coal-mining industry for both individuals and historically in the Crowsnest Pass region or first-hand accounts of interactions with members of anti-mining groups. The latter is especially interesting, as posts falling into this theme often contained screenshots or copied text

#speciesatriskalberta #protectalbertasrockiesandheadwaters #livingstonerange #valartists #voicevourtruth #savethemountians #albertasocietyofartists #speciesatrisk #mauntainsnotmines #cancelcoal #nomontem #speakup #savethemountains #protectourwater #ableg #nixthesix #bevmazurickfineart #hope #studioartist #saveourmountains #mountainsnotmines #rejectgrassy #forourchildren #waternotcoal #callstoaction #protectourmountains #protectourheadwaters #albertabeyondcoal #professionalartist #wildhorses #protectionnotpolicy #wildstonestories #federationofcanadianartists #loveoureasternslopes #saveoureasternslopes

Figure 4.6. A word cloud visualizing the hashtags used in the PARH group. Size of the word is proportional to the number of times the word was used, with larger words signifying more uses.

from conversations which were deleted or banned from the groups they were posted on, most often the PARH group. This type of post was rarely, if ever, seen in the PARH group. 'Letters and messages for...' consisted of two subcategories in the CPGT group- messages for Corb Lund ¹⁶ (60%) and messages for government officials (40%). Messages for Corb Lund included only posts on both his personal social media page and CPGT, with a primarily negative sentiment reflected in the posts coded to this category, such as "...Corb lobbied to kill any prospect for jobs and prosperity for the Piikani [N]ation so his rich ranch friends can continue to graze on traditional [F]irst [N]ations land." (Crowsnest Pass. Ghost Town, 2021). Unlike the PARH group, "calls to action" was not a major theme within the CPGT group, with only 12% of mentions coded as such (Figure 4.5b). Finally, only two mentions (6%) corresponded with the 'political content' and 'other' themes. Unlike PARH, posts containing explicitly political content were not common in the CPGT group and the only post falling into the 'political content' category was specifically regarding the

¹⁶ Corb Lund, a local country musician who was publicly outspoken against the Grassy Mountain Project

Liberal government (Figure 4.5b). This differs from the PARH group, which primarily consisted of political posts regarding the UCP and to a lesser extent, the government as a whole.

Hashtags were used fairly sparingly on the PARH page (7% of the coded content containing a unique hashtag) and were not used at all on the CPGT page. Figure 4.6 shows a word cloud with the hashtags used in the PARH group. Hashtags are a common strategy in online social movements as they provide a structure that allows for increased visibility and access to a collection of posts on a specific topic (Halpern et al., 2017; Faure & Beltran, 2021). This is especially useful on social media platforms with a high content turnover that lack the built-in features that allow for formal organization, such as Twitter or Instagram. Hashtags are ultimately a grassroots approach to carving out space for a social movement in the online sphere and identify a social



Figure 4.7. Embedded content in the PARH and CPGT groups. The doughnut graphs show the types of embedded content on posts within each group, with the tables showing the sub-types of embedded content. Colours correspond to categories, with subtypes of a lighter shade denoting a lower frequency and darker shades denoting a higher frequency, distributed on a scale from 1 (min) to 62 (max).

movement across multiple social networks (Bennett & Segerberg, 2012; Milan, 2015; Tufekci, 2018). Facebook has a relatively organized structure, meaning that the platform contains a variety of distinct, built-in features (such as Groups or Events) that have benefits similar to hashtags. This aligns with the observation that less than 10% of the Facebook group content examined contains hashtags. However, it is also interesting to note that hashtags are not entirely obsolete in the PARH group and there is unity in which hashtags are used, including the consistent use of #mountainsnotmines and #waternotcoal (Figure 4.6). The use of hashtags in a space where there is little structural value for their use shows the level of integration between different platforms and offer a glimpse of the larger activism network. This observation also touches on the relationship between the user and the platform, specifically the extent to which the structure of a platform shapes user behaviour while accommodating overlaps between networks and allowing the user to make space within these pre-set structures (cf. Milan, 2015; Mosquera et al., 2020).

As was briefly discussed in Section 4.1.0, over half of the posts in PARH and CPGT posts contained some form of embedded content (Figure 4.2). Embedded content refers to any non-text additions to a post that either internal (within the group's page) or external (leads to an external source away from the group). In data extracted from PARH and CPGT, internal content was considered to be images, videos, and attachments uploaded through Facebook, and GIFs. External content was reposts and links to either websites, documents, or videos. To provide further insight into the types of content within posts, embedded content was sorted into subtypes. Figure 4.7 shows the types of embedded content present in each group along with their respective subtypes and the frequency at which they occur.

The PARH group had a total of 171 posts with embedded content, spanning five main categories and 20 subcategories. The two most common types of embedded content were images (50% of embedded content) and links (32% of embedded content). Within the image category, 56% of content was photos, 26% was screenshots, and the remaining posts included maps and artwork.



Figure 4.8. References to external organizations. 54 posts and comments were coded (out of a total of 2241). Search words shown in the boxes were used, with parentheses denoting more than one key phrase used in the search.

Links were also a popular type of embedded content and led to a variety of external sites including government websites, mining company sites, other anti-mining actors, and news outlets. From the links posted, 44% of them led to various forms of related news websites such as the Lethbridge Sun, Calgary Herald, or CBC, that offered updates on events surrounding the proposed Grassy Mountain mine. Nearly another quarter of links (23%) linked directly to one of two Government Canada sitesa petition against the Grassy Mountain mine or the Environmental Impact Agency's page containing information and documents on the proposed project. Within links, the 'other' subcategory contained an assortment of links including political cartoons and personal business websites. Other popular subcategories within the remaining categories included Facebook reposts, as well as informational videos and documents (Figure 4.7).

The CPGT group contained a total of 11 posts with embedded content, which spanned three main categories- images, videos, and links (Figure 4.7). Images were the most common form of embedded content within CPGT, with the 'photo' and 'screenshot' as subcategories. These were also the most popular image choices in PARH as well. A unique subcategory titled 'home video' emerged as a form of embedded content that consisted of videos often taken directly by the user to document some aspect of their life or experiences. This is interesting to note with regard to Figure 4.5b showing 'personal experience' as a theme in CPGT posts, as the presence of home videos is

consistent with the observation that more of the CPGT content is centered around personal experiences as opposed to large scale, societal issues.

Another finding that emerged from the social media data examined is the existence of a larger activism network related to the Grassy Mountain mining project beyond the PARH and CPGT groups. The PARH group members specifically mentioned multiple external actors, with a variety of posts made by organizations promoting action and events via the Facebook group (Figure 4.8). 2241 posts and their comments were examined in this analysis, with 54 of them containing content related to external actors. Organizations or collectives frequently mentioned are groups advocating against the mine, including the Niitsitapi Water Protectors, Livingstone Landowners Group (LLG), Alberta Wilderness Association, and CPAWS. References to these organizations were generally either posted by the actor to share their activity or thoughts on the topic or by group members discussing their activity. Various activity regarding the mine by the Kainai (Blood Tribe), Siksika, Stoney Nakoda, Tsuut'ina, and Piikani Nations was also referred to on the page, denoted in Figure 4.9 as blue boxes. These Nations were generally referred to across a wide range of contexts, however, the official stances of the Nations varied¹⁷. Mentions were also made to a Go Fund Me page which was established to raise funds for group activities such as sign-making. To note- a variety of embedded content also referred to external actors through the use of non-text-based content such as links (see Figure 4.7) or screenshots that are not fully accounted for in this analysis.

In the CPGT group the Livingstone Landowners Group (2), CPAWS (1), Niitsitapi Water Protectors (1), and the Piikani Nation (2) were mentioned within posts and comments. The context

¹⁷ It is crucial to note that there is great complexity in the stances various Indigenous groups took on the Grassy Mountain mining project, both officially and unofficially. To do this topic justice given my current research design, a thorough exploration of this topic would move this work far beyond a Master's thesis. However, this would be a valuable topic for future research.

in which these organizations or groups were mentioned differed from PARH, as posts and comments were primarily critical of the external actors and used the CPGT page as a platform to vocalize disagreement. One exception is references to Indigenous groups, which were neutral and only mentioned within the context of consultations regarding the project or potential harm that might come as a result of the mine closing, not as an autonomous external actor.

4.3 Conclusion

The Chapter 4 analysis of general group dynamics sheds light on the varying levels of activity and perspectives within the Facebook communities examined. The emergence of PARH (Protect Alberta's Rockies and Headwaters) as the first group in response to the Grassy Mountain mining project resulted in a rapid increase in its following. CPGT (Crowsnest Pass. Ghost Town) can be seen as a countermovement to the more mainstream PARH group, established about a month later. CPGT often directed criticism towards anti-coal activists and openly expressed discontent with posts and users in the PARH group. In contrast, PARH focused more on direct calls to action and political content, while CPGT placed a higher emphasis on personal experiences, as evident in Figure 4.5b and 4.7. Both groups shared a variety of informational content. The breakdown of comments and posts across groups revealed that the engagement related to the Grassy Mountain Project was not necessarily proportional to the number of followers a group had. This is evident in the cases of AACP (Around and About Crowsnest Pass) and CPU (Crowsnest Pass Uncensored), which, despite having significant followings, had minimal posts meeting the search criteria. It is important to note that neither AACP nor CPU were specifically dedicated to discussing coal mining in Alberta, as indicated in Table 4.1.

Political and regulatory decisions, particularly those related to coal mining on the Eastern Slopes, also impacted posting patterns. For instance, the reinstatement of the 1976 Coal

Development Policy on February 18 triggered notable shifts in group activity. Internal group dynamics were also influential, as the groups demonstrated awareness of each other's existence and occasionally made references to one another. Posting spikes in groups like CPGT or CPU could potentially be responses to increased activity within the PARH group, indicating a level of intergroup influence. These observations highlight the interplay between external events, political decisions, and internal dynamics in shaping the posting trends across the examined Facebook groups. Understanding these factors can provide valuable insights into the dynamics of online communities and their responses to various stimuli, contributing to a more comprehensive understanding of digital activism and social media engagement. Furthermore, the analysis highlights a larger network of organizations within the PARH movement, indicating the presence of connective action among various groups working towards the same cause. This networked approach allows for a broader reach and amplification of messages.

One interesting aspect of this analysis is where blame for the perceived issues is placed within each group. CPGT had minimal posts explicitly conveying political content, but anti-coal activists and Corb Lund were often singled out as the problem. On the other hand, the PARH group predominantly discussed and assigned blame to coal mining companies and the UCP, highlighting concerns of corporate greed and political corruption. Interestingly, a secondary sentiment of discontent and distrust towards the federal government was present in both CPGT and PARH, which has also been reflected in the works of Suhay and their colleagues (2022). These observations underscore the diversity of perspectives and narratives within the Facebook groups, reflecting the complex dynamics of the Grassy Mountain Project and the broader discourse on coal mining in Alberta. Furthermore, the analysis highlights a larger network of organizations within the PARH movement, indicating the presence of connective action among various groups working towards the same cause. This networked approach allows for a broader reach and amplification of messages.

These findings align with the concepts of "cloud networking" proposed by Milan (2015) and at its foundation, "connective action" as discussed by Bennett and Segerberg (2012). The prevalence of customized content on Facebook, with text being the most common type of post and users frequently reposting content, supports the idea that content personalization is a key element of online social movements. To some extent, users also used hashtags to personalize content and though likely don't serve the same structural purpose on Facebook as they do on platforms like Instagram or Twitter, they were used by group members as a means of displaying support and solidarity (Bennett & Segerberg, 2012; Halpern et al., 2017, Xu et al., 2021). Chapter 5 builds upon this discussion, using participant interviews to examine personal accounts of interacting and creating personalized content.

Another set of key findings within Chapter 4 suggest that structure of Facebook itself plays a significant role in shaping user behavior and interactions. Links are less popular compared to other forms of content, likely due to the additional effort required for meaningful engagement with external sources. This aligns with the design of ad-based social media platforms with commercial interests, including Facebook, that retain users within the platform and encourage continuous scrolling. As a result, posting links to other content is a more convenient and less time-consuming option for users. These insights into the structure and dynamics of Facebook highlight the influence it has on user behavior and how people engage with content,. Understanding these patterns is essential for comprehending the impact of social media platforms on information dissemination, activism, and public discourse.

Selective participation and the phenomenon of social media echo chambers are evident in the examined groups. Groups specifically dedicated to the issue of coal mining, namely PARH and CPGT, exhibited more 'likes' and 'shares' of posts compared to comments. This suggests that while

there is a general agreement with the content shared in these groups, there is relatively less active discussion occurring on the topic of coal mining. This finding aligns with existing literature on online political discussion, which highlights the tendency for like-minded individuals to reinforce their beliefs without engaging in substantial dialogue (Heyman & Pierson, 2015). AACP, a community group for the Crowsnest Pass area and CPU, a community page emphasizing freedom of expression, offer contrasting examples. AACP's strict rules against unrelated, political, or negative content resulted in the deletion of posts and the banning of users who violated these guidelines. Consequently, only one post discussing the Grassy Mountain Project was retained in the group. In response, CPU was created as an alternative space where "anything goes" (Table 4.1) and censorship was no longer an issue. However, despite its more permissive environment, CPU did not become a central hub for protest or discussion on coal mining. Instead, the group primarily focused on Covid-19-related content. Furthermore, the observations regarding PARH and CPGT indicate the potential formation of echo chambers within the Facebook groups. On PARH, comments opposing the group's beliefs were deleted by the admin under the premise of violating community guidelines. However, the data presented in Figure 4.5a suggests a significant proportion of posts consisting of 'political content', with over half specifically targeting the UCP and reflecting a strongly negative sentiment. Interestingly, certain comments and posts that could be considered unkind towards procoal actors were not deleted, despite going against the stated community guidelines. The existence of censorship and the removal of posts from pro-coal advocates created conflict within the groups and prompted the establishment of alternative platforms such as CPU and CPGT in response to perceived censoring activities.

These dynamics also raise intriguing questions about the context of Covid-19 and its potential influence on concerns about control and freedom of speech. The observed conflicts within online spaces shed light on the processes and factors that contribute to individuals seeking niche

environments where they feel they can freely express their opinions, and how the online space in which occurs directly or indirectly facilitates these effects. Through in-depth participant interviews, Chapter 5 further explores some of these key themes including types of engagement, the greater activist network, the politics of Facebook as a platform, and how Covid-19 shaped all of these elements.

Chapter 5: Interview Findings

To further supplement social media analysis findings and provide deeper insight into the quantitative data discussed in Chapter 4, this Chapter presents the perspectives of individuals that participated in online political activism regarding the Grassy Mountain Mining Project throughout the Covid-19 pandemic. The interviews conducted paint a contentious story set in contentious times. First, the personal experiences of participants are examined, from motivations behind their involvement to the perceived impact of online activism. Larger processes within mobilizing an online political movement are then discussed, with a focus on both internal and external network building. The latter half of this Chapter focuses on two themes which emerged primarily through interviews with participants: the impacts of the Facebook platform structure on online activism, and the impacts of the Covid-19 pandemic on user's interaction with political settings in the online sphere.

Social media users who posted on at least one of the four Facebook groups in the social media analysis were invited to participate in semi-structured interviews conducted via WebEx, which were then analyzed through a thematic analysis. The goal was to obtain six interviews split across the four groups in proportion to their following. Unfortunately, not all group admin responded to the interview request and as a result, only four interview participants were recruited. As such, not all groups are represented proportional to their sizes. Due to the relatively small number of active posters in the groups, group affiliations are excluded from this analysis to preserve the anonymity of participants (refer to Chapter 3 for more information regarding methods).

5.1 Individuals as the Unit of a Social Movement

People across Alberta used online platforms to advocate for their political interests regarding the Grassy Mountain Project. In the interviews conducted, the locations of participants spanned both rural and urban regions across the province. When the 1976 Coal Policy, which designated various levels of protection against strip mining in Alberta's sensitive ecological areas, got rescinded by the UCP in late May 2020,



Figure 5.1. A word cloud created from interview data showing the most used words across all participants. The size of each term is proportional to the number of times it was mentioned. Font colour was assigned arbitrarily and does not represent any findings.

the decision got coverage through both traditional and non-traditional forms of media (news outlets and social media, respectively). Of the interview participants, equal numbers remember first hearing of the Grassy Mountain Mining Project through traditional media outlets, such as the CBC, and social media platforms. And all but one interview participant specifically became interested in the issue due to the 1976 Coal Policy being rescinded, with the final interview participant not specifying their motivation behind first encountering the project. However, participants first encountered the Facebook group in which they participated through either friend's social media channels, in a general internet search hoping to gain more information about the project, or through direct conversation (either online or through a phone call) with other people. This section first explores interviewees' motivations for both following the group and for sharing political content. This is followed by an exploration of interview participants' thoughts on the perceived impact of their political actions regarding the Grassy Mountain mine policy decision and expected involvement moving forward. Finally, this section concludes with the personal benefits and drawbacks participants feel they gained through their time engaging with this movement.

5.1.1 Motivations for participating

When asked about their personal experiences and motivations behind posting political content regarding the Grassy Mountain Mining Project, participants described a complex picture of their online activity. This included conversations describing an emotional connection to the issue, dissatisfaction with both federal and provincial governing bodies, and a desire to both obtain and generate accurate information about the topic. Participant responses regarding their motivation for both following and participating in a Facebook group regarding the project were relatively consistent between users, generally falling into a combination of the factors listed above.

All interview participants voiced feelings of a personal and emotional connection to the cause or the space itself as a primary reason behind following one of the four Facebook pages discussed in this thesis. The role and value of the physical region in identity-building is at the center of advocacy efforts related to the Grassy Mountain mining project. Speaker 1 describes feeling motivated to follow the page to advocate for the issue;

"...we see ourselves in the Eastern Slopes... that's the way we see ourselves in Alberta. This landscape...mountainous foothills, you know gently grading into the prairies is, it's kind of caught up in our identity." (Speaker 1)

Personal identity as a motivating factor for following Facebook pages regarding the Grassy Mountain Mining project is present both directly and indirectly in the above quote. Speaker 1 describes how the unique landscape of Alberta is part of not only their identity but is the basis of a collective identity within the province, described through language such as "we see ourselves" or "our identity" (Speaker 1). A similar sentiment with regards to personal identity built upon an emotional connection to the physical space is expressed by Speaker 3 when asked about their motivations for participating in the movement;

"[M]y province is really beautiful, and I just refer to it as my province cause this is where I grew up. I mean, I grew up, I spent my whole entire life here... There is nowhere in the world that is like Alberta. It is one of, it is one of its kind." (Speaker 3)

Again, a sense of identity for Speaker 3 is rooted in the physical space that is the province of Alberta, evident from the use of the language "my province", that within the context of the interview implied not ownership, but instead a sense of responsibility and kinship. Speaker 3 also captures the ability of the conceptualization of spaces to hold meaning free of time, through the speaker's clarification of their language in the possessive form (i.e. "my") to capture a lifetime worth of assigning meaning to those spaces. This links directly to Soja's work and the idea that the conceptualization of any given space is constantly deconstructed and reconstructed across time, as the meaning it holds to individuals and collectives is enacted (Gamson, 1990; Soja, 1998). Sharing experiences such as the ones of Speakers 1 and 3 captured above, in this instance through conversation in interviews and online posts, exemplify the enactment of meaning that the physical space of Grassy Mountain and the eastern slopes holds for these individuals.

Across all interviews, the motivations for following social media with the intention of protecting political interests is framed as synonymous to protecting the collective identity they have built upon the space they inhabit. This observation also provides further insight into the social media analysis presented in Chapter 4¹⁸, where posts that received the most interactions from users were ones that contained both user-created text and some form of visual, such as a landscape photo. Interview findings regarding the strong connection and value participants put in the physical space aligns with the postulations that visual content where the user can directly see the physical space

¹⁸ Refer to Section 4.1 and Figure 4.4 for further detail.

under discussion evokes a greater emotional response and inspires action (Bowler et al., 2010; Xu et al., 2021). Access to information regarding the project was also a factor that played into participant's motivations to follow the group and ties into previous reports of coming across the Facebook group in hopes of learning more about the project. Participants also reported that following the group provided them with reliable and up to date information and updates on the project. Simply put,

"Policy was still being looked at, changes were still being made and you could find out about it from the comfort of your home." (Speaker 4)

Generally, participants reported feeling a lack of trust in decision-makers and often, the sources of information provided by mainstream media, such as news outlets. Speaker 1 and Speaker 3 share their motivations for following the group(s);

"I felt that [the Facebook group] actually, it has helped me understand what exactly was going on ... I find sometimes with, uh, the mainstream media and even in, local media that it can be hard to find certain things unless you're actively looking for it. This actually helps streamline the information for me." (Speaker 3)

"That was the draw that brought all these groups together and made so many people interested in this site. They were also looking for, for a source of information. You could get good information." (Speaker 1)

Speaker 3, who took a more passive role in their group in comparison to other participants,

describes the value of this group in consolidating and verifying the information posted in the group.

The idea of using social media posts to verify that the information is reliable may seem

counterintuitive as not all things presented as facts on social media are true. However, given that the

users within the group can comment and/or react to posts, this serves as an informal mass review of

the content posted. This is described by Bennett and Segerberg (2012) as the ability for users to

indirectly influence the visibility of a post through an informal "vote" of giving the content either a

positive reaction or a negative reaction. Speaker 1 also echoes the view that these groups were

valuable in so-called "good information" and beyond this, believes that it was the search for

information on the subject that inspired people to join that group. The latter observation is also
consistent with various literature on online political participation, which suggests that passive attention to political content, such as reading news articles, is closely linked to information dissemination action and it the more passive levels of involvement that are commonly the beginnings of individual involvement (Cantijoch et al., 2016; Boulianne, 2019).

When asked specifically about what motivated participants to actively engage¹⁹ in posting content related to the Grassy Mountain Mining project, the responses differed slightly from interviewees' motivations to just follow the group. Most participants felt that they needed to contribute accurate information to the group they posted in and battle perceived misinformation either in mainstream media, across other groups (pro vs anti coal mining), and/or internally within the group they were participating in. Speaker 2 describes their motivations stemming from the desire to "help making sure some of the people I knew were aware of the issue." (Speaker 2). The idea of battling misinformation is also exemplified in Speaker 1's account of an instance where information perceived as false was posted online and "you'd see everybody [group members] pick apart their ridiculous fallacious arguments" (Speaker 1). This is closely linked to another motivator for political participation-holding governing bodies accountable for their decisions and the ability to publicly express discontent with both the actions and views of political figures via public social media pages related to the issue and the official Facebook or Twitter pages of governing officials. (Casero-Ripolles & Pepe-Oliva; 2022).

5.1.2 Perceived impact of activist actions

During the discussion of personal experiences as a participant within one of the four groups, the interviewees were asked to reflect on how their motivations evolved in the past year and a half.

¹⁹ As opposed to passive forms of engagement, such as reading articles without posting.

Perceptions on the overall impact of online activism varied greatly between participants, however the general sentiment throughout this discussion was the lack of meaningful progress or resolution of the issue and connections between current governance structures and decisions. This can be seen in the quote below;

"...I don't know if it's changed at all... nothing has been accomplished, absolutely nothing...Um, so the goals haven't changed. Um, my resentment, my anger has gone well beyond the, uh, provincial government to the federal government now. Uh, so that's changed for me personally." (Speaker 1)

Interestingly, all four participants drew parallels between government decisions throughout the Grassy Mountain Mining Project evaluation and other recent provincial and federal political events. For example, various perspectives on the change of Alberta's premier under the United Conservative Party (UCP) and the proposed Alberta Sovereignty Act were voiced, events which both occurred shortly prior to the interviews being conducted. Speaker 2 states;

"[M]y motivations haven't changed. I'm disappointed that it feels like we have to start all over again with a new premier. So, my motivations haven't changed. I'm just disappointed that the progress seems to, to be for naught now. I don't know. I could, I could be proven wrong, but everything is pointing against that." (Speaker 2)

The general agreement across interviews echoed the opinion of Speaker 2. Even in cases when the participant felt their original goal regarding the project has been accomplished, the overall sentiment was that a threat still exists and activist actions are still required. It is important to note that changes in provincial government leadership without a re-election and the introduction of the Alberta Sovereignty Act occurred in a similar timeframe as the interviews and had media attention at the time, likely these events and their implications at the forefront of interviewee's minds (Bennett, 2022). This observation provides an interesting example of interactions between various forms of both traditional and non-traditional media and the dynamic nature of online activism (Casero-Ripolles & Pepe-Oliva; 2022).

5.1.3 Personal benefits and drawbacks of social media use

All individuals interviewed spoke about spending significant amounts of time on social media platforms throughout the period in question. As part of conversations about their time spent online, participants reported experiencing both significant benefits and drawbacks because of their use of social media. For the purposes of this question, social media was defined as 'any online platform where you can directly interact with other people'. Though participants focused primarily on their experiences on Facebook, the question was intentionally left open-ended to gain a better understanding of the value participants gain from the use of different platforms and how this relates to their use of Facebook.

Three out of four interviewees were asked specifically about personal drawbacks and benefits to social media use during the period in question (due to participant time constraints, this question was unfortunately not asked to one of the participants). The benefits reported from the use of social media were closely related to the motivations for participating in social media activism. Increased access to information from sources that are perceived as reliable, improved skills such as writing and critical thinking²⁰, ease of contact with friends and family, and a sense of community were all benefits interviewees felt they gained from their participation in social media during the time period in question. Previous academic works on individuals perceived benefits of social media use throughout Covid-19 report similar experiences to those interview participants describe, with two major benefits being access to information on current events and feeling connected to friends and

²⁰ Critical thinking refers to participants' reports of feeling better equipped to process, question, apply, and pick out inconsistencies in the information they encounter. This topic which was not explored in depth within this project due to scope but is a fascinating direction for future research.

family (Saud et al., 2020; Igartua at el., 2020). Given that this research focuses on late 2020 to mid-2021, it is also important to examine these experiences within the context of Covid-19. Section 5.5.0 dives into this further and is dedicated entirely to participant's experiences and relationships with social media during the pandemic and ensuing public health policy.

Perhaps one of the most interesting interview findings was that three out of four participants described the major drawbacks of their social media use to be personal- namely, the impacts of social media use on their personal relationships and on their well-being. When asked the downsides of their social media use, Speaker 4 states;

"Well, I hate social media (laugh). There's always drawbacks. The biggest drawback is my husband sitting there at night watching a TV saying, 'will you get off your phone'... he doesn't understand how important it is." (Speaker 4)

This sentiment was far from uncommon, with Speaker 2 describing their experiences on social media as "...pretty exhausting" and "Oh, God, you can waste so much time...at some point I go 'What am I doing?" (Speaker 2). Speaker 1 was not asked directly about drawbacks, though they mentioned that they joined Facebook specifically to advocate for their views on the Grassy Mountain Mining project and within this context, expressed that they "...look forward to being off Facebook and never joining it again" and that "the value [Facebook] offers isn't all that great" (Speaker 1). These quotes highlight the two sides of participants' experiences with social media as both a powerful tool for communicating views and beliefs but at the same time, a tool which does not prioritize the user's well-being or best interests.

5.2 Social Media Processes: Building a Network

A social movement, regardless of the form it takes, is the product of individual motivations and goals coming together to form a united collective (Gamson, 1989; Bennett & Segerberg, 2012; Kavada, 2015). This can be observed in both the social media analysis discussed in Chapter 4 and

the interviews conducted with individuals who participated in the online groups. Within these collectives, distinct processes emerge along with a variety of forces that continuously shape the movement (Shafi & Ran, 2021). This section focuses specifically on the relational elements of the Grassy Mountain mine activist networks. For the purposes of this discussion, relational elements are used to refer specifically to the properties of connections between/within actors of a network, as opposed to technical elements which refer to how the system (i.e. the blend of online and physical space) in which the network is situated influence these connections (technical elements are further explored in section 5.4.0). Based on interviewee's accounts and literature on online political mobilization, these can be loosely categorized into two organizational forces- the relationships and actions shaping the movement from within and the relationships and actions shaping the movement externally through the larger activist ecosystem (Bennett & Segerberg, 2012; Milan, 2015; Brescia, 2020). Within the movement, this includes various group member's relationships with each other, the issue, and the activist actions themselves. External relationships impacting the movement include the greater activist network, the policy and political landscape, and the space in which political advocacy or protest occurs. This first half of section 5.3.0 focuses on the various actions and tactics which participants utilized to communicate their perspectives and get the attention of governing officials. The latter half of this section discusses how interview participants perceived the roles of various key actors and their relationship within the larger activist community network, with the intention of shedding further light into the activist network introduced in Chapter 4.

5.2.1 Tactics and processes within advocacy groups

Regardless of interviewees' perceptions on the overall impact of their engagement or the outcome of the collective activist action, participants were asked specifically about their thoughts on the general structure of the online movement and which specific activist tactics within the group

they felt had the greatest impact in motivating other group members to act and on the ultimate policy decision regarding the mine. Multiple speakers (Speaker 1 and 4) mentioned that from their experience, some types of content are more effective than others. Posts which could motivate action or resonate on a deeper level with page followers are most effective in online political activism, as they create a united front and coordinate individuals act in an organized manner to communicate the group's political message effectively.

"I think the least effective posts are, um, people just ranting, like I'm doing, about the government. That's not very effective at all. Um, I think you need to give people a nice factual story with, with a, uh, a bit of a, uh, a pull on the heartstrings." (Speaker 1)

In the above quote, Speaker 1 also highlights their view that "people just ranting" are the least effective form of online activism. Instead, group members prefer to interact with content that relates to them specifically. This is reflected in literature on framing within online social movements, such as the recent work of Ackland and O'Neil (2020) which finds that framing in online settings is a key element of online activism and has the biggest impact on individual actions when coming from weak-tie networks.

In terms of specific content, the ease of the action was a key consideration when actively engaging with online content, "because people like to do quick, easy things." (Speaker 4). Specifically, pre-written letter templates were well received by participants, including Speaker 4 who states, "those, those auto-generated emails. They are brilliant, a brilliant online tool.", and Speaker 2 who further elaborates on the value of this tool;

"Well, I do believe the action items they posted were, were good, well received by me, and I do believe well received by a lot of people because they often gave you a template of a letter that you could use. Um, sometimes I use the template, sometimes I decided otherwise and chose my own wording." (Speaker 2)

The experience of Speaker 2 being able to "decide otherwise and choose [their] own wording" is a key element of personalization within connective action (Bennett & Segerberg, 2012). Both quotes

also touch on the types of online political engagement that feels effective to participants- ones that are quick and feel impactful while requiring minimal time or effort. More broadly, the attitude perceived to be most effective in online political engagement within the groups among interviewees is captured briefly, yet effectively in the following statement;

"I think that that's a hopeful thing for me, that people seem to be looking for nice messages... this was, if anything, the biggest show of what people coming together can do." (Speaker 2)

The observation that people within the activist groups seem to be looking for "nice messages" and unity was also echoed by Speaker 1;

"We worked very hard to try to respect the other point of view when it was respectful in turn. I think that probably led to a lot of success in the group actually, that could include, um, disparate politics in it in a reasonable manner." (Speaker 1)

Both speakers attribute at least part of the success of their group(s) to the positive attitude of its members and seemingly, a desire to create an online space where participants felt empowered to share their perspectives and engage in discussion. For the most part, respectful types of engagement were considered by interview participants to be the most effective in communicating political views. However, when looking at the full snapshot of discourse on social media there are numerous other factors at play, some of which are discussed in the following section and section 5.4.0.

5.2.2 Greater activist network

Discussion with interview participants showed further insight into the greater activism network regarding the Grassy Mountain Mining project and its various actors. Due to the large group of anti-mine advocates and the highly organized nature of the affiliated groups, the majority of the organizations or collectives discussed by interviewees were advocating against the mining project. The overall activism effort against the Grassy Mountain Mining project consisted of various actors that filled distinct roles. For example, multiple interviewees described the role of CPAWS as the primary organizer of letter or email writing campaigns, including pre-written templates that individuals could either customize or just fill out with their information. One interview participant describes:

"...people like CPAWS who are doing the quick emails where you could literally just click on it and it would tell you your MLA... you could just like send it or you could just make little changes to it and then send it." (Speaker 4)

Speaker 4 captures some of the external network structures behind providing group members some of the tools for political participation discussed above. CPAWS is a more established organization-their Southern Alberta Chapter has been running a variety of campaigns throughout Canada since 1967 (CPAWS, 2023). The ability of the organization to provide tools capable of auto-generating emails that are customizable and accompanied by an email list of relevant political figures is one that significantly decreases the energy required for individuals to participate in online activist action. However, setting up an online system capable of automating email writing campaigns is also a resource-intensive process and one that is likely not a possibility for other actors within a movement, who may not have the hard or soft resources required to provide such tools, especially at early stages (Ackland & O'Neil, 2020).

Grassroots organizations that are primarily online, such as PARH, can provide other resources to larger or more established organizations. PARH gained a large and engaged audience spanning the province in a short amount of time with relatively few resources. CPAWS providing access to their tools or the Niitsitapi Water Protectors hosting informational webinars were both activities which were communicated at least in part through PARH. This can be seen as an example of the relationship between various actors within a greater movement, including the strategic sharing of resources between organizations to maximize their impact. Consistent with the literature on online social media movements, both the social media analysis findings in Chapter 4 and the interviews presented in this Chapter position the PARH group as a major communication venue for other organizations within this network, such as CPAWS (Bennett & Segerberg, 2012; Ackland & O'Neil, 2020).

Another dimension of the activist networks described by interview participants was not based on formal organizations themselves but on the interests of various actors. According to interview participants, this occasionally resulted in internal conflict within groups- even when they held the same ultimate political goals regarding the Grassy Mountain Mining project, Speaker 1 notes that "there's lots of room for everybody to fight everybody". Across interviews, hunters, conservationists, ATV users, industry, and ranchers emerged as some of the primary interest groups advocating for their political interests regarding mining in Alberta. The Livingstone Landowners Group (LLG) was discussed in some capacity throughout all interviews as a collective representing the interests of land users. Of this group, a significant proportion was described by interviewees to be the ranching community since most pastureland is downstream of the proposed mine and would impact water quality in that space. Within both the pro- and anti-mine activist networks, the ranching community was presented as a contentious player who had strong political sway regarding the issue but whose motivations were questioned by other actors both within and outside of the movement. The following quotes capture some of the complex and at times, conflicting, views the ranching community evoked within interviewees;

"I just kind of think it's ironic that... there's damage from everything and [the ranchers] kind of overlook the damage that they do, but I'm very grateful they were involved. They're very much feeling like they, um, are totally responsible for having stopped all this. They were super instrumental, there's no question, but they weren't alone." (Speaker 1)

"The ranchers involved was, was huge because, you know, they're the UCPs normal base, even though it seems that they haven't really pivoted as much as they should have done." (Speaker 4) The general attitude of the interviewees towards ranchers was a distrust of the political power the ranching community as an advocacy group holds within Alberta, through their possession of both political and monetary resources. The point of conflict for a variety of activists across all views on the issue of coal mining was that much of the prosperity within the ranching community was obtained through practices that are also environmentally destructive, which the ranching community seemingly failed to acknowledge. Some interviewees also expressed frustration with the views of the ranching community on the outcome of the project, saying that they felt the ranchers took most of the credit for the Joint Review Panel's decision to recommend against the project, failing to credit the variety of other groups and individuals who also put extensive effort and resources toward advocating for the issue. Tensions within the movement exemplified through the ranching community also raise interesting questions about the social dynamics within an online movement where individuals of diverse backgrounds, and the processes through which conflicting views on an issue are reconciled to work towards a shared outcome (Saunders, 2013; Kavada, 2015).

5.3 Politics of the Platform

The structure of Facebook as a social media platform and the complex relationships that participants have with it was a reoccurring theme throughout interviews. This includes the design of the platform itself (ie the application functions and algorithms), the tools Facebook gives to page administrators and participants, and how this shapes user's interactions with, within, and outside of the platform. Interviewees encountered these attributes from multiple perspectives and though there is great diversity within their experiences, this section touches on some of the features of Facebook that participants feel shaped their political activism around Grassy Mountain. First, interviewees interactions with Facebook platform itself are explored. Once this is established, how interviewees utilized these tools and how this impacted their interactions with other users is examined. When

thinking about the findings presented in this chapter, it is important to note the online space we interact in is formed not only due to the features of the technologies we use (whether this be material or immaterial), but also because of users' constant, complex interactions across these features (Faraj & Azad, 2012).

When asked about the social media platforms interviewees used to participate in online activism around the Grassy Mountain mining project, Facebook was the most used platform (this is unsurprising given the research design). Other platforms such as Twitter, LinkedIn, TikTok, Instagram, and dating apps were also mentioned within the context of social media use. However, these were not top choices for participating in political conversation regarding the Eastern slopes, with Twitter being the only other social media platform used specifically for activist actions. In comparison to other social media, interview participant the structural features of Facebook to be advantageous in bringing people together for political advocacy, especially the option of creating a group where certain conditions must be met to enter, such as answering vetting questions. Another feature which helped organize collective action was the ability to create and share events through Facebook, which became especially relevant within a Covid-19 context (discussed in Section 5.5.0).

Along with the abilities to organize collectives through built in features like events and groups, interview participants viewed Facebook as a platform for political advocacy in a positive light when discussing the benefits of social media technologies in spreading messages across a wide audience. Two of the four participants mentioned the skill of being able to 'play the algorithm' as a key factor in the success of an online social movement. 'Playing the algorithm' refers to users familiar with how social media algorithms function and the skills to strategically influence the reach or longevity of online content. Speaker 3 describes;

"I sometimes feel like... it is a double-edged sword if, you know, if you have someone who's savvy can play the algorithms. Cause that's what I think it comes down to. If you

can create a certain algorithm that's gonna pop up and keep people seeing people's feed over and over again, that way they can actually get interacted." (Speaker 3)

In this statement, Speaker 3 also captures a deeper and more complex relationship with the platform through their description of the algorithm as a "double-edged sword" (Speaker 3). They further elaborated on the other, less positive edge of this sword later in the interview when describing their belief that "...the more right-wing and more negative or violent the algorithm is, the more likely people are actually going to see it on their feeds" (Speaker 3). This sentiment frames the algorithm and the overarching structure of Facebook as one that intentionally favours more extremist thinking and pushes less polarized content into the margins. Speaker 3 flags this as a disadvantage for activists or movements that take less extreme approaches to political advocacy. When describing functions of the algorithm and the platform itself (as opposed to the manipulation of it by users), other interview participants also viewed Facebook as a divisive tool that was frequently counterproductive to the purpose of the activist groups. Speaker 1 noted that it "seems like they are trying to build a controversial platform half the time. I don't know what's going on. I couldn't really tell you, but they certainly don't give us tools that are helping us". In this quote, Speaker 1 is referring specifically to features Facebook provides when using the 'Group' function that is intended for organizing and managing collectives. Multiple speakers using these features expressed frustration and a lack of predictability with the tools and structures provided. When discussing their experiences participating in online political groups, Speaker 4 expressed a similar sentiment, stating that;

"They give us new tools and then the next thing we know all these people would be included that we wouldn't normally include and then it seemed like when they made a comment from outside... if you posted it on the site, they were automatically included." (Speaker 4)

Examining these accounts of interacting with Facebook, it is interesting to take note of the language used to describe to online space in the above quotes, such as reference to "the algorithm" as right-wing or violent (Speaker 3), "they are trying to build" (Speaker 1), and "they give us new

tools" (Speaker 4) refer to Facebook as not only a place where interaction can happen, but as a personified or perhaps, sentient external entity that facilitates these interactions with its own motivations or interests. In the above statements, interview participants present their experiences with the platform as far from neutral, apolitical, or transparent. On the contrary, in discussion the platform and the algorithm itself were painted by participants as highly political, with a level of power being exercised over user interactions within the space. Looping back to Speaker 3's statement on users who can "can play the algorithms" it is implied that with enough skill and knowledge of the platform, users can manipulate Facebook's internal systems and gain back some of this control over the algorithm, getting it to work in their interest. Again, Speaker 3's account of "playing the algorithms" presents the platform as not a tool that is used, but with a degree of sentience that requires skillful and strategic manipulation to work as the user intends it to. This theme is also reflected within the literature, which generally supports interviewees' accounts of Facebook as a polarized space where the design focus is not on the average individual using it, but on independent and commercial interests (Milan & Velden, 2016).

Though we established that Facebook is a political space with both the internal and external structures of the platform influencing user experience and behaviour (Milan & Velden, 2016), we have yet to discuss its impacts on online political advocacy and collective action. All interview participants shared some of their experiences with Facebook features that directly impacted their or others' engagement in political participation within the platform. Consistently across interviews, individuals reported encounters being blocked or blocking others, a feature that can entirely cut off users' ability to participate in certain forms of online activity, to an entire lack of control over users' political participation within groups or pages. Most interviewees reported being blocked by at least one government official or political figure on social media, on both Facebook and Twitter, because of posts or comments they have made on their pages related to this issue. However, some

interviewees have also experienced the blocking feature from an administrative side, in both the groups examined and in other instances throughout their time on social media.

5.4 Covid-19, Social Media, and Political Advocacy

Physical space is at the root of collective action surrounding the Grassy Mountain mining project. As was discussed in Section 5.2.0, interviewees describe the meaning that Grassy Mountain and the eastern slopes hold for them to be a key motivator in advocating for that space, often tied back to their personal encounters with that area and strong beliefs around how the value of that space. Grassy Mountain and the eastern slopes being at the center of this movement makes it the case study especially interesting given that it is situated within the context of a global pandemic, when existing in physical space was different than ever before. Behind all these elements, there is one underlying theme that has not yet been explored- a shift in how participants relate to and exist in the spaces they occupy. The purpose of this section is to explore how participants navigate online vs physical space during this time, how they have adapted to the circumstances, and the impacts of this shift on their political participation.

When asked about their social media use prior to the pandemic versus throughout, all participants responded to the question without hesitation. Most commonly, interviewees tied their increases in social media use during the pandemic back Covid-related shifts in personal motivations for participation, such as access to information and a desire for social connection. Speaker 2 stated that "[i]t's definitely gone up, um, cause if you couldn't go anywhere, you couldn't meet people. Social media was a way to connect or find out what's going on." (Speaker 2). Speaker 3 also ties their increase in social media use to the combination of a desire to keep in touch with friends and family and working from home during this time period, which made devices such as a cellphone or personal computer more accessible throughout the workday. Speaker 4 exclaimed;

"Oh my goodness. I think it's definitely tripled (laugh)... I've been working very hard to get that screen time to say your screen time has dropped by 14% this week instead of your screen time's gone up by 50% (laugh)." (Speaker 4)

Speaker 4 also attributed their increase in time spent online to their social activities being done virtually due to safety concerns. In addition to the influence of the Covid-19 public health measures on social media use, Speaker 4 also drew an explicit connection to political activism and their social media use beyond the pandemic, stating "...it's gone up since I became an activist, to be honest" (referring to a time prior to the pandemic). They also note that in the context of political activism, their patterns in social media use depend largely on the political landscape, with increased time spent online during periods of organized protest activity. It is interesting to note that a common sentiment throughout both the social media analysis and interviews is a distrust in governing bodies during time of crisis and a belief that controversial or harmful policy choices will be made while the general public's attention is on the pandemic. (Grant & Smith, 2021). Though Speaker 1 was not directly asked about their social media use throughout the pandemic due to time constraints, they were asked to share thoughts on their involvement levels in similar online settings moving forward into a (hopefully) pandemic-free future. Speaker 1 suspects that there will be less political involvement in online activities regarding coal mining on the Eastern slopes than there was throughout the pandemic.

Finding ways to overcome barriers towards engaging in physical protest was a key component of the Grassy Mountain mining project activism for interview participants. Three out of four interviewees participated in some form of physical protest between December 2020 and August 2021. With public health measures changing multiple times throughout this period, interview participants found creative ways of engaging in traditional forms of protest. Speaker 4 observed that in their time spend in a protest environment during Covid-19 "often starts with the online…most of our protests have started by something that's been said in one of our groups and then, you know,

you take it on and it snowballs from there." (Speaker 4). They went on to recall one experience of organizing collective action during the pandemic and related restrictions;

"We were sitting in a chat group thinking, what can we do? We can't go for a protest... we came up with ... we'd get out one by one and go chalk our message and get back in our car. And then when that person got back in the car, the next one, you know, we'd just like do this rolling thing... but that was started online totally." (Speaker 4)

Chalking refers to the act of writing messages in chalk in public spaces. This account exemplifies an adaptation of the traditional protest, while still working within the public health restrictions of existing in that space.

In the experience of Speaker 4, online spaces (in this case, Facebook) were used to organize and facilitate aspects of physical protest. This presents an interesting observation on relationship between the uses of physical and online spaces during the pandemic, where in this case, existing and participating in online space is a pre-requisite for navigating physical space (Tufekci, 2018). Speaker 1 also organized individuals within their network in an outdoors, rural space overlooking the eastern slopes, and "…had quite a little chat and that was very emotional and I think, it, it showed some real concern" (Speaker 1). This speaks to the impact that placing oneself within a physical space can have on perceptions around the issue. Speaker 1's description of this gathering as "very emotional" further emphasizes the personal value assigned to physical space that was described as a key motivator for participation across multiple interviewees in Section 5.2.0. Finally, Speaker 2 described their experience participating in a march organized to protest the Grassy Mountain Mining project and meeting new people through this experience;

"One lady who I marched with in that parade, cuz I had a great big, huge sign. It was four, five feet wide, so I needed somebody to carry it with me. So, um, I met her in the group to help carry the sign." (Speaker 2)

This march was also organized online, in collaboration with another advocacy group. This quote highlights an online component to the traditional protest forms, where the online network formed through the Facebook group enabled Speaker 2 to get assistance holding the giant sign and thus,

participate in the march in a way that may otherwise have not been possible or at the very least, been more difficult. Given the interviewees' experiences, even when protest was happening in physical space, Facebook was a focal point of organization efforts, information sharing, and valuable in situating individual experiences within this network to form a collective. However, Speaker 4 offers an interesting observation regarding their thoughts on the long-term impact of online versus physical action;

"I participated in one of those protests downtown and, um, that had a short-term efficacy. But I believe the fact that the, the thousands of people that follow that page is visible at any point in time makes it seem, it's more long term." (Speaker 4)

Though conversations with participants describe collective action within physical space to be an invaluable component of the Grassy Mountain Mine project social movements, Speaker 4 raises a key argument that, highlights a major benefit of online political action. In theory, the information posted online is eternal and moves beyond time, but in practice, this is often not the case.

5.5 Conclusion

Chapter 5 wraps up the analysis of the rejected Grassy Mountain mining project through indepth interviews with individuals who were active on the Facebook pages of interest during the period of this case study (December 2020-August 2021). These conversations were an excellent supplement to the social media analysis and together, provide insights into the objective at the core of this thesis- how Covid-19 and the subsequent increase in the use of social media as a communication tool has shaped how individuals advocate for their political interests in the online sphere and how this translates into change in physical spaces.

In Section 5.2.0, interviewees emphasized that it was the personal meaning and sense of identity attached to Grassy Mountain and the Eastern Slopes which drove advocacy efforts. Supporting this observation, the most effective posts were described as resonating with users on a deeper level, as this was believed to create united fronts. There were also significant drawbacks to participating in groups regarding the mine, with the majority of participants describing a negative personal relationship with Facebook. This tension of Facebook as a powerful communication tool that users don't plan on leaving, even when the negatives of its use may not outweigh the positives, hints at a theme which previously emerged in Chapter 4 and constitutes a key finding within this research as a whole- the relationship between a seemingly neutral, apolitical platform and its users.

Though the overall experience with Facebook use on a personal level was negative, interviewees also reported a variety of positive outcomes of their time spent on the Facebook groups in question and social media more generally. Keeping up to date with current events, contact with friends and family, improved writing skills, and the perceived ability to critically examine online content were all reported as benefits of social media use, especially within the context of Covid-19. Improved critical thinking skills were an especially interesting positive outcome of engaging in online political groups, as one person's definition of what constitutes a reliable source may be seen as misinformation by another.

The various ways that Facebook as a platform impacts user experiences and interactions within the online sphere was intertwined throughout nearly all segments of participant interviews. Working off these accounts, Section 5.4.0 focused on how Facebook's structure and features influenced political activism related to the Grassy Mountain mining project. User experiences with the design and features of Facebook (including its algorithms) was generally described as unpredictable from both a group administration and group member perspective, which resulted in various issues such as individuals who did not pass screening questions entering groups or vice-versa, group members being kicked-out or banned for posting content which may have been acceptable in that group the week before. Within interviews, another major concern arose about

Facebook's algorithms favoring extremist content and providing divisive features that hindered activist efforts. The interviewees' observations regarding these features are backed by literature (Milan & Velden, 2016). Furthermore, participants shared experiences of being blocked by government officials or administrators, highlighting limitations on political participation and conversation within Facebook. This view of Facebook as a politicized space aligns with existing literature and is one of the primary topics of discussion within Chapter 6.

The analysis provided in Section 5.3.0 also showed that, consistent with existing literature, quick actions that are easy to personalize (such as letter templates) and positive messages which emphasized unity within the movements were seen as most valuable. The Chapter 4 analysis examining the activist network regarding the Grassy Mountain Mining project was discussed by interview participants in greater depth and highlighted the roles of various actors and organizations. Interviewees highlighted conflicts arising from different interests within actors on the same side of the movements. The Livingstone Landowners Group (LLG), a pro-mine collective consisting largely of the ranching community, caused controversy as they were described to hold disproportionate power within the pro-mine activist network due to their significant political influence and historically right-leaning votes. These tensions within the movement shed light on the complexity of social dynamics and conflicting perspectives within a social movement, even when members of an activist network had same ultimate goals.

The Grassy Mountain mining project case study was largely of interest due to the context within which the mobilization of political interests occurred. Overcoming these newfound barriers to traditional forms of protest, participants found creative ways to engage within the restrictions that were placed on existing in physical space at this time. The analysis within Section 5.5.0 shows that participants still highly valued physical activist action during public health restrictions and felt that in-person engagement led to stronger, more emotion-based connections to the cause. Regardless, online spaces have become crucial for organizing and facilitating these physical protests, with all participant accounts of physical activist action being organized online via Facebook.

Chapter 6: Discussion and Conclusions

The past three decades have witnessed a digital transformation that has profoundly impacted various aspects of our lives through increased accessibility to the internet, the proliferation of personal electronic devices, and digital communication. Additionally, the Covid-19 pandemic has further accelerated the integration of technology into our daily experiences (Stahl, 2021). This blurring of the line between the physical and virtual realms has also led to a reimagining of political advocacy and mobilization. Given this context, the research in question focuses on the intersection of political participation, social media, and the environment. Specifically, it examines the tensions surrounding environmental political participation in physical spaces and how such participation is conceptualized and enacted in the online sphere. The research employs a case study of the Grassy Mountain Mine in southwest Alberta, exploring the voices that influenced the decision to terminate the project. The study delves into the complexities of navigating and constructing meaning between tangible and intangible realities in a time when our relationship with physical spaces is undergoing significant change. With political advocacy highly relevant given the social impacts of the pandemic, social movements adapted. This reshaped the ways in which both movements as a whole and the individuals they comprise of occupied and engaged with both physical and online spaces. This Chapter wraps up concepts and findings explored throughout this thesis through a brief overview of their content, then continues on to a broad discussion of notable thesis findings in relation to existing literature, including future directions and research limitations.

6.1 Thesis Overview

6.1.1 Background and Theoretical Basis

Though work on social movements has existed since the 1800's and these frameworks have shaped much of today's perspectives on collective action, widespread access to the internet in the mid-2000's marked a new era connectivity and social movement theory (cf. Olsen, 1965; McCarthy & Zald, 1977; Tilly, 1978; Habermas, 1981; Melucci, 1996). Notably, the Arab Spring of 2011 represents a shift towards social movements utilizing largescale digital media and online platforms, resulting in the emergence of a new era of collective organizing- coined "connective action" (Bennett & Segerberg, 2012; Kharroub, 2016; Tufekci, 2018). Connective action emphasizes personalized communication, individualized action frames, and the role of digital networks in mobilizing collective action. It highlights the decentralization of organizing agents and the formation of weak tie networks, with more recent works also raising questions about the political nature of online spaces and the impact of social media as a commercial entity on collective action (Milan, 2015; Milan & Velden, 2016).

With social media platforms acting as an organizational agent to most of today's social movements in some capacity, it is valuable to examine how the platform itself may influence activist action. Social media platforms, driven by profit and a lack of transparency to maintain their interests, have implications for politics, economies, and culture, forcing social actors to recalibrate their positions within society. These platforms facilitate rapid content turnover, personalized narratives, and weak tie networks which may be beneficial qualities for online collective action. However, this also contributes to the short-lived nature of online social movements, fosters echo chambers, and in some cases, has been shown to decrease political deliberation. It is clear that social media entities have biased interests shaping user experience, but online social movements are not just a product of their setting. Literature within the field suggests that there are generally positive relationships between online and offline activism on an individual level, although the context in which participation is situated plays a significant role.

Van Deth's (2021) model was the chosen approach to conceptualizing political participation, as it considers different levels of political participation, encompassing both action and intention, in order to comprehensively understand the various forms of online political participation while accounting for their differences. Research indicates that an individual's belief in their capacity to reach their goals as a collective (collective efficacy) along with a belief in their own capacity to influence the political system (internal efficacy), is linked to increased offline political participation. Facebook users demonstrate higher collective efficacy, and the platform is thought to have a greater impact on offline political participation compared to other platforms like Twitter (Halpern et al., 2017).

One concept that can guide interpretations of interactions across online and physical space is Soja's thirdspace. Thirdspace refers to the understanding that the value we assign to a space, whether it be online or physical, is one that is dynamic and continuously enacted by the experiences of individuals to whom it holds meaning (Gamson, 1990; Soja, 1998). It challenges existing binaries and allows for the emergence of the "Other". When applied to the online sphere, the flexibility and ambiguity of thirdspace find parallels in the dynamics of social media. This enables connectivity and meaning-making through interaction while transcending traditional (geographic and social) boundaries and giving space to reimagine them. This links back to digital connective action through Milan's (2015) notion of "cloud networking", which emphasizes the sociotechnical nature of digital collective action. Social media enables the renegotiation of boundaries by giving users a space where they can come together and form personalized narratives, identities, and networks (Bennett & Segerberg, 2012; Shafi & Ran, 2021). However, this space isn't fully public and no matter how radically transformed- the space is owned, facilitated, and ultimately, constructed, through the commercial interests of profit-driven corporations. The overall purpose of this thesis is situated deep within these conflicting dynamics.

6.1.2 Case Study

Crowsnest Pass is a specialized municipality and historical mining town, located in the Rocky Mountains of southwest Alberta. It is overlooked by Grassy Mountain, the proposed home of a metallurgical coal mine until August 2021. An online grassroots movement, under the name "Protect Alberta's Rockies and Headwaters" emerged to advocate for the denial of the mining project, highlighting concerns about environmental degradation and water pollution throughout Alberta. The group gained significant support, amassing over 37,000 followers over a few months. In response, a pro-mine group called "Crowsnest pass. Ghost town" emerged. As of June 2021, the project was denied by the Alberta Energy Regulator and was shut down by the Government of Canada's Impact Assessment Agency by the end of August 2021, resulting in mixed feelings across political interest groups.

6.1.2 Research Design

Throughout this thesis, it is evident that social media platforms as venues for political participation impact various elements of how politics are enacted online. The case study of the Grassy Mountain coal mining project was used to examine interactions within the online and physical world, especially when movement was restricted due to the Covid-19 pandemic. Specifically, this thesis set out to explore the following questions;

- 1. Within the case of the Grassy Mountain Mine project, how do various actors conceptualize and enact political activism across online and physical space?
- 2. How have individuals communicated their political interests regarding the Grassy Mountain Mine project through online channels since the start of Covid-19 pandemic and ensuing restrictions?

3. What role does the digital infrastructure in which this social movement was situated play in facilitating these communications and processes?

A mixed methods approach was taken to conducting this research, consisting of the social media analysis presented in Chapter 4 and the semi-structured interviews presented in Chapter 5. The social media analysis was conducted using anonymized data from four public Facebook groups related to the Grassy Mountain mining project, resulting in a total of 612 post "bundles". Due to limitations in data collection methods, posts were collected manually using keyword searches. Though this decision was a product of some unexpected data collection challenges, this approach allowed the researcher to gain familiarity with the content, which was ultimately of great benefit. To gain a deeper understanding of personal experience as an online activist, a total of four supplementary interviews were conducted with individuals who have contributed social media content in at least one of the four groups related to the Grassy Mountain Mine project. This resulted in a more nuanced understanding of the issues and allowed the clarification of points from the social media analysis.

6.1.3 Research Findings and Analysis

Through a social media analysis, Chapter 4 examines the dynamics of Facebook communities related to the Grassy Mountain mining project in Alberta. Building on the social media analysis, Chapter 5 presents findings on individual experiences advocating for political interests regarding Grassy Mountain. The analyses explore the how individuals participated in each group, their differing approaches and content, and the interplay between external events, political decisions, and group dynamics. A diversity of perspectives and narratives within the groups are revealed, with perceived blame assigned to different entities such as coal mining companies, anti-coal activists, political parties, and the federal government. The findings also highlight the presence of a larger

activist network of organizations within the PARH movement, demonstrating connective action and amplification of messages through different venues. The findings from this analysis also provide evidence that the structure of Facebook influences user behavior, with certain types of content (such as links) seeming to be easier to post but more difficult to engage with, potentially due to the platform's design and commercially motivated aim to retain users. Evidence supports selective participation and echo chamber phenomena, with groups dedicated to defending a specific stance on the issue of coal mining exhibiting more likes and shares than active discussion. Likely influenced by the context of the Covid-19 pandemic, concerns over various forms of censorship and conflicts over freedom of speech within the groups were observed.

Consistent with reports of social media use throughout Covid-19 (Grant, 2021; Brückmann 2022), participants interviewed in Chapter 5 report that their general social media use increased during the pandemic, often due to a desire to seek out and contribute reliable information. This is supported by the social media analysis, where "Informational" posts are a top category across the groups examined. Chapter 4 findings also show that the amount of activity across all four Facebook groups increases in response to external events, such as the Government of Alberta's policy decision to reinstate the 1976 Coal Policy governing surface mining, which is consistent with the above observation that individuals both seek out and contribute informational content through their participation in the groups. At this same time, trust in governing bodies during the Covid-19 pandemic was low, with participants believing that controversial decisions would be made while the attention of mainstream media and the general public was focused on the pandemic (Suhay et al., 2022). Regardless of political stances on mining in the Eastern Slopes, none of the interview participants feel that meaningful progress has been made and multiple participants describe their overall experience as further affirming their views and discontent with decision-makers. However, all

participants also made it clear that even though a concrete policy decision regarding the Grassy Mountain Mine had been made, they don't plan to stop advocating for their political interests.

6.2 Discussion and Future Directions

As was discussed in Chapter 3, perceived political efficacy is generally thought to have a positive impact on political action (Halpern et al., 2017). This is not reflected in the findings, as users describe both individual and collective perceived efficacy to be low but draw direct links between low efficacy and high political participation through their frustration with various facets of activismwhether this be the outcome of the project, governance systems, or both. Individuals do not feel as if they have achieved their goals or made impactful contributions through their political participation within the movements and in interviews, often describe disliking their time spent on Facebook. However, they also do not feel that their views have changed over the course of their participation and are adamant that they will not cease their contributions until the perceived issue is resolved. This speaks to perceived political efficacy, the interactions between online participation, and the greater political landscape. One possible reason for this juxtaposition is that most details regarding personal views on advocacy (i.e. individual perceptions) in this case study were gained through a interviews with a segment of individuals who were active enough in the group to see the recruitment post over a year after the conclusion of this event and also desired to share their experiences. This highlights a limitation of this research, as it is likely that the people who were interviewed do not represent the views of all people participating from December 2021 to August 2021. Given enough resources, social media data would ideally be collected in real-time using automated methods, as opposed to a social media analysis being done retroactively.

6.2.1 Emotion, Identity, and Space

Throughout the case study of activism regarding the Grassy Mountain mining project, it is evident that individual engagement with the issue in online settings is shaped by a strong emotional connection to the physical space. The social media analysis and interviews show that though organized online, physical advocacy was a key component of expressing political views regarding the mining project. Both traditional and novel physical protest tactics were used, such as chalking or rolling convoys used to spread awareness of the issue in-person and often later documented in the Facebook groups. This key connection between online and physical spaces reveals an iterative process where the meaning gained from interactions with/within the physical space is enacted online, which shapes and transforms this conceptualization in the digital realm through the input of other (often likeminded) users, in turn impacting how this space is navigated outside of the online and shaping both collective and personal identities.

This was clearly observed in participant interviews, where individuals described the value of physical space not only for them and their identity, but also for other Albertans interacting with space. A key motivator for activist action was stated to be the desire to preserve the existing meaning the space holds to them. Within the social media analysis (Figure 4.6b), it was found that content within CPGT is centered largely around personal experiences, which are communicated primarily through the form of home videos. This may also speak to the fact that CPGT was a group more local to Crowsnest Pass, where individuals seem to have shared networks in physical space to a greater extent than PARH members, where individuals were dispersed across Alberta. Differences in motivations behind physical and online activism across groups may also hold insights at the intersection of environmental activism and human psychology. Applying place attachment theory to the Grassy Mountain Mine case study is suggested as an avenue for future research on social media activism regarding physical space.

These findings complement existing work on emotions as a key component of collective action and online political participation, which indicate that emotional attachment in combination with group cohesion and solidarity is a foundational element of forming a united collective, which in turn impacts both how injustice is perceived through the process of collective meaning-building and the overall collective efficacy of the movement (Garcia, 2016; Brünker et al., 2019). The role of emotion within political participation is also consistent within a Covid-19 context, with observations of a positive relationship between strong emotion regarding events in physical space and online political participation during the pandemic (Faure & Beltran, 2022). Xu et al. (2021) also shows that within the context of Covid-19 restrictions, online interactions via social media regarding physical, natural spaces evoked strong emotion from participants, such as feeling "moved, uplifted, or inspired" (pg.9). In the works of Xu et al (2021), online expressions of emotion regarding physical space was shown to take attention and worry away from the pandemic. This raises interesting connections to this case study, as multiple interview participants stated that they joined Facebook specifically for this cause after the beginning of the pandemic and describe spending much of their time engaging with the issue on social media. Along with the finding that individuals who participate in these online settings generally do not enjoy their time on Facebook and describe significant personal drawbacks to their use of social media, this presents an interesting direction for future research on online political participation during the pandemic.

6.2.2. Impacts of Platform Structure on Collective Action

Given the ease of connectivity in today's digital era, organizing as a collective based on the constructed meanings discussed above is done largely online, transferring our conceptualization of this physical space into the digital world. However, as we transition into continuing the defense of this meaning through activist action within online space, the various attributes of a platform with its own hidden interests are also powerful at shaping how an issue is constructed and in turn how we

interact with the space around us, whether this be consciously or unconsciously (Dijck & Poell, 2015). This dichotomy situates social media platforms as a space where users have the freedom to express themselves, but by engaging in this self-expression, the corporate entity behind the platform gains ownership over the content and the power to regulate this expression. Though evidence of this exists in literature (Heyman & Pierson, 2015; Milan, 2015), rapid technological advancements in combination with a general lack of transparency on behalf of corporations makes it difficult to make conclusive statements on the techno-commercial impacts of social media platforms on online behaviours.

One major finding regarding Facebook as the platform facilitating online activism was a lack of transparency and predictability. Interview participants describe tools that keep changing and ultimately make it more difficult to manage and feel in control navigating the group. This also ties into ideas of censorship, where due to this unpredictability, group members held the belief that people were entering the group when they *shouldn't* be to intentionally sabotage the advocacy efforts or cohesiveness of the group (labelled as "trolls"). As a result, this evoked strong feelings and drastic actions among group members, often resulting a perceived need to censor content within the group to maintain cohesiveness. This lack of transparency impacts both perceptions of other activists within a network and emotions around online participation within the groups, requiring Facebook users to recalibrate their existence within these online collectives accordingly (Edirisinghe, 2011, Milan & Velden, 2016).

Both Alvarez et al. (2015) and Garcia et al. (2016) found that individuals participate in online discussions when their emotional arousal passes a certain threshold and that this state is achieved much faster when viewers are viewing polarizing content, as opposed to neutral content. This is consistent with interview findings that polarizing content evokes more emotion, which, according to

Garcia et al. (2016) in turn increases participation and online discussion. This is consistent with the "political" category emerging from the social media analysis across both the CPGT and PARH groups, where distinct sides were taken with one of the two main Alberta political parties (New Democratic Party versus United Conservative Party). It also provides insight into the observation that non-political parties (CPU and AACP) had many more comments than reactions to posts, signifying more conversation happening. Though much of the time the post itself was not polarized, users in the comments voiced strong opinions on the issue, leading to deliberation within the comment sections and emotion being evoked by reading other's comments.

This lack of transparency and predictability was also reflected in my own experience conducting the social media analysis, where I gained first-hand experience with some accessibility challenges of the platform. Though presented as a public platform, my attempts at data collection made it clear that there are great barriers to accessing this data. As was discussed in Chapter 3, it is nearly impossible to pull large amounts of public data from the platform without a third-party service (most of which have been banned by Facebook). The option presented by the platform is the use of their API, which required not only a high degree of coding knowledge and time, but the data pulled cannot leave the API interface, meaning that the user does not hold any intellectual ownership over the data. When collecting data manually, loading the main feeds displaying the actual content is designed in such a way that requires an amount of processing power that even a gaming or business analytics PC cannot handle. It would require extensive amounts of computing power which are far from accessible to most people. Finally, the "search" function is far from transparent and will selectively show posts with no clear explanation of the processes behind it (which is also largely why I had to adopt the keyword approach and collection of 5 posts per week). As such, these barriers are a limitation to this research, as only a sample of data could be pulled and there was little control over the posts shown by the platform.

The formation of echo chambers within Facebook advocacy is supported, complementing work such as that of Greijdanus et al. (2020) or Zhong (2022). This can be seen through the discussion on information above, where each participant feels that their group is the true source of information regarding the issue and dismisses other sources as misinformation. This can also be clearly seen through the social media analysis, where posts in groups that are specifically dedicated to the issue of coal mining in the Eastern Slopes (CPGT and PARH) have a high proportion of reactions and shares on posts, but few textual comments. The inverse is observed in pages that are not specific to the issue (AACP and CPU).

6.2.3 Concluding Remarks

This examination of political participation regarding the Grassy Mountain Mining project during the Covid-19 pandemic provides insights into the ways that political activism can transcend boundaries between online and physical space. Rather than the binary of the physical vs. the online world, this research supports the hybrid and iterative nature of social media activism, where physical activism reinforced online activism (both through the constructions of the physical space in a digital environment as discussed above and by individuals who participated in physical activism (most physical advocacy efforts were organized online and many of these networks were formed entirely online).

Beyond the Covid-19 pandemic, the findings of this case study remain applicable in numerous ways. The increasing tendency of social movements to adopt hybrid forms paired with a political spotlight on the diminishing future of non-renewable resource extraction raises interesting questions around environmental activism. The implications of people using Facebook as a news source while also interacting more extensively with content that evokes strong emotion raises questions regarding the type of content that gets the most visibility on social media. This is pertinent to conversations on the spread of misinformation via social media. If activists engaging in hybrid settings do most of their framing work within this quick turnover cycle and have to work against Facebook's algorithms to ensure longevity, it may be more advantageous to post content that evokes strong emotions than it is to post accurate content, making users more likely to interact with it and as a result, help the content maintain its visibility on the platform. This provides interesting avenues for future research in terms of both the psychology of online activism and the socio-technological impacts of the digital infrastructure on information consumption.

Appendices

Appendix A: Supplementary Social Media Data

Appendix D contains a variety of supplementary information regarding the social media analysis, including:

- Weekly summaries: the count and sum of number of posts in each group, split by week and keyword.
- Embedded content structures: the template used to ensure consistency in data input between different types of embedded content and a diagram showing the different types of embedded content that were encountered, by type and subtype.

Weekly summaries, by group

Around and About Crowsnest Pass (4 posts total)

	ALBERTA	COAL	GRASSY MOUNTAIN	MINING	ENVIRONMENT	SUM
Dec 16-Dec 19	0	0	0	0	0	0
Dec 20-Dec 26	1	0	1	1	1	4
Dec 27-Jan 2	0	0	0	0	0	0
Jan 3-Jan 9	0	0	0	0	0	0
Jan 10-Jan 16	0	0	0	0	0	0
Jan 17-Jan 23	0	0	0	0	0	0
Jan 24-Jan 30	0	0	0	0	0	0
Jan 31-Feb 6	0	0	0	0	0	0
Feb 7-Feb 13	0	0	0	0	0	0
Feb 14-Feb 20	0	0	0	0	0	0
Feb 21-Feb 27	0	0	0	0	0	0
Feb 28-Mar 6	0	0	0	0	0	0
Mar 7-Mar 13	0	0	0	0	0	0
Mar 14-Mar 20	0	0	0	0	0	0

Mar 21- Mar 27	0	0	0	0	0	0
War 21- War 27	0	0	0	0	0	U
Mar 28- Apr 3	0	0	0	0	0	0
Apr 4-Apr 10	0	0	0	0	0	0
Apr 11-Apr 17	0	0	0	0	0	0
Apr 18-Apr 24	0	0	0	0	0	0
Apr 25-May 1	0	0	0	0	0	0
May 2-May 8	0	0	0	0	0	0
May 9-May 15	0	0	0	0	0	0
May 16-May 22	0	0	0	0	0	0
May 23-May 29	0	0	0	0	0	0
May 30-Jun 5	0	0	0	0	0	0
Jun 6-Jun 12	0	0	0	0	0	0
Jun 13-Jun 19	0	0	0	0	0	0
Jun 20-Jun 26	0	0	0	0	0	0
Jun 27-Jul 3	0	0	0	0	0	0
Jul 4-Jul 10	0	0	0	0	0	0
Jul 11-Jul 17	0	0	0	0	0	0
Jul 18-Jul 24	0	0	0	0	0	0
Jul 25-Jul 31	0	0	0	0	0	0
Aug 1-Aug 7	0	0	0	0	0	0
Aug 8-Aug 14	0	0	0	0	0	0
SUM	1	0	1	1	1	4

Crowsnest Pass Ghost Town (40 posts total)

	ALBERTA	COAL	GRASSY MOUNTAIN	MINING	ENVIRONMENT	sum
Dec 16-Dec 19	0	0	0	0	0	0
Dec 20-Dec 26	0	0	0	0	0	0
Dec 27-Jan 2	0	0	0	0	0	0
Jan 3-Jan 9	0	0	0	0	0	0
Jan 10-Jan 16	0	0	0	0	0	0
Jan 17-Jan 23	0	0	0	0	0	0
Jan 24-Jan 30	0	0	0	0	0	0
Jan 31-Feb 6	0	0	0	0	0	0
Feb 7-Feb 13	0	0	0	0	0	0
Feb 14-Feb 20	0	0	0	0	0	0
Feb 21-Feb 27	0	0	0	0	0	0
Feb 28-Mar 6	0	0	0	0	0	0
Mar 7-Mar 13	0	0	0	0	0	0
Mar 14-Mar 20	0	0	0	0	0	0
Mar 21- Mar 27	0	0	0	0	0	0
Mar 28- Apr 3	0	0	0	0	0	0
Apr 4-Apr 10	0	0	0	0	0	0
Apr 11-Apr 17	0	0	0	0	0	0
Apr 18-Apr 24	0	0	0	0	0	0
Apr 25-May 1	0	0	0	0	0	0
May 2-May 8	0	0	0	0	0	0
May 9-May 15	0	0	0	0	0	0
May 16-May 22	0	0	0	0	0	0
May 23-May 29	0	0	0	0	0	0
May 30-Jun 5	0	0	0	0	0	0
---------------	---	----	---	----	---	----
Jun 6-Jun 12	0	0	0	0	0	0
Jun 13-Jun 19	2	3	3	2	0	10
Jun 20-Jun 26	3	3	3	3	0	12
Jun 27-Jul 3	1	1	0	0	0	2
Jul 4-Jul 10	0	0	0	0	0	0
Jul 11-Jul 17	0	2	1	2	0	5
Jul 18-Jul 24	3	3	2	3	0	11
sum	9	12	9	10	0	40

CPU (15 posts total)

	ALBERTA	COAL	GRASSY MOUNTAIN	MINING	ENVIRONMENT	sum
Dec 16-Dec 19	0	0	0	0	0	0
Dec 20-Dec 26	0	0	0	0	0	0
Dec 27-Jan 2	0	0	0	0	0	0
Jan 3-Jan 9	0	0	0	0	0	0
Jan 10-Jan 16	0	0	0	0	0	0
Jan 17-Jan 23	0	0	0	0	0	0
Jan 24-Jan 30	0	0	0	0	0	0
Jan 31-Feb 6	0	0	0	0	0	0
Feb 7-Feb 13	0	0	0	0	0	0
Feb 14-Feb 20	0	0	0	0	0	0
Feb 21-Feb 27	0	0	0	0	0	0
Feb 28-Mar 6	0	0	0	0	0	0

Mar 7-Mar 13	0	0	0	0	0	0
Mar 14-Mar 20	0	0	0	0	0	0
Mar 21- Mar 27	0	0	0	0	0	0
Mar 28- Apr 3	0	0	0	0	0	0
Apr 4-Apr 10	0	0	0	0	0	0
Apr 11-Apr 17	0	0	0	0	0	0
Apr 18-Apr 24	0	0	0	0	0	0
Apr 25-May 1	0	0	0	0	0	0
May 2-May 8	0	0	0	0	0	0
May 9-May 15	0	0	0	0	0	0
May 16-May 22	0	0	0	0	0	0
May 23-May 29	0	0	0	0	0	0
May 30-Jun 5	0	0	0	0	0	0
Jun 6-Jun 12	0	0	0	0	0	0
Jun 13-Jun 19	1	1	1	1	0	4
Jun 20-Jun 26	0	0	0	0	0	0
Jun 27-Jul 3	0	0	0	0	0	0
Jul 4-Jul 10	0	0	0	0	0	0
Jul 11-Jul 17	0	1	1	1	1	4
Jul 18-Jul 24	1	2	2	2	0	7
Jul 25-Jul 31	0	0	0	0	0	0
Aug 1-Aug 7	0	0	0	0	0	0
Aug 8-Aug 14	0	0	0	0	0	0
sum	2	4	4	4	1	15

PARH (549 posts total)

	ALBERTA	COAL	GRASSY MOUNTAIN	MINING	ENVIRONMENT	sum
Dec 16-Dec 19	2	3	0	4	1	10
Dec 20-Dec 26	5	5	5	5	5	25
Dec 27-Jan 2	5	5	5	5	5	25
Jan 3-Jan 9	0	1	1	2	0	4
Jan 10-Jan 16	3	1	5	4	0	13
Jan 17-Jan 23	5	5	5	5	3	23
Jan 24-Jan 30	2	5	5	5	1	18
Jan 31-Feb 6	5	5	3	5	3	21
Feb 7-Feb 13	5	5	5	5	3	23
Feb 14-Feb 20	5	5	5	5	3	23
Feb 21-Feb 27	1	2	5	5	0	13
Feb 28-Mar 6	1	3	4	5	2	15
Mar 7-Mar 13	2	2	2	3	1	10
Mar 14-Mar 20	5	3	3	3	1	15
Mar 21- Mar 27	2	5	2	5	1	15
Mar 28- Apr 3	2	3	5	5	0	15
Apr 4-Apr 10	3	3	2	5	3	16
Apr 11-Apr 17	5	4	0	5	2	16
Apr 18-Apr 24	5	4	4	1	5	19
Apr 25-May 1	5	0	2	3	5	15
May 2-May 8	5	1	4	1	3	14
May 9-May 15	5	3	5	1	5	19
May 16-May 22	5	4	0	1	4	14

May 23-May 29	1	3	1	2	4	11
May 30-Jun 5	5	5	2	3	5	20
Jun 6-Jun 12	4	4	1	3	4	16
Jun 13-Jun 19	5	5	5	0	5	20
Jun 20-Jun 26	5	5	5	1	5	21
Jun 27-Jul 3	4	3	0	1	4	12
Jul 4-Jul 10	3	3	1	1	0	8
Jul 11-Jul 17	2	5	2	0	2	11
Jul 18-Jul 24	2	5	5	2	3	17
Jul 25-Jul 31	3	3	0	1	5	12
Aug 1-Aug 7	3	4	5	0	5	17
Aug 8-Aug 14	1	1	0	1	0	3
sum	121	123	104	103	98	549

Embedded content structures

Embedded content types encountered were documented using the following diagram to ensure consistency in the naming and designation of each type of embedded content item:



The format that was followed to input any links or other embedded media was as follows:

Template: [content type]; [content title/brief description]; [content link; if applicable]

Example 1: link; Alberta government knew bighorn sheep contaminated with coal-mine selenium: scientist; https://www.cbc.ca/news/canada/edmonton/alberta-government-knew-bighorn-sheep-contaminated-with-coal-mine-selenium-scientist-1.6022966?fbclid=IwAR3Fa7AXUdfrUeO66Qy3rZN3nFdhVvcCaPX-FYH_vexSr8dHsYNhzFMdEIY

Example 2: map; Alberta watershed planning and advisory map

Appendix B: Application for Ethics Review Form (GC-REB)



September 14, 2022

Reference number: 20230638

Dear Katarina Nedeljakova,

Thank you for your application for ethical clearance for your proposal Impacts of social media activism on energy policy throughout Covid-19: a case study of the Grassy Mountain Mining project. The Grenfell Campus Research Ethics Board (GC-REB) has reviewed your application and finds this application in ethical compliance with the Tri-Council Guidelines.

Your approval for this project expires on September 14, 2023. To remain in compliance with Article 6.14 (Continuing Research Ethics Review) of the Tri-Council Policy Statement on Ethics in Human Research (TCPS2), should your project continue past that date, you are required to renew your ethics approval before that time. As well, please note that any changes to the proposed study will need to be cleared by the GC-REB first.

Note: One final suggestion. The researcher may want to modify their date of participant withdrawal to an earlier date than their thesis submission date. We would suggest an anticipated date when a first draft would be completed since that draft would be based on the participant data. Again, this is merely a collegial opinion and giving without obligation or expectation.

The Board wishes you success with your research.

Best wishes,

John Bodner, Ph.D., Chair

IMPORTANT REMINDERS - PLEASE READ:

Important Notice regarding COVID-19: As the situation changes and develops with COVD-19, it is up to the PI to ensure that the research team remains in compliance with Memorial's current status on in-person data collection. You can follow information on the current status of policy here: https://www.mun.ca/research/.

GRENFELL CAMPUS, MEMORIAL UNIVERSITY 20 University Drive, Corner Brook, NL, Canada, A2H 5G4 Tel: 709 637 6200 Fax: 709 639 8125 www.grenfell.mun.ca

FIND YOUR CORNER

Appendix C: 1976 Coal Development Policy Map

As referred to in the "Case Study" section of Chapter 3, the rescindment/reinstatement of the 1976 Coal Development Policy was a large source of public discontent. This shows the original protected areas and brief description of each protected area to aid in interpreting this map. Obtained from: <u>https://open.alberta.ca/dataset/cc40f8f5-a3f7-42ce-ad53-7521ef360b99</u>; https://www.alberta.ca/coal-policy-guidelines.aspx

Category breakdown:

Category 1: National Parks, present or proposed Provincial Parks, Wilderness Areas, Designated Recreation Areas, Wildlife Sanctuaries, etc.

Category 2: Parts of the southern Rocky Mountains and Foothills

Category 3: Northern Forested Region and eastern portions of the Eastern Slopes

Category 4: Areas of the province not covered in other 3 categories

Spring 2021 updates:

"With the rescission of the Coal Policy, all restrictions on issuing coal leases within the former coal categories 2 and 3 have been removed. Alberta will continue to restrict coal leasing, exploration and development within public lands formerly designated as coal category 1. This prohibition on coal activities is being continued to maintain watershed,





Appendix D: Social Media Terminology

Comments vs. comment responses: comments refer to direct function; comment responses refer to the use of the "Reply to comment" function on an existing comment. This allows for separate sub-conversations within each comment posted.

Reactions: an upgrade from the original "like" function of Facebook, users now have a selection of various 'reactions' to a post. This allows a wider range of emotion, including negative emotion to be shown. Reactions are as follows:



- Like, Love, HaHa are considered positive reactions
- Wow is considered neutral/ not positive or negative
- Sad and Angry are considered negative reactions

Shares: users have the ability to share content they see on their feeds, meaning that their network (depending on their selected privacy preferences) can see, comment, react, and further share the post.

Tagging: using the tag function, users can add the name of another user (usually someone they 'friended') on a comment or post so the other user will get a notification and be able to see the post. This does not override any content the tagged individual does not have access to.

Facebook group structures

"Facebook groups give Page owners a platform and tools to build an engaged, relevant community among existing and potential customers. Groups are hubs where valuable conversations flourish between brands and customers, and among customers themselves. ... groups offer 2 privacy settings:

-Public groups: Public groups are searchable and all content is publicly visible. If you choose this privacy setting, you should expect to monitor your join requests.

-Private groups: Private groups are searchable and anyone can request to join them, but only members can see their content." -Meta Business Help Center

Group admin: the administrator of a group. They, theoretically, have full control over their groups and can accept join requests, delete posts and comments, ban people, remove people from group, etc.

Vetting questions: a function where the administrators of any group (public or private) can set a series of questions for anyone wanting to join the group. This is designed to provide administrators more control over group membership and ensure there are no bots joining. Individuals who did not answer the questions on public groups can still see, search, and react to content, but they cannot post or comment in that group.

Group search mechanisms bar and filtering system:

The following image is an example of Facebook's search system within public groups and the various filters which can be set. Below, "example" is the word being searched and the results are being filtered by only posts made in the year of 2020.

Search Results	
in Protect Alberta's Rockies and Headwater	s

Q	example
---	---------

Filters	Reset
Posts You've Seen	
Most Recent	
Posted by	-
Tagged Location	-
2020	8

Works Cited

- Addo, I.Y. (2020). Double pandemic: racial discrimination amid coronavirus disease 2019, Social Sciences & Humanities Open. 2(1): 100074, https://doi.org/10.1016/j.ssaho.2020.100074.Cipriano.
- Adkin, L. (2016) Ecology and Governance in a First World Petro-State. In L. Adkin (Ed.), First World Petro-Politics: The Political Ecology and Governance of Alberta (pp.3-50), University of Toronto Press.
- Adkin, L.E., Hanson, L.L., Kahane, D., Parkins, J.R., Patten, S. (2017). Can public engagement democratize environmental policymaking in a resource dependent state? Comparative case studies from Alberta, Canada. Environmental Politics, 26:2,301-321, DOI: 10.1080/09644016.2016.1244967"
- Agostini, M., & van Zomeren, M. (2021). Toward a comprehensive and potentially cross-cultural model of why people engage in collective action: A quantitative research synthesis of four motivations and structural constraints. Psychological Bulletin, 147(7), 667–700. https://doi.org/10.1037/bul0000256
- Allcott, H., Braghieri, L., Eichmeyer, S., & Gentzkow, M. (2020). The Welfare Effects of Social Media. American Economic Review, 110(3), 629–676. https://doi-org.qe2a proxy.mun.ca/10.1257/aer.20190658
- Alvarez, R., Garcia, D., Moreno, Y., & Schweitzer, F. (2015). Sentiment cascades in the 15M movement. EPJ Data Science, 4(1), 1-13. https://doi.org/10.1140/epjds/s13688-015-0042-4
- Andreotta, M., Nugroho, R., Hurlstone, M.J. et al. (2019). Analyzing social media data: A mixedmethods framework combining computational and qualitative text analysis. Behav Res 51, 1766–1781. https://doi-org.qe2a-proxy.mun.ca/10.3758/s13428-019-01202-8
- Armstrong, E. A., & Bernstein, M. (2008). Culture, Power, and Institutions: A Multi-Institutional Politics Approach to Social Movements. Sociological Theory, 26(1), 74–99. https://doi.org/10.1111/j.1467-9558.2008.00319.x
- Around and About Crowsnest Pass [Facebook group]. Retrieved October 13, 2022, from https://www.facebook.com/groups/Aroundandabout
- Arya, D., Henn, M. (2021) COVID-ized ethnography: Challenges and opportunities for young environmental activists and researchers. Societies,11(2), 58. doi: http://dx.doi.org.qe2aproxy.mun.ca/10.3390/soc11020058
- Assaly, R. (2021) How we got here: a timeline of Alberta's response to the Covid-19 pandemic.. The Star. September 16. https://www.thestar.com/news/canada/2021/09/16/how-we-got-here-a-timeline-of-albertas-response-to-the-covid-19-pandemic.html
- Balhara, Y., Kattula, D., Singh, S., Chukkali, S., & Bhargava, R. (2020). Impact of lockdown following Covid-19 on the gaming behavior of college students. Indian journal of public health, 64(Supplement), S172–S176. https://doi.org/10.4103/ijph.IJPH_465_20

- Bandura, A. (1997). Self-efficacy: The exercise of control. W H Freeman/Times Books/ Henry Holt & Co.
- Batrinca, B., Treleaven, P.C. (2016). Social media analytics: a survey of techniques, tools and platforms. AI & Soc 30, 89–116. https://doi-org.qe2a-proxy.mun.ca/10.1007/s00146-014-0549-4
- Bavel, J. J. V., Baicker, K., Boggio, P. S., et al. (2020). Using social and behavioural science to support COVID-19 pandemic response. Nature Human Behaviour, 4, 460-471. https://doi.org/10.1038/s41562-020-0884-z
- Beech, M. (2020). Covid-19 Pusher Up Internet Use 70% And Streaming More Than 12%, First Figures Reveal. Forbes. https://www.forbes.com/sites/markbeech/2020/03/25/covid-19pushes-up-internet-use-70-streaming-more-than-12-first-figures-reveal/?sh=2938583f3104
- Bennett, D. (2022, December 8). Alberta passes Sovereignty Act, but first strips out sweeping powers for cabinet. CBC News. Retrieved from https://www.cbc.ca/news/canada/edmonton/alberta-sovereignty-act-1.6678407
- Bennett, W.L., Segerberg, A. (2012). T, Information, Communication & Society, 15:5, 739-768, DOI: 10.1080/1369118X.2012.670661
- Bhabha, Homi K. (1994). The Location of Culture. New York and London: Routledge.
- Bhui, K. (2021) Ethnic inequalities in health: The interplay of racism and Covid-19 in syndemics, EClinicalMedicine,Volume 36 (100953). https://doi.org/10.1016/j.eclinm.2021.100953.
- Blue, G., Rajewicz, L., Daub, S., Yunker, Z. 2018. In the Corporate Interest: Fossil Fuel Industry Input into Alberta and British Columbia's Climate Leadership Plans. Canadian Journal of Communication. 43:1, 93-110
- Blumer, H. (1969). Symbolic interactionism; perspective and method. Englewood Cliffs, N.J., Prentice-Hall
- Boulianne, S. (2019) Revolution in the making? Social media effects across the globe, Information, Communication & Society, 22:1, 39-54, DOI: 10.1080/1369118X.2017.1353641
- Bowler, D.E., Buyung-Ali, L.M., Knight, T.M., et al. (2010). A systematic review of evidence for the added benefits to health of exposure to natural environments. BMC Public Health, 10, 456. https://doi.org/10.1186/1471-2458-10-456
- Brescia, R. (2020). The Future of Change : How Technology Shapes Social Revolutions. Cornell University Press. https://www.jstor.org/stable/10.7591/j.ctvq2vz42.10
- Brown, G., & Raymond, C. (2007). The relationship between place attachment and landscape values: Toward mapping place attachment. Applied Geography, 27, 89-111. https://doi.org/10.1016/j.apgeog.2006.11.002

- Brückmann, G. (2022). The effects of policies providing information and trialling on the knowledge about and the intention to adopt new energy technologies. Energy Policy, 167, 113047. https://doi.org/10.1016/j.enpol.2022.113047
- Brunker, F., Deiteloff, F., Mirbabaie, M (2019) Collective Identity Formation on Instagram Investigating the Social Movement Fridays for Future, Australasian Conference on Information Systems
- Burkell, J., Fortier, A., Yeung, L., Wong, C., & Simpson, J. L. (2014). Facebook: Public space, or private space? Information, Communication & Society, 17(8), 974-985. DOI: 10.1080/1369118X.2013.870591
- Canadian Association of Petroleum Producers (2022). Advocacy Toolkit. https://www.ccpa-accp.ca/advocacy-toolkit/
- Canadian Energy Center (2022). Research. https://www.canadianenergycentre.ca/category/research/
- Caniglia, B., & Carmin, J. (2005). Scholarship on Social Movement Organizations: Classic Views and Emerging Trends. Mobilization: An International Quarterly, 10(2), 201-212. https://doi.org/10.17813/maiq.10.2.q848h37832857t5j
- Caniglia, Beth & Carmin, JA. (2005). Scholarship on Social Movement Organziations: Classic Views and Emerging Trends. Mobilization. 10. 201-212. 10.17813/maiq.10.2.q848h37832857t5j.
- Cantijoch, M., Galandini, S., & Gibson, R. (2016). 'It's not about me, it's about my community': A mixed-method study of civic websites and community efficacy. New Media & Society, 18(9), 1896–1915. https://doi.org/10.1177/1461444815616225
- Carothers, T., O'Donohue, A. (2020). Polarization and the Pandemic. Carnegie Endowment for International Peace. https://carnegieendowment.org/2020/04/28/polarization-andpandemic-pub-81638
- Carroll, C. (2014) Native enclosures: Tribal national parks and the progressive politics of environmental stewardship in Indian Country. Geoforum 53: 31–40.
- Casero-Ripollés, A., Pepe-Oliva, R. (2022). Social Media and Online Political Activism in Protest Communication Strategies: Friends or Foes?. In: Rocha, Á., Barredo, D., López-López, P.C., Puentes-Rivera, I. (eds) Communication and Smart Technologies. ICOMTA 2021. Smart Innovation, Systems and Technologies, vol 259. Springer, Singapore. https://doi-org.qe2aproxy.mun.ca/10.1007/978-981-16-5792-4_
- Castells, M. (2012). Networks of Outrage and Hope: Social Movements in the Internet Age. Cambridge: Polity Press.
- Chamberlain, M.J. (2020). Information Letter. https://inform.energy.gov.ab.ca/Documents/Published/IL-2020-23.pdf

Chwalisz, C. (2020) The pandemic has pushed citizen panels online. Nature. 589.171-172.

- Coal. (2022) Alberta Culture and Tourism. http://history.alberta.ca/energyheritage/coal/the-earlydevelopment-of-the-coal-industry-1874-1914/default.aspx#page-3
- Coleman, C. and Rich, J. (1973), Line, staff and the systems perspective. Hum. Resour. Manage., 12: 20-27. https://doi-org.qe2a-proxy.mun.ca/10.1002/hrm.3930120304
- Corry, O., & Reiner, D. (2021). Protests and Policies: How Radical Social Movement Activists Engage with Climate Policy Dilemmas. Sociology, 55(1), 197–217. https://doi.org/10.1177/0038038520943107
- Crowsnest Pass Uncensored [Facebook group]. Retrieved October 13, 2022, from https://www.facebook.com/groups/593331698312767
- Crowsnest pass. Ghost town [Facebook group]. Retrieved October 13, 2022, from https://www.facebook.com/groups/2964715677148359,
- Davidsen, C. (2016). The Alberta Oil/Tar Sands and 'Mainstream' Media Framings between Globalization and Polarization. In L. Adkin (Ed.), First World Petro-Politics: The Political Ecology and Governance of Alberta (pp.241-262), University of Toronto Press.
- DiCicco-Bloom, B., Crabtree, B.F. (2006) The qualitative research interview. Medical Education. Blackwell Publishing LTD. 40: 314-321. doi: 10.1111/j.1365-2929.2006.02418.x
- Dienlin, T., & Johannes, N. (2020). The impact of digital technology use on adolescent wellbeing [F]. Dialogues in clinical neuroscience, 22(2), 135–142. https://doi.org/10.31887/DCNS.2020.22.2/tdienlin
- Dreiling, Michael & Wolf, Brian. (2001). Environmental Movement Organizations and Political Strategy: Tactical Conflicts Over NAFTA. Organization & Environment - ORGAN ENVIRON. 14. 34-54. 10.1177/1086026601141002.
- Drucker, S. J., & Gumpert, G. (1996). The regulation of public social life: Communication law revisited. Communication Quarterly, 44(3), 280. doi:https://doi-org.qe2a-proxy.mun.ca/10.1080/01463379609370018
- Eamen, L., Brouwer, R., & Razavi, S. (2020). The economic impacts of water supply restrictions due to climate and policy change: A transboundary river basin supply-side input-output analysis. Ecological Economics, 172, 106532. https://doi.org/10.1016/j.ecolecon.2019.106532
- Edirisinghe, C., Nakatsu, R., Cheok, A., Widodo, J. (2011). Exploring the Concept of Third Space within Networked Social Media. In: Anacleto, J.C., Fels, S., Graham, N., Kapralos, B., Saif El-Nasr, M., Stanley, K. (eds) Entertainment Computing – ICEC 2011. ICEC 2011. Lecture Notes in Computer Science, vol 6972. Springer, Berlin, Heidelberg. https://doi-org.qe2aproxy.mun.ca/10.1007/978-3-642-24500-8_51
- Edwards, J. (2004). After the Fall. Discourse & Society, 15(2–3), 155–184. https://doi.org/10.1177/0957926504041015

- Elder, A. (2020). The interpersonal is political: Unfriending to promote civic discourse on social media. Ethics and Information Technology, 22(1), 15-24. doi:http://dx.doi.org.qe2a-proxy.mun.ca/10.1007/s10676-019-09511-4
- Elk Valley Coal News. (2019, September 4). The Crowsnest Grassy Mountain Coal Project. Retrieved from https://elkvalleycoal.com/the-crowsnest-grassy-mountain-coal-project/
- Esaiasson, Peter & Kolln, A.-K & Turper, Sedef. (2015). External Efficacy and Perceived Responsiveness--Similar but Distinct Concepts. International Journal of Public Opinion Research. 27. 10.1093/ijpor/edv003.
- Faraj, S., Azad, B. (2012). Materiality and Organizing: Social Interaction in a Technological World (Online ed.). Oxford: Oxford Academic. https://doi.org/10.1093/acprof:oso/9780199664054.001.0001
- Faure, A., Jara, R., Beltran, J. (2021). The Emotions of the Outbreak. Topics, Sentiments and Politics on Twitter During Chilean October. 10.1007/978-981-16-5792-4_22. Castells
- Finkel, S. E. (1985). Reciprocal Effects of Participation and Political Efficacy: A Panel Analysis. American Journal of Political Science, 29(4), 891–913. https://doi.org/10.2307/2111186
- Forrest, M. (2020). Environmental activists take pipeline protests online during Covid-19. Politico. Accessed 07/21/22. https://www.politico.com/news/2020/05/29/environmental-activiststake-pipeline-protests-online-during-covid-19-289245
- Franz, D., Marsh, H. E., Chen, J. I., & Teo, A. R. (2019). Using Facebook for Qualitative Research: A Brief Primer. Journal of medical Internet research, 21(8), e13544. https://doi.org/10.2196/13544
- Gahan, P., Pekarek, A. (2013). Social Movement Theory. British Journal of Industrial Relations, 51: 754-776. Doi: https://doi-org.qe2a-proxy.mun.ca/10.1111/j.1467-8543.2012.00912.x
- Gamson, W. A. (1989) Reflections on "The Strategy of Social Protest.". Review of The Strategy of Social Protest. Sociological Forum, 4(3), 455–467. http://www.jstor.org/stable/684616
- Gamson, W.A. (1990). The strategy of social protest. Wadsworth Pub Co.
- Garcia, D., Kappas, A., Küster, D., & Schweitzer, F. (2016). The dynamics of emotions in online interaction. Royal Society Open Science, 3(8). https://doi.org/10.1098/rsos.160059
- Gil de Zúñiga, H., Jung, N., & Valenzuela, S. (2012). Social Media Use for News and Individuals' Social Capital, Civic Engagement, and Political Participation. Journal of Computer-Mediated Communication, 17(3), 319-336. https://doi.org/10.1111/j.1083-6101.2012.01574.x
- Gillespie, T. (2010). The politics of 'platforms.' New Media & Society, 12(3), 347–364. https://doi.org/10.1177/1461444809342738
- Ginneken, J.V. (2003). Collective Behavior and Public Opinion : Rapid Shifts in Opinion and Communication. Routledge.

- Goodday, V. (2021). ENVIRONMENTAL REGULATION AND THE Covid-19 PANDEMIC: A REVIEW OF REGULATOR RESPONSE IN CANADA. The School of Public Policy Publications. 14:10. doi: https://doi.org/10.11575/sppp.v14i.71913"
- Government of Alberta. (1976). A Coal Development Policy for Alberta. https://open.alberta.ca/dataset/cc40f8f5-a3f7-42ce-ad53-7521ef360b99/resource/802d6feb-04ae-4bcc-aac3-3b3be31a0476/download/1114651976coal-development-policy-foralberta1976-06.pdf
- Government of Alberta. (2007). Aquatic and Riparian Condition Assessment of the South Saskatchewan River Basin. https://open.alberta.ca/dataset/274d8f92-9dde-4fb9-85c7c783fadb5969/resource/93661e85-3cd3-4d0f-a6f9b9c8c20f6ee0/download/aquaticriparianassessment-ssrb-jun2007.pdf
- Government of Alberta. (2020a). Annual Report: Energy 2019-2020. https://open.alberta.ca/dataset/cbd7147b-d304-4e3e-af28-78970c71232c/resource/fd81b900-5a31-4124-937f-0c7bcb23cebb/download/energy-annualreport-2021-2022.pdf
- Government of Alberta. (2020b). Year End Report: 2019-2020 Final Results. https://open.alberta.ca/dataset/cbd7147b-d304-4e3e-af28-78970c71232c/resource/fd81b900-5a31-4124-937f-0c7bcb23cebb/download/energy-annualreport-2021-2022.pdf
- Government of Alberta. (2021). Gross Domestic Product, Economic Dashboard. https://economicdashboard.alberta.ca/grossdomesticproduct#type
- Government of Alberta. (2022a). Coal Policy. https://www.alberta.ca/coal-policyguidelines.aspx#jumplinks-2
- Government of Alberta. (2022b). Emissions Reduction Programs. https://www.alberta.ca/emissions-management-reduction-programs.aspx
- Government of Canada (2021). Grassy Mountain Coal Project. https://iaacaeic.gc.ca/050/evaluations/proj/80101?culture=en-CA
- Government of Canada (2021b). Social inequalities in Covid-19 deaths in Canada. https://healthinfobase.canada.ca/covid-19/inequalities-deaths/index.html
- Government of Canada. (2020a). Background: Canada's climate finance commitments. https://www.international.gc.ca/consultation/climate-finance-financementclimatique/background-contexte.aspx?lang=eng
- Government of Canada. (2020b). Grassy Mountain Coal Project Environmental Assessment. https://iaac-aeic.gc.ca/050/evaluations/proj/80101
- Government of Canada. (2020c). Letter to hearing participants and detailed procedures. https://iaac-aeic.gc.ca/050/documents/p80101/135752E.pdf

- Government of Canada. (2021a). Statement by the Government of Canada on thermal coal mining. https://www.canada.ca/en/environment-climate-change/services/managingpollution/energy-production/electricity-generation/statement-government-canada-thermalcoal-mining.html
- Government of Canada. (2021c). Decision Statement. Impact Assessment Agency https://iaac-aeic.gc.ca/050/evaluations/document/140985
- Government of Canada. (2022a). Canada's 2030 Emissions Reduction Plan. Environment and Climate Change Canada https://publications.gc.ca/site/eng/9.909338/publication.html
- Government of Canada. (2022b). Proposed Coal Mining Effluent Regulations: overview. Environment and Climate Change Canada. https://www.canada.ca/en/environmentclimate-change/services/managing-pollution/sources-industry/proposed-coal-miningeffluent-regulations.html#toc0
- Gracia, D.V., Martínez, V.C. (2022). 'Quarantine' on Google and YouTube During the Covid-19 Pandemic: United States, United Kingdom, Spain and Italy. In: Rocha, Á., Barredo, D., López-López, P.C., Puentes-Rivera, I. (eds) Communication and Smart Technologies. ICOMTA 2021. Smart Innovation, Systems and Technologies, vol 259. Springer, Singapore. https://doiorg.qe2a-proxy.mun.ca/10.1007/978-981-16-5792-4_51
- Granovetter, M. S. (1973). The Strength of Weak Ties. American Journal of Sociology, 78(6), 1360–1380. http://www.jstor.org/stable/2776392
- Grant, P. R., & Smith, H. J. (2021). Activism in the time of Covid-19. Group Processes & Intergroup Relations, 24(2), 297–305. https://doi.org/10.1177/1368430220985208
- Greijdanus, C.A. de Matos Fernandes, F. Turner-Zwinkels, A. Honari, C.A. Roos, H. Rosenbusch, T. Postmes. (2020). The psychology of online activism and social movements: Relations between online and offline collective action. Current Opinion in Psychology, 35 (2020), pp. 49-54"
- Griffin, A. (2020) Why people are posting black squares to their Instagram. The Independent. June 2 Retrieved 07-04-22. https://www.independent.co.uk/tech/blackout-tuesday-instagram-blacksquares-how-to-post-box-a9543896.html

Habermas, J. (1981). 'New Social Movements'. Telos, 49: 33-37, doi: 10.3817/0981049033

- Halpern, D., Valenzuela, S. and Katz, J.E. (2017), We Face, I Tweet: How Different Social Media Influence Political Participation through Collective and Internal Efficacy. J Comput-Mediat Comm, 22: 320-336. https://doi-org.qe2a-proxy.mun.ca/10.1111/jcc4.12198
- Hancock, B., Bordes, A., Mazare, P.-E., & Weston, J. (2019). Learning from Dialogue after Deployment: Feed Yourself, Chatbot!. In Proceedings of the 57th Annual Meeting of the Association for Computational Linguistics (pp. 3667–3684). Florence, Italy: Association for Computational Linguistics.

- Helmond, A. (2015). The platformization of the web: Making web data platform ready. Social Media + Society. DOI: 10.1177/2056305115603080"
- Heyman, R., & Pierson, J. (2015). Social Media, Delinguistification and Colonization of Lifeworld: Changing Faces of Facebook. Social Media + Society. https://doi.org/10.1177/2056305115621933
- Hsieh, H. & Shannon, S. E. (2005). Three Approaches to Qualitative Content Analysis. Qualitative Health Research, 15(9), 1277-1288. DOI: 10.1177/1049732305276687
- Igartua, J.-J., Ortega-Mohedano, F., & Arcila-Calderón, C. (2020). Communication use in the times of the coronavirus: A cross-cultural study. El profesional de la información, 29(3), e290318. https://doi.org/10.3145/epi.2020.may.18
- Judge, M., Fernando, J. W., & Begeny, C. T. (2022). Dietary behaviour as a form of collective action: A social identity model of vegan activism. Appetite, 168, 105730. https://doi.org/10.1016/j.appet.2021.105730
- Kavada, A. (2015) Creating the collective: social media, the Occupy Movement and its constitution as a collective actor, Information, Communication & Society, 18:8, 872-886, DOI: 10.1080/1369118X.2015.1043318
- Kelly, A. B. (2011) Conservation practice as primitive accumulation. Journal of Peasant Studies 38: 683–701. https://doi.org/10.1080/03066150.2011.607695
- Kent, E. E., Prestin, A., Gaysynsky, A., Galica, K., Rinker, R., Graff, K., & Chou, W. Y. (2016).
 "Obesity is the New Major Cause of Cancer": Connections Between Obesity and Cancer on Facebook and Twitter. Journal of cancer education : the official journal of the American Association for Cancer Education, 31(3), 453–459. https://doi.org/10.1007/s13187-015-0824-1
- Kharroub, T., & Bas, O. (2016). Social media and protests: An examination of Twitter images of the 2011 Egyptian revolution. New Media & Society, 18(9), 1973–1992. https://doi.org/10.1177/1461444815571914
- Kim, H. K., Ahn, J., Atkinson, L., & Kahlor, L. A. (2020). Effects of COVID-19 Misinformation on Information Seeking, Avoidance, and Processing: A Multicountry Comparative Study. Science Communication. https://doi.org/10.1177/1075547020959670
- Klandermans, P. G., & Goslinga, S. (1996). Media discourse, movement publicity and the generation of collective action frames: theoretical and empirical exercises in meaning construction. In D. McAdam, & J. McCarthy (Eds.), Oppotunity structures and meaning construction Cambridge University Press.
- Kolers, A. (2016) Social movements. Philosophy Compass, 11: 580-590. doi: 10.1111/phc3.12351.

- Kountouris, Y., & Williams, E. (2023). Do protests influence environmental attitudes? evidence from extinction rebellion. Environmental Research Communications, 5(1), 011003. doi:https://doi.org/10.1088/2515-7620/ac9aeb
- Kuai, H., Yu, P., Liu, W., Zhang, Y., & Fan, J. (2023). Climate network analysis detects hot spots under anthropogenic climate change. Atmosphere, 14(4), 692. doi:https://doi.org/10.3390/atmos14040692
- Lammie, G. (2022). Restrictions to Crown coal leasing. https://inform.energy.gov.ab.ca/Documents/Published/IL-2022-09.pdf
- Latour, B. (2005). Reassembling the social: An introduction to Actor-network-theory. Oxford, UK: Oxford University Press.
- Le Bon, G. (1895, 1960). The Crowd : A Study of the Popular Mind. The Floating Press.
- Lee, M. (2021). Policing the Pedal Rebels: A Case Study of Environmental Activism Under Covid-19. International Journal for Crime, Justice and Social Democracy, 10(2), 156-168. https://doi.org/10.5204/ijcjsd.1887
- Lefebvre, H. (1974) Writings on Cities/Henri Lefebvre. Selected, translated and introduced by Eleonore Kofman and Elizabeth Lebas. Oxford, UK and Cambridge, MA: Blackwell.
- Little, W., McGivern, R. (2016) Social Movements and Social Change. Introduction to Sociology -1st Canadian Edition.
- Liu, D., Baumeister, R.F., Yang, C., Hu, B. (2019) Digital Communication Media Use and Psychological Well-Being: A Meta-Analysis, Journal of Computer-Mediated Communication, Volume 24, Issue 5, September, Pages 259–273, https://doi.org/10.1093/jcmc/zmz013
- Loader, B., Vromen, A., Xenos, M. (2014). The networked young citizen: social media, political participation and civic engagement INTRODUCTION. Information. 17. 10.1080/1369118X.2013.871571.
- Lovari, A., Bowen, S. A. (2020). Social media in disaster communication: A case study of strategies, barriers, and ethical implications. Journal of Public Affairs, 20(1), e1967.
- Lu, Y., Zhao, J., Wu, X., Lo, S. (2020). Escaping to nature in pandemic: A natural experiment of Covid-19 in Asian cities. Science of the Total Environment. Advance online publication. https://doi.org/10.31235/osf.io/rq8sn
- Macmillan, T. (2020, June 10). Learning from lockdown survey. FFCC. Retrieved from https://ffcc.co.uk/publications/learning-from-lockdown
- Marshall, S. J., White, E. C., Demuth, M. N., Bolch, T., Wheate, R., Menounos, B., Beedle, M. J., & Shea, J. M. (2011). Glacier Water Resources on the Eastern Slopes of the Canadian Rocky Mountains. Canadian Water Resources Journal / Revue canadienne des ressources hydriques, 36(2), 109-134. https://doi.org/10.4296/cwrj3602823

- Matarrita-Cascante, D., Stedman, R., & Luloff, A. E. (2010). Permanent and Seasonal Residents' Community Attachment in Natural Amenity-Rich Areas: Exploring the Contribution of Landscape-Related Factors. Environment and Behavior, 42(2), 197–220. https://doi.org/10.1177/0013916509332383
- Mathieu O'Neil & Robert Ackland. (2020). Risk issue adoption in an online social movement field. Information, Communication & Society, 23(13), 1854-1873. https://doi.org/10.1080/1369118X.2019.1620823
- McAdam, D. (1999). Political Process and the Development of Black Insurgency, 1930-1970. University of Chicago Press.
- McAdam, D., Tarrow, S. G., Tilly, C. (2001). Dynamics of Contention. Cambridge University Press.
- McCarthy, J. D., Zald, M. N. (1977). Resource Mobilization and Social Movements: A Partial Theory. American Journal of Sociology, 82(6), 1212–1241. http://www.jstor.org/stable/2777934
- Melucci, A. (1996). Challenging Codes: Collective Action in the Information Age (Cambridge Cultural Social Studies). Cambridge: Cambridge University Press. doi:10.1017/CBO9780511520891
- Milan, S. (2015). When Algorithms Shape Collective Action: Social Media and the Dynamics of Cloud Protesting. Social Media + Society. https://doi.org/10.1177/2056305115622481
- Milan, S., Velden, L. (2016). The Alternative Epistemologies of Data Activism. Digital Culture & Society. 2. 10.14361/dcs-2016-0205.
- Millward, P., Takhar, S. (2019). Social Movements, Collective Action and Activism. Sociology, 53(3), NP1–NP12. https://doi.org/10.1177/0038038518817287
- Mosquera, R., Odunowo, M., McNamara, T. (2020). The economic effects of Facebook. Exp Econ 23, 575–602 . https://doi-org.qe2a-proxy.mun.ca/10.1007/s10683-019-09625-y
- Municipality of Crowsnest Pass. (2019). Heritage Management Plan. https://www.crowsnestpass.com/public/download/files/87280
- Municipality of Crowsnest Pass. (2020). Community Survey Summary. Municipal Development for Crowsnest Pass. https://iaac-aeic.gc.ca/050/documents/p80101/136573E.pdf
- North Saskatchewan Watershed Alliance. Answering your questions about Alberta's Coal Policy. (2021). http://www.nswa.ab.ca/wp-content/uploads/2021/04/Coal-Policy-Info-Sheet-April-2021.pdf
- Oláh, J., Aburumman, N., Popp, J., Khan, M. A., Haddad, H., & Kitukutha, N. (2020). Impact of industry 4.0 on environmental sustainability. Sustainability, 12(11), 4674. doi:https://doi.org/10.3390/su12114674

- Olson, M. (1971). The Logic of Collective Action: Public Goods and the Theory of Groups, Second Printing with a New Preface and Appendix. Harvard University Press. https://doi.org/10.2307/j.ctvjsf3ts
- O'Shaughnessy, S., Doğu, G. (2016). The Gendered and Racialized Subjects of Alberta's Oil Boomtown. In L. Adkin (Ed.), First World Petro-Politics: The Political Ecology and Governance of Alberta (pp.263-297), University of Toronto Press.
- Parlee, B. (2016). Mobilizing to Address the Impacts of Oil Sands Development: First Nations in Environmental Governance. In L. Adkin (Ed.), First World Petro-Politics: The Political Ecology and Governance of Alberta (pp.329-355), University of Toronto Press.
- Proshansky, M., Fabian, A. K., & Kaminoff, R. (1983). Place-identity: Physical world socialization of the self. Journal of Environmental Psychology, 3(1), 57-83. https://doi.org/10.1016/S0272-4944(83)80021-8
- Protect Alberta's Rockies and Headwaters [Facebook group]. Retrieved October 13, 2022, from https://www.facebook.com/groups/189175089530814
- Rader, B., Scarpino, S. V., Nande, A., et al. (2020). Crowding and the shape of COVID-19 epidemics. Nature Medicine, 26, 1829-1834. https://doi.org/10.1038/s41591-020-1104-0
- Rajala, K, Sorice, MG, Thomas, VA. The meaning(s) of place: Identifying the structure of sense of place across a social–ecological landscape. People Nat. 2020; 2: 718–733. https://doi.org/10.1002/pan3.10112
- Rauchfleisch, A., Siegen, D., Vogler, D. (2021) How Covid-19 Displaced Climate Change: Mediated Climate Change Activism and Issue Attention in the Swiss Media and Online Sphere, Environmental Communication, DOI: 10.1080/17524032.2021.1990978
- Ritchie, H., Mathieu, E., Rodés-Guirao, L., Appel, C., Giattino, C., Ortiz-Ospina, E., Hasell, J., Macdonald, B., Beltekian, D., & Roser, M. (2020). Coronavirus Pandemic (Covid-19). Our World in Data. Retrieved from https://ourworldindata.org/coronavirus
- Ruess, Christina & Hoffmann, Christian & Boulianne, Shelley & Heger, Katharina. (2021). Online political participation: the evolution of a concept. Information, Communication & Society. 1-18. 10.1080/1369118X.2021.2013919.
- Rume, T., & Islam, S. (2020). Environmental effects of Covid-19 pandemic and potential strategies of sustainability. Heliyon, 6(9), e04965. https://doi.org/10.1016/j.heliyon.2020.e04965
- Saud, M, Mashud, M, Ida, R. (2020). Usage of social media during the pandemic: Seeking support and awareness about Covid-19 through social media platforms. J Public Affairs. 2020; 20:e02417. https://doi-org.qe2a-proxy.mun.ca/10.1002/pa.2417
- Saunders. (2013). Collective Identity and Solidarity: Unity or Factionalism? Environmental Networks and Social Movement Theory. Bloomsbury Academic.

- Shafi, S., & Ran, B. (2021). Social movements as complex adaptive systems: The antecedents and consequences of movement participation in the age of social media. The Social Science Journal. Advance online publication. https://doi.org/10.1080/03623319.2021.1949551
- Snow, D. A., & Benford, D. (2000). Identity work processes in the context of social movements: Clarifying the identity/movement nexus. In S. Stryker, T. J. Owens, & R. W. White (Eds.), Self, identity, and social movements (pp. 41–67). University of Minnesota Press.
- Snow, D.A., Benford, R.D. (1992). Master frames and cycles of protest. See Morris & Mueller 1992, pp. 133–55
- Soja, E. W. (1998). Thirdspace: Journeys to Los Angeles and other Real-and-Imagined Places. Capital & Class, 22(1), 137–139. https://doi.org/10.1177/030981689806400112
- Stahl, B. C. (2021). Emerging technologies as the next pandemic?: Possible consequences of the covid crisis for the future of responsible research and innovation. Ethics and Information Technology, 23, 135-137. doi:https://doi.org/10.1007/s10676-020-09551-1
- Statistics Canada. (2016). Crowsnest Pass- 2016 Socio-Economic Profile Report. Western Economic Diversification Canada.
- Statistics Canada. (2021). Population, Crowsnest Pass. Government of Canada. https://www12.statcan.gc.ca/census-recensement/2021/as-sa/fogsspg/Page.cfm?Lang=E&Dguid=2021A00054815007&topic=1
- Suhay, E., Soni, A., Persico, C., & Marcotte, D. E. (2022). Americans' Trust in Government and Health Behaviors During the COVID-19 Pandemic. RSF: The Russell Sage Foundation Journal of the Social Sciences, 8(8), 221–244. https://www.jstor.org/stable/48704791
- Suk, J., Abhishek, A., Zhang, Y., Ahn, S. Y., Correa, T., Garlough, C., & Shah, D. V. (2021).
 #MeToo, Networked Acknowledgment, and Connective Action: How "Empowerment Through Empathy" Launched a Social Movement. Social Science Computer Review, 39(2), 276–294. https://doi.org/10.1177/0894439319864882
- Takach, G. (2017). Images and Frames of Alberta. (pp. 29-41). Tar wars: oil, environment and Alberta's image. The University of Alberta Press.
- Tarrow, S. G. (1998). Power in Movement : Social Movements and Contentious Politics: Vol. 2nd ed. Cambridge University Press.
- The Canadian Parks and Wilderness Society (CPAWS). (n.d.). About CPAWS. Retrieved from https://cpaws.org/
- Tilly, C. (1978). From Mobilization to Revolution. Random House New York
- Tilly, C., Tarrow, S. (2015). Contentious Politics: Vol. Second edition, fully revised and updated. Oxford University Press.

- Townsend, M. (2020). Revealed: Surge in domestic violence during Covid-19 crisis. The Guardian. Retrieved April 20, 2020 from https://www.theguardian.com/society/2020/apr/12/domestic-violence-surges-sevenhundred-per-cent-uk-coronavirus.
- Tufekci, Z. (2018). Twitter and tear gas: the power and fragility of networked protest. Yale University Press.
- Tufekci, Z., & Wilson, C. (2012). Social media and the decision to participate in political protest: Observations from Tahrir Square. Journal of Communication, 62(2), 363–379. https://doi.org/10.1111/j.1460-2466.2012.01629.x
- United Conservative Party. (2020). Statement of Principles. https://www.unitedconservative.ca/wp-content/uploads/2020/07/Alberta-Strong-and-Free-Platform-1.pdf
- Usher, K., Bhullar, N., Durkin, J., Gyamfi, N., & Jackson, D. (2020). Family violence and Covid-19: Increased vulnerability and reduced options for support. International journal of mental health nursing, 29(4), 549–552. https://doi.org/10.1111/inm.12735
- Valentino-DeVries, J., Lu, D., & Dance, G. J. X. (2020, April 3). Location Data Says It All: Staying at Home During Coronavirus Is a Luxury. New York Times.
- van der Zwet, K., Barros, A.I., van Engers, T.M. et al. (2022) Emergence of protests during the Covid-19 pandemic: quantitative models to explore the contributions of societal conditions. Humanit Soc Sci Commun 9 (68). https://doi.org/10.1057/s41599-022-01082-y
- van Deth, Jan W. (2021). What Is Political Participation?. 10.1093/acrefore/9780190228637.013.68.
- van Dijck, J., & Poell, T. (2015). Social Media and the Transformation of Public Space. Social Media + Society. https://doi.org/10.1177/2056305115622482
- Vanderloo, L. M., Carsley, S., Aglipay, M., Cost, K. T., Maguire, J., and Birken, C. S. (2020). Applying Harm Reduction Principles to Address Screen Time in Young Children amidst the Covid-19 Pandemic. J. Dev. Behav. Pediatr. 41 (5), 335–336. doi:10.1097/dbp.00000000000825
- Vitak J, Zube P, Smock A, Carr CT, Ellison N, Lampe C. (2010) It's complicated: Facebook users' political participation in the 2008 election. Cyberpsychol Behav Soc Netw. 2011 Mar;14(3):107-14. doi: 10.1089/cyber.2009.0226.
- Vitak, J., Zube, P., Smock, A., Carr, C. T., Ellison, N., & Lampe, C. (2011). It's Complicated: Facebook Users' Political Participation in the 2008 Election. Cyberpsychology, Behavior, and Social Networking, 14(3), 107-114. https://doi.org/10.1089/cyber.2009.0226
- Wetherell, D. G. (2005). Making new identities: Alberta small towns confront the city, 1900-1950. Journal of Canadian Studies, 39(1), 175-197. doi:https://doi-org.qe2aproxy.mun.ca/10.1353/jcs.2006.0013

- Wood, T. (2019). The many voices of business: Framing the Keystone pipeline in the US and Canadian news. Journalism. 20(2), pp. 292-231. doi: 10.1177/1464884917717536
- Xu S, Murrell G, Golding SE, Brockett BT, Gatersleben B, Scarles C, White EV, Willis C and Wyles KJ (2021) #Springwatch #WildMorningswithChris: Engaging With Nature via Social Media and Wellbeing During the Covid-19 Lockdown. Front. Psychol. 12:701769. doi: 10.3389/fpsyg.2021.701769
- Zald, M. N., & Ash, R. (1966). Social Movement Organizations: Growth, Decay and Change. Social Forces, 44(3), 327–341. https://doi.org/10.2307/2575833
- Zhong, B., Sun, T., Hu, B. (2022).Information processing on mobile apps: A multilevel analysis of personality and behavioral traits that predict the credibility of smartphone content. China Media Research. Vol. 18 Issue 2, p60-79. 20p.